UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 1	0-0
--------	-----

X	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the quarterly period ended March 31, 2004
	OR
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the transition period from to

CHENIERE ENERGY, INC.

Commission File No. 001-16383

(Exact name as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

95-4352386 (I.R.S. Employer Identification No.)

717 Texas Avenue, Suite 3100 Houston, Texas (Address of principal executive offices)

> 77002 (Zip Code)

(713) 659-1361

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square .

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Securities Exchange Act of 1934). Yes 🗆 No 🗵.

As of May 6, 2004, there were 18,880,592 shares of Cheniere Energy, Inc. Common Stock, \$.003 par value, issued and outstanding.

CHENIERE ENERGY, INC. INDEX TO FORM 10-Q

			r ag
Part I.	Financial Inf	formation	
	Item 1.	Consolidated Financial Statements	
		Consolidated Balance Sheet	3
		Consolidated Statement of Operations	2
		Consolidated Statement of Stockholders' Equity	4
		Consolidated Statement of Cash Flows	(
		Notes to Consolidated Financial Statements	7
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	15
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk	20
	Item 4.	Disclosure Controls and Procedures	20
Part II.	Other Inform	nation_	
	Item 1.	<u>Legal Proceedings</u>	20
	Item 6.	Exhibits and Reports on Form 8-K	2:

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This quarterly report contains certain statements that may include "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. All statements, other than statements of historical facts, included herein or incorporated herein by reference are "forward-looking statements." Included among "forward-looking statements" are, among other things: statements regarding our business strategy, plans and objectives; statements expressing beliefs and expectations regarding our ability to successfully raise the additional capital necessary to meet our obligations under our current exploration agreements; statements expressing beliefs and expectations regarding our ability to secure the leases necessary to facilitate anticipated drilling activities; statements expressing beliefs and expectations regarding our ability to attract additional working interest owners to participate in the exploration and development of our exploration areas; and statements about non-historical information. These forward-looking statements are often identified by the use of terms and phrases such as "expect," "estimate," "project," "plan," "believe," "achievable," "anticipate" and similar terms and phrases. Although we believe that the expectations reflected in these forward-looking statements, which speak only as of the date of this quarterly report.

CHENIERE ENERGY, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEET

ASSETS	March 31, 2004	December 31, 2003
	(Unaudited)	
CURRENT ASSETS		
Cash and Cash Equivalents Accounts Receivable	\$ 14,590,317	\$ 1,257,693
Affiliates	_	1,000,000
Other	1,655,903	1,828,065
Prepaid Expenses	359,783	401,594
Total Current Assets	16,606,003	4,487,352
OIL AND GAS PROPERTIES, full cost method		
Proved Properties, net	1,047,681	1,087,152
Unproved Properties, not subject to amortization	18,539,329	18,047,802
Total Oil and Gas Properties	19,587,010	19,134,954
LNG SITE COSTS	504,967	310,500
FIXED ASSETS, net	942,350	578,281
INVESTMENT IN UNCONSOLIDATED AFFILIATE	_	_
INVESTMENT IN LIMITED PARTNERSHIP	2,155,175	_
INTANGIBLE LNG ASSETS	79,670	79,670
Total Assets	\$ 39,875,175	\$ 24,590,757
LIABILITIES AND STOCKHOLDERS' EQUITY		
CURRENT LIABILITIES		
Accounts Payable	\$ 2,012,748	\$ 1,984,314
Accrued Liabilities Note Payable	1,292,013	1,347,512 1,000,000
Note rayable		1,000,000
Total Current Liabilities	3,304,761	4,331,826
DEFERRED REVENUE	1,000,000	1,000,000
MINORITY INTEREST	_	120,032
COMMITMENTS AND CONTINGENCIES	_	_
STOCKHOLDERS' EQUITY Preferred Stock, \$.0001 par value		
Authorized: 5,000,000 shares Issued and Outstanding: none	_	_
Common Stock, \$.003 par value		
Authorized: 40,000,000 shares		
Issued and Outstanding: 18,716,425 shares at March 31, 2004 and 16,488,187	56140	40.465
shares at December 31, 2003 Additional Paid-in-Capital	56,149 69,044,867	49,465 48,034,244
Deferred Compensation	(3,510,833)	46,034,244
Accumulated Deficit	(30,019,769)	(28,944,810)
Total Stockholders' Equity	35,570,414	19,138,899
Total Liabilities and Stockholders' Equity	\$ 39,875,175	\$ 24,590,757
	ψ <i>55,5,5,175</i>	÷ = .,e> =,r= r

CHENIERE ENERGY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF OPERATIONS (Unaudited)

Three Months Ended March 31,

		c 51,
	2004	2003
Revenues		
Oil and Gas Sales	\$ 332,345	\$ 110,120
Total Revenues	332,345	110,120
Operating Costs and Expenses		
Production Taxes	7,378	
Depreciation, Depletion and Amortization	205,425	58,692
General and Administrative Expenses	203,423	36,092
LNG Terminal Development	3,875,776	394,061
	1,825,834	394,001
Non-Cash Compensation		520 111
Other	1,635,570	520,111
General and Administrative Expenses	7,337,180	914,172
Total Operating Costs and Expenses	7,549,983	972,864
Loss from Operations	(7,217,638)	(862,744)
•		` ′ ′
Equity in Net Income (Loss) of Limited Partnership	2,155,175	(1,200,000)
Gain on Sale of LNG Assets	_	4,760,000
Gain on Sale of Limited Partnership Interest	2.500.000	423,454
Reimbursement from Limited Partnership Investment	2,500,000	
Interest Income	6,032	599
Income (Loss) Before Income Taxes and Minority Interest	(2,556,431)	3,121,309
Provision for Income Taxes	_	<u> </u>
Income (Loss) Before Minority Interest	(2,556,431)	3,121,309
•	` ' ' '	-,,
Minority Interest	1,481,472	
Net Income (Loss)	\$ (1,074,959)	\$ 3,121,309
Net Income (Loss) Per Share—Basic	\$ (0.06)	\$ 0.23
ivet meonie (Loss) i ei Share—Basic	\$ (0.00)	\$ 0.23
Net Income (Loss) Per Share—Diluted	\$ (0.06)	\$ 0.23
Weight J. Access Namber of Observation Decision	10 100 250	12 207 202
Weighted Average Number of Shares Outstanding—Basic	18,109,258	13,297,393
Weighted Average Number of Shares Outstanding—Diluted	18,109,258	13,500,481

CHENIERE ENERGY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (Unaudited)

	Common St	Common Stock Additiona				Total
	Shares	Amount	Paid-In Capital	Deferred Compensation	Accumulated Deficit	Stockholders' Equity
Balance—December 31, 2003	16,488,187	\$ 49,465	\$ 48,034,244	\$ —	\$ (28,944,810)	\$ 19,138,899
Issuances of Stock	1,972,905	5,918	18,146,639	_	_	18,152,557
Issuance of Restricted Stock	255,333	766	3,829,234	(3,830,000)	_	_
Amortization of Deferred Compensation	_	_	_	319,167	_	319,167
Expenses Related to Offerings	_	_	(965,250)	_	_	(965,250)
Net Loss					(1,074,959)	(1,074,959)
Balance—March 31, 2004	18,716,425	\$ 56,149	\$ 69,044,867	\$ (3,510,833)	\$ (30,019,769)	\$ 35,570,414

CHENIERE ENERGY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS (Unaudited)

Three Months Ended March 31,

	Marci	131,
	2004	2003
CASH FLOWS FROM OPERATING ACTIVITIES:		
Net Income (Loss)	\$ (1,074,959)	\$ 3,121,309
Adjustments to Reconcile Net Income (Loss) to Net Cash Used in Operating Activities:	+ (-,···,,)	7 2,121,217
Depreciation, Depletion and Amortization	205,425	58,692
Non-Cash Compensation	1,825,834	
Equity in Net (Income) Loss of Limited Partnership	(2,155,175)	1,200,000
Gain on Sale of LNG Assets	_	(4,760,000)
Gain on Sale of Limited Partnership Interest	_	(423,454)
Reimbursement from Limited Partnership Investment	(2,500,000)	_
Minority Interest	(1,481,472)	_
Other	32,898	(3,636)
Changes in Operating Assets and Liabilities	,	(0,000)
Accounts Receivable—Affiliates	1,000,000	_
Other Accounts Receivable	94,699	456,729
Prepaid Expenses	(89,554)	(236,008)
Accounts Payable and Accrued Liabilities	(161,353)	(730,546
NET CASH USED IN OPERATING ACTIVITIES	(4,303,657)	(1,316,914)
CASH FLOWS FROM INVESTING ACTIVITIES:		
Purchases of Fixed Assets	(563,403)	(12,215)
Oil and Gas Property Additions	(422,823)	(454,847)
Sale of Interest in Oil and Gas Prospects	768,200	391,350
LNG Site Costs	(96,000)	_
Sale of LNG Assets	_	1,000,000
Reimbursement from Limited Partnership Investment	2,500,000	_
NET CASH PROVIDED BY INVESTING ACTIVITIES	2,185,974	924,288
CASH FLOWS FROM FINANCING ACTIVITIES:		
Proceeds from Issuance of Notes Payable	-	225,000
Repayment of Note Payable	(1,000,000)	(18,750)
Sale of Common Stock	16,237,557	_
Offering Costs	(965,250)	_
Partnership Contributions by Minority Owner	1,178,000	_
NET CASH PROVIDED BY FINANCING ACTIVITIES	15,450,307	206,250
NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	13,332,624	(186,376)
	, ,	` ′
CASH AND CASH EQUIVALENTS—BEGINNING OF PERIOD	1,257,693	590,039
CASH AND CASH EQUIVALENTS—END OF PERIOD	\$ 14,590,317	\$ 403,663

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 1—Basis of Presentation

The unaudited consolidated financial statements of Cheniere Energy, Inc. ("Cheniere") have been prepared in accordance with generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In our opinion, all adjustments, consisting only of normal recurring adjustments necessary for a fair presentation, have been included.

For further information, refer to the consolidated financial statements and footnotes included in our Annual Report on Form 10-K for the year ended December 31, 2003. Interim results are not necessarily indicative of results to be expected for the full fiscal year ending December 31, 2004. Certain reclassifications have been made to conform prior period amounts to the current period presentation. These reclassifications have no effect on net income (loss) or stockholders' equity.

New Accounting Pronouncements

In January 2003, the FASB issued Interpretation No. 46, Consolidation of Variable Interest Entities, and subsequently revised the Interpretation in December 2003 (FIN 46R). This Interpretation of Accounting Research Bulletin No. 51, Consolidated Financial Statements, addresses consolidation by business enterprises of variable interest entities, which have certain characteristics. As revised, FIN 46R is now generally effective for financial statements for interim or annual periods ending on or after March 15, 2004. We adopted FIN 46R effective January 1, 2004, with no material effect on our consolidated financial statements.

Stock-Based Compensation

We account for employee stock-based compensation granted under our long-term incentive plans using the intrinsic value method prescribed by Accounting Principles Board Opinion No. 25, *Accounting for Stock Issued to Employees*, and related interpretations. Stock-based compensation expense associated with option grants were not recognized in net income (loss) for the three month periods ended March 31, 2004 and 2003, as all options granted had exercise prices equal to the market value of the underlying common stock on the dates of grant. The following table illustrates the effect on net income (loss) and earnings (loss) per share if we had applied the fair value recognition provisions of SFAS 123 to stock-based employee compensation:

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Three Months Ended March 31. 2004 2003 Net income (loss), as reported \$ (1,074,959) \$ 3,121,309 Deduct: Total stock-based employee compensation expense determined under fair value based method for all awards, net of related tax effects (405,110)(141,062)Pro forma net income (loss) \$ (1,480,069) \$ 2,980,247 Net income (loss) per share: Basic-as reported (0.06)\$ 0.23 Basic-pro forma (0.08)0.22 Diluted-as reported (0.06)0.23 Diluted-pro forma (0.08)0.22

Note 2—Investment in Unconsolidated Affiliate

Prior to January 1, 2003, we accounted for our investment in Gryphon Exploration Company ("Gryphon") using the equity method of accounting because our participation on the Gryphon board of directors provided us with the ability to exercise significant influence over the operating and financial policies of Gryphon. In December 2002, the extent of such influence was diminished when one of the two Cheniere-appointed representatives on the Gryphon board of directors resigned from his position as an officer of Cheniere. Accordingly, effective January 1, 2003, we began accounting for our investment in Gryphon using the cost method of accounting. As of December 31, 2002, Warburg, Pincus Equity Partners, L.P. ("Warburg") had invested \$85,000,000 in Gryphon convertible preferred stock. If Warburg had converted its investment to common stock as of such date, our ownership interest would have been 9.3%. This effective percent ownership remains unchanged as of March 31, 2004.

As of December 31, 2002, as a result of Gryphon's cumulative losses and preferred dividend arrearages, our investment in Gryphon was reduced to zero, but not below zero, because we have not guaranteed any obligations of Gryphon and we are not committed to provide additional financial support to Gryphon.

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 3—Investment in Limited Partnership

In August 2002, we entered into an agreement with entities controlled by Michael S. Smith ("Smith") to sell a 60% interest in the Freeport site and project. On February 27, 2003, we sold our interest in the site and project to Freeport LNG Development, L.P. ("Freeport LNG"), in which we held a 40% limited partner interest. Smith holds a 60% limited partner interest in Freeport LNG. We recovered \$1,740,426 in costs that we had incurred on the project and received an additional \$5,000,000 (\$2,500,000 during 2003 and \$2,500,000 in January 2004) from Freeport LNG. For the funding of Freeport LNG project development costs, Smith also committed to contribute up to \$9,000,000 and to allocate available proceeds from any sales of options or capacity reservations and/or proceeds from loans related to capacity reservations to these costs. In connection with the closing, we issued warrants to Smith to purchase 700,000 shares of Cheniere common stock at a price of \$2.50 per share, exercisable for a period of 10 years.

Effective March 1, 2003, we sold a 10% limited partner interest in Freeport LNG to an affiliate of Contango Oil & Gas Company ("Contango") for \$2,333,333 payable over time, including the cancellation of the Company's \$750,000 short-term note payable. We also issued warrants to Contango to purchase 300,000 shares of Cheniere common stock at a price of \$2.50 per share, exercisable for a period of 10 years. As a result of the sale, we now hold a 30% limited partner interest in Freeport LNG.

We accounted for the transfer of the site and planned LNG receiving terminal to Freeport LNG in accordance with Emerging Issues Task Force Issue No. 01-2, *Interpretations of APB Opinion No. 29.* Accordingly, we recorded a \$4,760,000 gain on sale of LNG assets to the extent of the 60% interest not retained.

We account for our 30% limited partnership investment in Freeport LNG using the equity method of accounting. During 2003, we received installment payments totaling \$2,500,000 from Freeport LNG, which amounts were recorded as a reduction to the basis of our investment in the partnership. In addition, our 30% limited partner interest in the operations of Freeport LNG resulted in our sharing in the net loss of the partnership in the amount of \$4,471,529 for 2003. This non-cash loss reduced the basis of our investment in Freeport LNG to zero, and as a result, we did not record \$278,071 of our equity share of the loss of the partnership as of December 31, 2003.

In January 2004, we received the final \$2,500,000 payment from Freeport LNG. As our investment basis in Freeport LNG had been reduced to zero, the payment was recorded as a reimbursement from limited partnership investment in our consolidated statement of operations for the three months ended March 31, 2004. During the first quarter of 2004, our 30% equity share of net income from the Freeport partnership was \$2,155,175, after deducting the \$278,071 loss that was not recorded as of December 31, 2003.

The financial position of Freeport LNG at March 31, 2004 and December 31, 2003 and the results of Freeport LNG's operations for the quarters ended March 31, 2004 and March 31, 2003 are summarized as follows:

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

	March 31, 2004	December 31, 2003
Current assets	\$ 794,074	\$ 295,000
Fixed assets, net, and security deposit	166,446	150,000
Total assets	\$ 960,520	\$ 445,000
Current liabilities	\$ 1,366,918	\$ 5,887,000
Partners' capital	(406,398)	(5,442,000)
Total liabilities and partners' capital	\$ 960,520	\$ 445,000

	March 31,	
	2004	2003
Revenue	\$ 10,000,150	\$ —
Income (loss) from continuing operations	8,110,820	(4,000,000)
Net income (loss)	8,110,820	(4,000,000)
Cheniere's equity in net income (loss) of limited partnership	2,155,175(1)	(1,200,000)

Three Months ended

Note 4—Minority Interest in Limited Partnership

In May 2003, we formed a limited partnership, Corpus Christi LNG, L.P. ("Corpus LNG") to develop an LNG receiving terminal near Corpus Christi, Texas. Under the terms of the limited partnership agreement, we contributed our technical expertise and know-how, and all of the work in progress related to the Corpus Christi project, in exchange for a 66.7% interest in Corpus LNG.

Our consolidated financial statements include the accounts of Corpus LNG. The \$1,481,472 minority interest included in the consolidated statement of operations for the quarter ended March 31, 2004 is equal to the entire net loss of Corpus LNG due to our investment basis being zero and the minority owner's obligation to fund 100% of project expenses until a total of \$4,500,000 has been reached. Through March 31, 2004, our partner had credited contributions totaling \$4,003,000. At March 31, 2004, other accounts receivable included \$183,440 from our partner, which are attributable to accrued expenses that our partner is obligated to fund.

⁽¹⁾ Represents equity in net income for the first quarter of 2004, less \$278,071 equity in loss not recorded as of December 31, 2003.

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 5—Accrued Liabilities

Accrued liabilities consist of the following:

	March 31, 2004	December 31, 2003
	(Unaudited)	
Taxes other than income	\$ 36,128	\$ 36,986
LNG terminal development costs	931,655	1,183,191
Other accrued liabilities	324,230	127,335
	\$ 1,292,013	\$ 1,347,512

Note 6—Deferred Revenue

On December 23, 2003, Cheniere LNG Services, Inc. ("Cheniere LNG Services"), a wholly-owned subsidiary of Cheniere, entered into a shareholders agreement whereby it became a minority owner of J&S Cheniere S.A., a Switzerland joint-stock company ("J&S Cheniere"). We account for this investment using the cost method of accounting. At March 31, 2004 and December 31, 2003, our investment basis was zero.

Also on December 23, 2003, Cheniere LNG, Inc. ("Cheniere LNG"), a wholly-owned subsidiary of Cheniere, and J&S Cheniere entered into an option agreement providing J&S Cheniere an option to purchase liquefied natural gas ("LNG") storage tank capacity and regas capacity of up to 200 million cubic feet per day ("Mmcf/d") in each of Cheniere LNG's Sabine Pass and Corpus Christi LNG facilities. Following execution of the option agreement, \$1,000,000 was paid by J&S Cheniere to Cheniere LNG in January 2004. The payment was recorded as deferred revenue because the option fee is refundable if Cheniere LNG does not receive approval from the Federal Energy Regulatory Commission ("FERC") for at least one of the terminals or it does not proceed with the development of at least one of the terminals. Upon FERC approval and other related approvals and receipt of permits for each terminal, J&S Cheniere has 60 days to exercise its option at each terminal. The option agreement contemplates negotiation of a definitive terminal use agreement for each of the facilities, which will specify the terms and conditions of the purchase and sale of the capacity and related services. Cheniere LNG will record the option fee as revenue once it is no longer subject to refund.

Note 7—Note Payable

In July 2003, we established a \$5,000,000 line of credit with a commercial bank, with an initial borrowing base of \$2,000,000. The facility is secured by our assets, and its term, as amended, runs through December 31, 2004. Borrowings bear interest at the bank's prime rate plus 2.5% per annum, and a commitment fee of 0.5% per annum is assessed on the unused borrowing base capacity. The interest and commitment fee are payable quarterly. A loan origination fee of 1% of the initial borrowing base was paid at closing. At December 31, 2003, we had a debt obligation of \$1,000,000 and an \$865,142 letter of credit outstanding against this line of credit. The \$1,000,000 debt obligation was repaid in January 2004.

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 8-Non-Cash Transactions

In February 2003, in connection with the sale of a 60% interest in its Freeport LNG project, we issued warrants valued at \$540,015 to purchase 700,000 shares of Cheniere common stock to Freeport LNG Investments LLC. As a result of the closing on the Freeport transaction, we issued additional warrants valued at \$173,576 to purchase 225,000 shares of Cheniere common stock to LNG consultants for services previously performed for us. In March 2003, in connection with the sale of a 10% interest in the limited partnership, we issued warrants valued at \$241,893 to purchase 300,000 shares of Cheniere common stock to Contango, and Contango canceled the \$750,000 note previously payable by Cheniere to Contango.

On February 2, 2004, under the Cheniere, Energy, Inc. 2003 Stock Incentive Plan (the "2003 Stock Incentive Plan"), 383,000 shares were issued to employees and outside directors in the form of bonus and restricted stock awards. We recorded \$1,915,000 of non-cash compensation in February 2004 related to the issuance of 127,667 shares (bonus stock awards) valued at \$15.00 per share, which shares were fully vested on the date of grant. In addition, we recorded \$3,830,000 of deferred compensation as a reduction to stockholders' equity related to the issuance of 255,333 shares (restricted stock awards) valued at \$15.00 per share on the grant date that vest on each of the first and second anniversaries of the grant date. As of March 31, 2004, \$319,167 of deferred compensation had been amortized.

During the first quarter of 2004, 146,516 shares of Cheniere common stock were issued in satisfaction of cashless exercises of stock options to purchase 174,612 shares.

Note 9-Net Income (Loss) Per Share

Basic net income (loss) per share is computed by dividing net income (loss) by the weighted average number of common shares outstanding for the period. The computation of diluted net income (loss) per share reflects the potential dilution that could occur if securities or other contracts to issue common stock that are dilutive to net income were exercised or converted into common stock or resulted in the issuance of common stock that would then share in the earnings of the Company.

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

The following table is a reconciliation of the basic and diluted weighted average shares outstanding for the three months ended March 31, 2004 and 2003:

		Three Months Ended March 31,	
	2004	2003	
Weighted average common shares outstanding:			
Basic	18,109,258	13,297,393	
Dilutive common stock options (a)	_	186,489	
Dilutive common stock warrants (b)	_	16,599	
Diluted	18,109,258	13,500,481	

- (a) In-the-money options representing 1,398,754 weighted average equivalent shares of common stock were not included in the computation of diluted net loss per share for the three months ended March 31, 2004, since they have an anti-dilutive effect to net loss per share. Weighted average options to purchase 171,209 and 784,861 shares of common stock were outstanding but not included in the computations of diluted net (loss) income per share for the three months ended March 31, 2004 and March 31, 2003, respectively, because the exercise prices of the options were greater than the average market price of the common shares and would be anti-dilutive to the computations.
- (b) In-the-money warrants to purchase 1,189,583 weighted average equivalent shares of common stock were not included in the computation of diluted net loss per share for the three months ended March 31, 2004, since they have an anti-dilutive effect to net loss per share. Warrants to purchase 3,604,271 shares of common stock were outstanding but not included in the computations of diluted net income per share for the three months ended March 31, 2003 because the exercise prices of the warrants were greater than the average market price of the common shares and would be anti-dilutive to the computations.

Note 10—Business Segment Information

Our business activities are conducted within two principal operating segments: LNG receiving terminal development and oil and gas exploration and development. These segments operate independently, and there are no intercompany revenues or expenses between them.

The LNG receiving terminal segment is in the preliminary stage of developing LNG receiving terminals along the U.S. Gulf Coast.

The oil and gas exploration and development segment explores for oil and natural gas using a regional database of 7,000 square miles of regional 3D seismic data. Exploration efforts are focused on the shallow waters of the Gulf of Mexico offshore of Louisiana and Texas and consist primarily of seismic data interpretation and prospect generation activities. The segment participates in drilling and production operations with industry partners on the prospects that we generate.

CHENIERE ENERGY, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

	Three Mo	nths Ended
	March 31, 2004	March 31, 2003
Revenues:		
LNG receiving terminal	\$ —	\$ —
Oil & gas exploration and development	332,345	110,120
Total	332,345	110,120
Corporate and other (1)	_	_
Total consolidated	\$ 332,345	\$ 110,120
Net Income (Loss):		
LNG receiving terminal	\$ 1,735,589	\$ 3,572,993
Oil & gas exploration and development	251,123	91,188
·		
Total	1,986,712	3,664,181
Corporate and other (1)	(3,061,671)	(542,872)
Total consolidated	\$ (1,074,959)	\$ 3,121,309
	March 31,	December 31,
	2004	2003
Total Assets:		
LNG receiving terminal	\$ 3,454,694	\$ 2,952,816
Oil & gas exploration and development	20,200,588	20,219,541
Total	23,655,282	23,172,357
Corporate and other (1)	16,219,893	1,418,400
Total consolidated	\$ 39,875,175	\$ 24,590,757
	, ,	. , , ,

⁽¹⁾ Includes corporate activities and certain intercompany eliminations.

Note 11—Subsequent Events

During April 2004, we issued 6,667 shares of common stock pursuant to the exercise of stock options at an average price of \$1.06 per share, generating proceeds of \$7,067. We also issued 157,500 shares of common stock upon the exercise of warrants with an average price of \$4.67 per share, generating \$735,625 in proceeds.

In April 2004, we increased our standby letter of credit by \$257,951 to \$1,123,094 outstanding against our line of credit.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

General

We are engaged primarily in the development of a liquefied natural gas, or LNG, receiving terminal business and related LNG business opportunities centered on the U.S. Gulf Coast. The LNG receiving terminal business consists of receiving deliveries of LNG from LNG ships, processing such LNG to return it to a gaseous state and delivering it to pipelines for transportation to purchasers. We are also engaged in oil and gas exploration, development and exploitation activities in the Gulf of Mexico.

Our unaudited consolidated financial statements and notes thereto relate to the three-month periods ended March 31, 2004 and 2003. These statements, the notes thereto and the consolidated financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2003 contain detailed information that should be referred to in conjunction with the following discussion.

Results of Operations—Comparison of the Three-Month Periods Ended March 31, 2004 and 2003

Overview—Our financial results for the three months ended March 31, 2004 reflect a loss of \$1,074,959, or \$0.06 per share (basic and diluted), compared to net income of \$3,121,309, or \$0.23 per share (basic and diluted), during the corresponding period in 2003.

The major factors contributing to our loss during the first quarter of 2004 were: (1) LNG receiving terminal development expenses of \$3,875,776 (which were offset by a \$1,481,472 minority interest in the operations of Corpus LNG), (2) non-cash compensation of \$1,825,834 (net of \$408,333 capitalized as oil and gas property costs) related to 2004 stock awards to employees and non-employee directors based on Cheniere's performance in 2003 and (3) other general and administrative expenses of \$1,635,570. These factors were offset by a \$2,500,000 reimbursement from our limited partnership investment in Freeport LNG and our equity share of the net income in Freeport LNG of \$2,155,175.

Net income was recorded in the first quarter of 2003 as a result of a \$4,760,000 gain on the sale of our Freeport LNG assets coupled with a \$423,454 gain on the sale of a limited partnership interest in the Freeport LNG terminal. These gains were partially offset by our equity share of the loss in Freeport LNG of \$1,200,000, and LNG receiving terminal development expenses and other general and administrative expenses totaling \$914,172.

LNG Terminal Development Activities—We are in the preliminary stage of developing LNG receiving terminals. As a result, substantially all related costs are expensed as incurred. Such costs primarily include professional fees associated with front-end engineering and design, obtaining FERC approval and other required permitting for two LNG receiving terminals and their related natural gas pipelines. We currently anticipate FERC approval and completion of the permitting process near the end of 2004 with construction of the LNG receiving terminals beginning shortly thereafter.

In the first quarter of 2003, we phased out our direct involvement in developing the Freeport LNG terminal site, but in subsequent periods, we accelerated the schedule of terminal development at two new sites located near Sabine Pass, Louisiana, and near Corpus Christi, Texas. Accordingly, terminal development expenses were 884% higher in the first quarter of 2004 (\$3,875,776) than in the first quarter of 2003 (\$394,061).

In May of 2003, we formed a limited partnership to develop the Corpus LNG terminal. We are the general partner and have a 66.7% limited partner interest. During the first quarter of 2004, we recorded \$1,481,472 in terminal development expenses related to this site; however, this amount was completely offset by the minority interest of our 33.3% partner who provided funding for all development costs through March 31, 2004. We also incurred \$1,917,000 in direct terminal development expenses during the first quarter of 2004 related to the Sabine Pass site, where we own 100% of the project. Additionally, we incurred \$313,123 in LNG employee related costs. In connection with the expansion of our LNG business, our employee costs increased as we expanded our LNG staff from 4 to 15 employees from the first quarter of 2003 to the first quarter of 2004. During the first quarter of 2003, we incurred \$394,061 in LNG receiving terminal development costs. These costs were primarily related to the Freeport LNG project (incurred prior to the sale), LNG employee costs and initial costs related to our Sabine Pass project.

In February 2003, our Freeport LNG terminal project was acquired by Freeport LNG in which we received a 40% limited partnership interest and payments to us totaling \$5,000,000 over time. In connection with the sale of LNG assets to Freeport LNG, we reported a gain of \$4,760,000. We also sold a 10% interest in Freeport LNG in March 2003 for \$2,333,333, resulting in a gain of \$423,454. During 2003, we received payments totaling \$2,500,000 from Freeport LNG, which were recorded as a reduction to our investment in the partnership. In addition, during 2003 we recorded equity in the 2003 loss incurred by Freeport LNG attributable to our 30% limited partner interest, which reduced our investment to zero as of December 31, 2003. In January 2004, we received the final \$2,500,000 payment from Freeport LNG. As our investment basis in Freeport LNG had been reduced to zero, the payment was recorded as a reimbursement from limited partnership investment in our consolidated statement of operations for the three months ended March 31, 2004.

During the first quarter of 2004, our 30% equity share of net income from Freeport LNG was \$2,155,175, after deducting \$278,071 of loss that was not recorded as of December 31, 2003. This compares to our equity share of the loss of \$1,200,000 for the first quarter of 2003. The significant improvement from a loss to net income between quarters for Freeport LNG was a result of Freeport LNG's receipt of a nonrefundable capacity reservation fee of \$10,000,000 from ConocoPhillips in January 2004.

Other General and Administrative Expenses—Other general and administrative ("G&A") expenses relate to our general corporate and other activities. These expenses increased \$1,115,459, or 214%, to \$1,635,570 in the first quarter of 2004 compared to \$520,111 in the corresponding quarter in 2003. The primary reasons for the increase in G&A are related to the expansion of our business (including increases in corporate staff from 4 to 15 employees from the first quarter of 2003 to the first quarter of 2004) and increased professional and other fees in connection with securities compliance filings and increased securities registrations. We capitalize as oil and gas property costs that portion of G&A expenses directly related to our exploration and development activities. We capitalized \$327,604 (in addition to the \$408,333 non-cash compensation mentioned earlier) in the first quarter of 2004 compared to \$232,000 during the comparable period in 2003.

Oil and Gas Activities—Oil and gas revenues increased by \$222,225, or 202%, to \$332,345 in the first quarter of 2004 from \$110,120 in the first quarter of 2003 as a result of increased production volumes (57,638 thousand cubic feet of natural gas equivalent ("Mcfe") in the first quarter of 2004 compared with 14,487 Mcfe in the first quarter of 2003) partially offset by a decrease in average natural gas prices to \$5.77 per thousand cubic feet ("Mcf") in the first quarter of 2004 from \$7.60 per Mcf in the first quarter of 2003. We had production from 11 wells in the first quarter of 2004 as compared with 5 wells in the first quarter of 2003. We incurred little or no production cost in 2003 and 2004 because all of our revenues were generated from non-cost bearing overriding royalty interests. The small amount of production cost in 2004 is attributable to our share of production taxes on a producing well located in Texas state waters.

Liquidity and Capital Resources

Management expects that we will meet our liquidity requirements for the next twelve months through a combination of cash balances, collection of receivables, issuances of our debt or equity securities, issuances of common stock pursuant to exercises by the holders of existing warrants and options, sales of regas capacity in our planned LNG receiving terminals, sales of prospects generated by our exploration group and borrowings under our line of credit. In the event that we are unable to obtain additional capital from one or more of these sources, our operations could be materially adversely affected.

At March 31, 2004, we had working capital of \$13,301,242 compared to \$155,526 at December 31, 2003. The increase is primarily attributable to the sale of our common stock through a private placement offering in January 2004 and exercises of warrants and stock options that resulted in aggregate net proceeds of \$15,272,308. We also received a \$2,500,000 payment from Freeport LNG, which was payable pursuant to the sale of a 60% interest in the Freeport LNG project.

Cash Flow Used in Operating Activities

Net cash used in operations for the three months ended March 31, 2004 totaled \$4,303,657, compared to net cash used in operations of \$1,316,914 for the same period in 2003. The increase in cash used in operations is a direct result of the expansion of our LNG receiving terminal business. In the first quarter of 2003, we phased out our direct involvement in developing the Freeport LNG terminal site, but in subsequent periods, we accelerated the development schedule of our Sabine Pass LNG and Corpus Christi LNG terminals.

Issuances of Common Stock

In January 2004, we issued 1,100,000 shares of Cheniere common stock in a private placement under Regulation D to twelve accredited investors for total consideration of \$14,850,000, or \$13.50 per share. The Company paid a 6.5% sales commission totaling \$965,250, resulting in \$13,884,750 of net proceeds received from the offering. The proceeds of the private placement will be used primarily for the development of LNG receiving terminals and for general corporate purposes.

On February 2, 2004, under the 2003 Stock Incentive Plan, 383,000 shares were issued to employees and outside directors in the form of bonus and restricted stock awards related to our overall 2003 performance. We recorded \$1,915,000 of non-cash compensation in February 2004 related to the issuance of 127,667 shares (bonus stock awards) valued at \$15.00 per share that were fully vested on the date of grant. In addition, we recorded \$3,830,000 of deferred compensation as a reduction to stockholders' equity related to the issuance of 255,333 shares (restricted stock awards) valued at \$15.00 per share on the grant date that vest on each of the first and second anniversaries of the grant date. As of March 31, 2004, \$319,167 of deferred compensation had been amortized.

In the first quarter of 2004, a total of 488,722 shares of Cheniere common stock were issued pursuant to the exercise of stock options, resulting in net cash proceeds of \$937,558. An additional 146,516 shares of Cheniere common stock were issued in satisfaction of cashless exercises of options to purchase 174,612 shares. Also, a total of 110,000 shares of Cheniere common stock were issued pursuant to the exercise of warrants, resulting in net proceeds of \$450,000.

Bank Line of Credit

On July 25, 2003, we established a \$5,000,000 line of credit with a commercial bank, with an initial borrowing base of \$2,000,000. The facility is secured by our assets, and its term, as amended, runs through December 31, 2004. Borrowings bear interest at the bank's prime rate plus 2.5% per annum. In

addition, a commitment fee of 0.5% per annum is assessed on the unused borrowing base capacity. A loan origination fee of 1% of the initial borrowing base was paid at closing. During 2003, we borrowed \$1,000,000 under the facility to acquire oil and gas leases. The balance was repaid in January 2004. We also used the facility to establish a standby letter of credit in the amount of \$865,142 in connection with our office lease. In April 2004, the letter of credit amount was increased to \$1,123,094 related to the planned expansion of our office space.

Off-Balance Sheet Arrangements

As of March 31, 2004, we had no "off-balance sheet arrangements" that may have a current or future material affect on our consolidated financial condition or results of operations.

Freeport LNG Terminal

In August 2002, we entered into an agreement to sell a 60% interest in our planned LNG receiving facility near Freeport, Texas. In February 2003, our Freeport LNG project was acquired by Freeport LNG, in which we held a 40% interest. Effective March 1, 2003, we sold a 10% interest in Freeport LNG to Contango for \$2,333,333 payable over time. We now retain a 30% interest in Freeport LNG paid us cash and assumed liabilities related to the Freeport LNG project for costs, which represented an aggregate amount of \$1,740,426, in addition to paying us a \$1,000,000 initial installment at closing. We received additional payments of \$1,500,000 in 2003 and \$2,500,000 in January 2004 from Freeport LNG.

In June 2003, Dow Chemical Company ("Dow") signed an agreement with Freeport LNG for the potential long-term use of the receiving terminal. Under the agreement, Dow will have regas rights to as much as 500 Mmcf/d beginning with commercial start-up of the facility in 2007. In February 2004, Freeport LNG and Dow entered into a 20-year Terminal Use Agreement providing for a firm commitment by Dow for the use of 250 Mmcf/d of regas capacity and an option by Dow until August 2004 to acquire an additional 250 Mmcf/d of regas capacity.

On December 21, 2003, ConocoPhillips and Freeport LNG signed an agreement under which ConocoPhillips will reserve one billion cubic feet per day of regas capacity in the terminal at the Freeport facility. ConocoPhillips will also obtain a 50% interest in the general partner of Freeport LNG and provide a substantial majority of the financing to construct the facility, which is currently estimated to cost in excess of \$500 million. Freeport LNG received a nonrefundable capacity reservation fee of \$10,000,000 from ConocoPhillips in January 2004. The ConocoPhillips transaction is expected to close in the spring of 2004, subject to completion of remaining documentation and satisfaction of closing conditions.

Under the Freeport LNG partnership agreement, development expenses of the Freeport LNG project are to be funded out of the partnership's cash flows and by its 60% limited partner. Accordingly, we have not been called upon to contribute any cash to Freeport for development activities, nor do we anticipate being called upon to make capital contributions in order for Freeport LNG to complete construction of the facility to be financed pursuant to the agreements with ConocoPhillips. If, in the future, after closing of the ConocoPhillips transaction, we were to be called upon to contribute additional capital to Freeport LNG, for expansion of the project or any other purpose, we would have the option either to contribute the requested capital or else to decline to contribute, in which case the other partners could elect to make our contribution and receive back, without interest, twice the contributed amount out of future Freeport LNG cash flows otherwise distributable to us.

Corpus Christi LNG Terminal

Under the terms of the limited partnership agreement of Corpus LNG, which was formed for our Corpus Christi LNG receiving terminal project, we contributed our technical expertise and know-how, and all of the work in progress related to the Corpus Christi project, in exchange for a 66.7% limited partner interest in Corpus LNG. BPU LNG committed to contribute its approximately 210-acre tract of land plus related easements along with rights to an additional 400 acres, and cash to fund 100% of the first \$4,500,000 of Corpus LNG project expenses in exchange for a 33.3% limited partner interest. We will manage the project through the general partner interest held by our wholly-owned subsidiary. By mid-April 2004, BPU LNG had contributed \$4,378,000 to the partnership. As a result, we plan to begin funding our 66.7% limited partner share of project expenses by May 2004, and BPU LNG will fund their 33.3% limited partner share.

Other Matters

New Accounting Pronouncements

In January 2003, the FASB issued Interpretation No. 46, Consolidation of Variable Interest Entities, and subsequently revised the Interpretation in December 2003 (FIN 46R). This Interpretation of Accounting Research Bulletin No. 51, Consolidated Financial Statements, addresses consolidation by business enterprises of variable interest entities, which have certain characteristics. As revised, FIN 46R is now generally effective for financial statements for interim or annual periods ending on or after March 15, 2004. We adopted FIN 46R effective January 1, 2004, with no material effect on our consolidated financial statements.

Other Recent Developments

In July 2003, an issue was brought before the Financial Accounting Standards Board (FASB) regarding whether or not contract-based oil and gas mineral rights held by lease or contract ("mineral rights") should be recorded or disclosed as intangible assets. The issue presents a view that these mineral rights are intangible assets as defined in Statement of Financial Accounting Standards (SFAS) No. 141, "Business Combinations," and, therefore, should be classified separately on the balance sheet as intangible assets. SFAS No. 141 and SFAS No. 142, "Goodwill and Other Intangible Assets," became effective for transactions subsequent to June 30, 2001, with the disclosure requirements of SFAS No. 142 required as of January 1, 2002. SFAS No. 141 requires that all business combinations initiated after June 30, 2001 be accounted for using the purchase method and that intangible assets be disaggregated and reported separately from goodwill. SFAS No. 142 established new accounting guidelines for both finite lived intangible assets and indefinite lived intangible assets. Under the statement, intangible assets should be separately reported on the face of the balance sheet and accompanied by disclosure in the notes to financial statements. SFAS No. 142 does not apply to accounting utilized by the oil and gas industry as prescribed by SFAS No. 19, and is silent about whether or not its disclosure provisions apply to oil and gas companies. The Emerging Issues Task Force (EITF) has added the treatment of oil and gas mineral rights to an upcoming agenda, which may result in a change in how we classify these assets.

Should such a change be required, the amounts related to business combinations and major asset purchases that would be classified as "intangible undeveloped mineral interest" are immaterial as of March 31, 2004 and December 31, 2003. The amounts related to business combinations and major asset purchases that would be classified as "intangible developed mineral interest" are also immaterial as of March 31, 2004 and December 31, 2003.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

The development of our LNG receiving terminal business is based upon the foundational premise that prices of natural gas in the U.S. will be sustained at levels of \$3.00 per Mcf or more. Should the price of natural gas in the U.S. decline to sustained levels below \$3.00 per Mcf, our ability to develop and operate LNG receiving terminals could be materially adversely affected.

We produce and sell natural gas, crude oil and condensate. As a result, our financial results can be affected as these commodity prices fluctuate widely in response to changing market forces. We have not entered into any derivative transactions.

In the normal course of business, our financial condition is exposed to minimal market risk associated with interest rate movements on our borrowings. A one percent increase or decrease in the levels of interest rates on variable rate debt would not result in a material change to our results of operations.

Item 4. Disclosure Controls and Procedures

We maintain a set of disclosure controls and procedures that are designed to ensure that information required to be disclosed by us in the reports filed by us under the Securities Exchange Act of 1934, as amended ("Exchange Act"), is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms. As of the end of the period covered by this report, we evaluated, under the supervision and with the participation of our management, including our President, Chief Executive Officer and Chairman of the Board and our Vice President & Chief Financial Officer, Secretary and Treasurer, of the effectiveness of our disclosure controls and procedures pursuant to Rule 13a-15 of the Exchange Act. Based on that evaluation, our President, Chief Executive Officer and Chairman of the Board and our Vice President & Chief Financial Officer, Secretary and Treasurer concluded that our disclosure controls and procedures are effective.

During the most recent fiscal quarter, there have been no changes in our internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II. Other Information

Item 1. Legal Proceedings

The Company has been and may in the future be involved as a party to various legal proceedings, which are incidental to the ordinary course of business. Management regularly analyzes current information and as necessary, provides accruals for probable liabilities on the eventual disposition of these matters. In the opinion of management and legal counsel, as of March 31, 2004, there were no threatened or pending legal matters that would have a material impact on the Company's consolidated results of operations, financial position or cash flows.

Item 6. Exhibits and Reports on Form 8-K

(a) Each of the following exhibits is incorporated by reference or filed herewith

Exhibit No.	Description
10.1	LNG Terminal Use Agreement, dated March 1, 2004 between The Dow Chemical Company and Freeport LNG Development, L.P. (confidential treatment has been requested with respect to certain portions of this exhibit)
10.2	Form of Subscription Agreement between the Company and the investors to be identified therein in connection with the private placement completed in January 2004.
31.1	Certification by Chief Executive Officer required by Rule 13a-14(a) and 15d-14(a) under the Exchange Act
31.2	Certification by Chief Financial Officer required by Rule 13a-14(a) and 15d-14(a) under the Exchange Act
32.1	Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

(b) Current Reports on Form 8-K:

January 7, 2004—Cheniere filed a Current Report on Form 8-K announcing that its Chairman would be making an energy conference presentation.

January 14, 2004—Cheniere filed a Current Report on Form 8-K announcing that it had entered into an option with Atlantic Marine, Inc. for the purchase of a potential LNG receiving terminal site in an industrial zone on Pinto Island in Mobile Bay, Alabama.

January 16, 2004—Cheniere filed a Current Report on Form 8-K announcing the adjournment of its special meeting of stockholders.

January 26, 2004—Cheniere filed a Current Report on form 8-K, as amended by Form 8-K/A filed on January 27, 2004, announcing that it had completed a private placement to certain accredited investors of 1,100,000 shares of its common stock.

February 3, 2004—Cheniere filed a Current Report on Form 8-K announcing that Mr. J. Robinson West had been elected to its board of directors.

March 2, 2004—Cheniere filed a Current Report on Form 8-K announcing that Freeport LNG had entered into a 20-year terminal use agreement with Dow.

March 31, 2004—Cheniere filed a Current Report on Form 8-K announcing its results of operations for the fiscal year ended December 31, 2003.

SIGNATURES

Pursuant to the requirements of the Securities and Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

CHENIERE ENERGY, INC.

/s/ Craig K. Townsend

Vice President and Controller (on behalf of the registrant and as principal accounting officer)

Date: May 13, 2004

*** indicates material has been omitted pursuant to a Confidential Treatment Request filed with the Securities and Exchange Commission. A complete copy of this agreement has been filed separately with the Securities and Exchange Commission.

LNG TERMINAL USE AGREEMENT

between

THE DOW CHEMICAL COMPANY

and

FREEPORT LNG DEVELOPMENT, L.P.

dated March 1, 2004

TABLE OF CONTENTS

		Page
ARTICL	E 1 DEFINITIONS	1
ARTICL	LE 2 SERVICES AND SCOPE	10
2.1 2.2 2.3	Services to be Provided by FLNG Additional Services Activities Outside Scope of this Agreement	10 10 10
ARTICL	LE 3 SALE AND PURCHASE OF SERVICES	11
3.1 3.2 3.3 3.4 3.5	Services Quantity Customer's Use of Services Quantity Gas Redelivery Failure to Take Delivery of Gas at Delivery Point Freeport Services Manual	11 12 12 14 14
ARTICL	LE 4 COMPENSATION FOR SERVICES	15
4.1 4.2 4.3 4.4 4.5 4.6 4.7	Fee Retainage Services Unavailability Construction Cost Deviation Amount and Additional Capital Costs New Taxes Services Provided to Other Customers Certain Adjustments to Reservation Fee	15 16 16 17 18 19
ARTICL	E 5 SCHEDULING	20
5.1 5.2 5.3 5.4 5.5	Customer LNG Receipt Schedule Gas Delivery Procedure Standard Scheduling Representative Scheduling Coordination Among Customer and Other Customers	20 23 25 26 26
ARTICL	LE 6 RELEASE OF SERVICES	26
6.1 6.2	General Temporary Release	26 26
ARTICL	LE 7 TERM	29
7.1 7.2 7.3 7.4	Term Commencement of Deliveries Delay Caused by Force Majeure Construction Progress Reports	29 29 31 31
ARTICL	LE 8 FREEPORT FACILITY	31
8.1 8.2 8.3 8.4	Freeport Facility Modifications to Freeport Facility Customer Inspection Rights Responsible Practices	31 33 34 34

i

ARTICLE	E 9 TRANSPORTATION AND UNLOADING	36
9.1	LNG Vessels	36
9.2	Freeport Facility Marine Operations Manual	39
9.3	LNG Vessel Inspections; Right to Reject LNG Vessel	40
9.4	Advance Notices Regarding LNG Vessel and Cargoes	40
9.5	Notice of Readiness	42
9.6	Berthing Assignment	43
9.7	Unloading Time	44
9.8	Unloading at the Freeport Facility	45
9.9	LNG Vessel Not Ready for Unloading; Excess Berth Time	45
ARTICLE	E 10 RECEIPT OF LNG	47
10.1	Title, Custody and Risk of Loss	47
10.2	No Encumbrance	47
10.3	Receipt of LNG	48
10.4	Quality and Measurement of Customer's LNG	48
10.5	Off-Specification LNG	49
10.6	Customer's Responsibility and Reimbursement	50
10.7	Subsequent Deliveries	50
ARTICLE	E 11 REDELIVERY OF GAS	51
11.1	General	51
11.2	Customer's Responsibility	51
11.3	Specifications and Measurement of Gas at the Delivery Point	52
11.4	Nonconforming Gas	53
ARTICLE	E 12 PAYMENT	53
12.1	Monthly Statements	53
12.2	Other Statements	54
12.3	Adjustments	54
12.4	Payment Due Dates	54
12.5	Payment	55
12.6	Nonpayment	55
12.7	Disputed Statements	56
12.8	Final Settlement	56
ARTICLE	E 13 SECURED FINANCING	56
13.1	UCC 1	56
13.2	Secured Financing	56
ARTICLE	E 14 DUTIES, TAXES AND OTHER GOVERNMENTAL CHARGES	57
ARTICLE	E 15 INSURANCE	57
15.1	FLNG's Insurance	57
15.2	Customer's Insurance	58
15.3	Port Liability Agreement	59
ARTICLE	E 16 LIABILITIES	59
16.1	Limitation of Liability	59
16.1	Limitation on Damages	59
10.2	Difficultion on Duffinger	3)

Limitation on Damages

ARTICLE	2.17 FORCE MAJEURE	60
17.1	Events of Force Majeure	60
17.2	Limitation on Scope of Force Majeure Protection	60
17.3	Notice	60
17.4	Measures	61
17.5	No Extension of Term	61
17.6 17.7	Settlement of Industrial Disturbances Allocation of Services	61 61
	2.18 CURTAILMENT OF SERVICES OR TEMPORARY DISCONTINUATION OF SERVICES	62
18.1	Scheduled Curtailment or Temporary Discontinuation of Services	62
18.2	Unscheduled Curtailment or Temporary Discontinuation of Services	62
	2.19 ASSIGNMENT	63
19.1	Restrictions on Assignment	63
19.2	Permitted Assignments	63
19.3	Assignment as Novation	64
ARTICLE	20 CONDITIONS PRECEDENT	65
20.1	Conditions Precedent	65
20.2	Satisfaction of Conditions Precedent	65
ARTICLE	21 TERMINATION	65
21.1	Early Termination Events	65
21.2	Exercise of Right of Termination	66
21.3	Other Termination Provisions	66
21.4	Consequences of Termination	66
ARTICLE	22 APPLICABLE LAW	66
ARTICLE	23 DISPUTE RESOLUTION	66
23.1	Dispute Resolution	66
23.2	Expert Determination	67
ARTICLE	24 CONFIDENTIALITY	68
24.1	Confidentiality Obligation	68
24.2	Public Announcements	69
ARTICLE	2 25 REPRESENTATIONS AND WARRANTIES	70
25.1	Representations and Warranties of Customer	70
25.2	Representations and Warranties of FLNG	71
ARTICLE	2 26 NOTICES	71
ARTICLE	2 27 MISCELLANEOUS	72
27.1	Amendments	72
27.2	Approvals	72
27.3	Successors and Assigns	72
27.4	Waiver	72
27.5	No Third Party Beneficiaries	72
27.6	Supplemental Terms	73

27.7	Rules of Construction	73
27.8	Survival of Rights	73
27.9	Rights and Remedies	73
27.10	Interpretation	74
27.11	Disclaimer of Agency	74
27.12	Severance of Invalid Provisions	74
27.13	Compliance with Laws	75
27.14	Conflicts of Interest	75
27.15	Expenses	76
27.16	Scope	76
27.17	Counterpart Execution	76
Annex I	 Measurements and Tests of LNG at Receipt Point 	
Annex II	 Measurements and Tests for Gas at Delivery Point 	
Exhibit A	 Terms of Freeport Facility Lease 	
Exhibit B	 Freeport Services Manual 	
Exhibit C	 Port Liability Agreement 	
Exhibit D	Supplemental Terms	

LNG TERMINAL USE AGREEMENT

This LNG TERMINAL USE AGREEMENT (the "Agreement"), dated as of this 1st day of March 2004 (the "Effective Date"), is made by and between THE DOW CHEMICAL COMPANY, a Delaware corporation with its principal office at 2030 Dow Center, Midland, Michigan, U.S.A. 48674 ("Customer"); and FREEPORT LNG DEVELOPMENT, L.P., a Delaware limited partnership with a place of business at 1200 Smith Street, Suite 600, Houston, Texas, U.S.A. 77002 ("FLNG").

RECITALS

WHEREAS, FLNG intends to construct, own and operate an LNG terminal facility near Freeport, Texas capable of performing certain LNG terminalling services, including: the berthing of LNG vessels; the unloading, receiving and storing of LNG; the regasification of LNG; the storage of natural gas; and the transportation and delivery of natural gas to a pipeline interconnection point at Stratton Ridge, Texas;

WHEREAS, Customer will purchase LNG for importation into the United States and desires to purchase such LNG terminalling services from FLNG; and

WHEREAS, FLNG desires to make such LNG terminalling services available to Customer and to Other Customers in accordance with the terms hereof.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by the Parties hereto and for the mutual covenants contained herein, FLNG and Customer hereby agree as follows:

ARTICLE 1 DEFINITIONS

In addition to any terms or expressions defined elsewhere in this Agreement, the terms or expressions set forth below shall have the following meanings in this Agreement:

- **"Adverse Weather Conditions"** means weather and sea conditions actually experienced at or near the Freeport Facility that are sufficiently severe either: (a) to prevent an LNG vessel from proceeding to berth, or unloading or departing from berth, in accordance with one or more of the following: (i) regulations published by a Governmental Authority, (ii) an Approval, or (iii) an order of a Pilot; or (b) to cause an actual determination by the Master of an LNG Vessel that it is unsafe for such vessel to berth, unload or depart from berth.
- **1.2** "Affiliate" means a Person (other than a Party) that directly or indirectly controls, is controlled by, or is under common control with, a Party to this Agreement, and for such purposes the terms "control", "controlled by" and other derivatives shall mean the direct or indirect ownership of more than fifty percent (50%) of the voting rights in a Person.
- **1.3** "Aggregate Contracted Capacity" shall have the meaning set forth in Section 4.7(a)(i).

- **1.4** "Agreement" means this agreement, together with the Annexes and Exhibits attached hereto, which are hereby incorporated into and made a part hereof, as the same may be hereafter amended.
- 1.5 "Alternate ***" shall have the meaning set forth in Section 3.3(c)(ii).
- 1.6 "Approvals" means all consents, authorizations, licenses, waivers, permits, approvals and other similar documents from or by a Governmental Authority.
- 1.7 "Arrival Date" shall have the meaning set forth in Section 5.1(b)(i).
- **1.8** "Arrival Location" shall have the meaning set forth in Section 9.5(a).
- 1.9 "Base Rate" means: (a) the interest rate per annum equal to (i) the prime rate (sometimes referred to as the base rate) for corporate loans as published by the Wall Street Journal in the money rates section on the applicable date, or (ii) in the event The Wall Street Journal ceases or fails to publish such a rate, the prime rate (or an equivalent thereof) in the United States for corporate loans determined as the average of the rates referred to as prime rate, base rate or the equivalent thereof, quoted by Chase Manhattan Bank, Citibank and Bank of New York, or any successor thereof, for short term corporate loans in New York on the applicable date; plus (b) *** percent (****%). The Base Rate shall change as and when the underlying components thereof change, without notice to any Person.
- **1.10 "British Thermal Unit"** or **"BTU"** means the amount of heat required to raise the temperature of one (1) avoirdupois pound of pure water from 59.0 degrees Fahrenheit to 60.0 degrees Fahrenheit at an absolute pressure of 14.696 pounds per square inch.
- 1.11 "Business Day" means any day other than a weekend day or U.S. Federal banking holiday in Houston, Texas.
- 1.12 "Cargo" means a quantity of LNG expressed in MMBTU carried by an LNG Vessel in relation to which FLNG will render Services hereunder.
- 1.13 "Central Time" means Central Time Zone, as adjusted for Daylight Saving Time and Standard Time.
- **1.14** "Claims" shall have the meaning set forth in Section 10.2 of this Agreement.
- **1.15 "Commercial Start Date"** shall have the meaning set forth in Section 7.2.
- **1.16 "Conditions Precedent"** shall have the meaning set forth in Section 20.1.
- **1.17 "Contract Year"** means each annual period starting on October 1 and ending on September 30 during the Term of this Agreement; provided, however, that (a) the first Contract Year shall commence on the Commercial Start Date and end on the following September 30, and (b) the last Contract Year shall commence on October 1 immediately preceding the last day of the Term and end on the last day of the Term as set forth in Section 7.1.

- **1.18** "Cubic Meter" means a volume equal to the volume of a cube each edge of which is one (1) meter.
- **1.19** "Customer" means The Dow Chemical Company, subject to Article 19.
- 1.20 "Customer LNG Receipt Schedule" shall have the meaning set forth in Section 5.1(f).
- 1.21 "Customer's LNG" means, for the purposes of Services, LNG received at the Receipt Point for Customer's account.
- 1.22 "Customer's Inventory" means, at any given time, the quantity in MMBTUs that represents LNG and Gas (whether or not such Gas is Regasified LNG or Gas derived or produced from sources other than Regasified LNG) held for Customer's account. For the avoidance of doubt, Customer's Inventory shall (i) be determined after deduction of Retainage in accordance with Sections 3.1(b)(ii) and 4.2, and (ii) include Temporary Release Inventory.
- 1.23 "Delivery Point" means the point of interconnect between the Freeport Facility Pipeline and a Downstream Pipeline at Stratton Ridge, Texas.
- **"Dispute"** means any dispute, controversy or claim (of any and every kind or type, whether based on contract, tort, statute, regulation, or otherwise) arising out of, relating to, or connected with this Agreement, including any dispute as to the construction, validity, interpretation, termination, enforceability or breach of this Agreement, as well as any dispute over jurisdiction.
- 1.25 "Downstream Pipeline" means all Gas pipelines downstream of the Delivery Point which transport Gas from the Freeport Facility.
- **1.26 "Effective Date"** means the date set forth in the preamble of this Agreement.
- 1.27 "Excess Reception Fee" shall have the meaning set forth in Section 4.1(c).
- **"Expected Receipt Quantity"** means, with respect to a given Cargo, Customer's reasonable estimate of the quantity of LNG (in MMBTUs) that such Cargo is expected to unload at the Freeport Facility, taking into consideration (a) the expected composition of the Cargo anticipated to be loaded at the Loading Port, (b) the expected loaded volume of such Cargo, (c) the natural boil-off, and fuel gas used by the vessel in excess of natural boil-off, expected during shipment of the Cargo, (d) the anticipated time required from the commencement of loading of such Cargo to the completion of unloading of such Cargo, and (e) the anticipated quantity of heel to be retained by such Cargo. For purposes of this Agreement, the term "Expected Receipt Quantity" shall be the estimate set forth in the notice delivered pursuant to Section 5.1(b)(ii), as may be subsequently amended pursuant to Section 9.4(a).

- **1.29** "Extension Term" shall have the meaning set forth in Section 7.1(c).
- **1.30 "Fee"** shall have the meaning set forth in Section 4.1.
- 1.31 "FERC" means the Federal Energy Regulatory Commission or a successor regulatory agency.
- 1.32 "Fixed Fee Component" shall have the meaning set forth in Section 4.1(a)(i).
- **1.33 "FLNG"** means Freeport LNG Development, L.P., subject to Article 19.
- **1.34 "Force Majeure"** shall have the meaning set forth in Section 17.1.
- 1.35 "for Customer", "for Customer's account", "on behalf of Customer" or other phrases containing similar wording shall include LNG delivered to the Freeport Facility at Customer's direction or LNG delivered to the Freeport Facility by a Temporary Customer, as the context may require, and shall include both Customer's Inventory and any Temporary Release Inventory.
- 1.36 "Freeport Facility" means the following FLNG facilities in Brazoria, County near Freeport, Texas used to provide Services hereunder, as such facilities will be constructed and modified from time to time in accordance with this Agreement: (a) the LNG receiving terminal facility as further described in Section 8.1(b) (including the port, berthing and unloading facilities, LNG storage facilities, and regasification facilities, together with equipment and facilities related thereto); and (b) the Freeport Facility Pipeline.
- **1.37 "Freeport Facility Lease"** means that certain Ground Lease and Development Agreement between Brazos River Harbor Navigation District of Brazoria County, Texas and FLNG dated as of December 12, 2002.
- **1.38** "Freeport Facility Marine Operations Manual" shall have the meaning set forth in Section 9.2.
- 1.39 "Freeport Facility Pipeline" means the pipeline approximately nine (9) miles in length and at least thirty-six (36) inches in diameter, with a maximum allowable operating pressure of no less than 1440 psig, which pipeline is constructed and operated as a part of the Freeport Facility to (a) interconnect with one or more Downstream Pipelines at Stratton Ridge, Texas and (b) transport to the Delivery Point either Regasified LNG and/or Gas stored in one or more Gas Storage Facilities.
- **1.40 "Freeport Services Manual"** shall have the meaning set forth in Section 3.5.
- 1.41 "***" means the internet based computer system used by FLNG to communicate with Customer and Other Customers regarding the scheduling of LNG terminalling services at the Freeport Facility.

- 1.42 "Gas" means any hydrocarbon or mixture of hydrocarbons consisting predominantly of methane which is in a gaseous state.
- 1.43 "Gas Storage Facility" means an underground Gas storage facility near Stratton Ridge, Texas owned, operated, contracted or otherwise made available to FLNG for the purposes of storing Gas solely for the account of Customer and Other Customers.
- **1.44 "Governmental Authority"** means, in respect of any country, any national, regional, state, or local government, any subdivision, agency, commission or authority thereof (including any maritime authorities, port authority or any quasi-governmental agency) having jurisdiction over a Party, the Freeport Facility, a Gas Storage Facility, Customer's Inventory, an LNG Vessel, a Transporter, or a Downstream Pipeline, as the case may be, and acting within its legal authority.
- **1.45** "GPA" shall have the meaning set forth in Annex I.
- **"Gross Heating Value"** means the quantity of heat expressed in BTUs produced by the complete combustion in air of one (1) cubic foot of anhydrous gas, at a temperature of 60.0 degrees Fahrenheit and at an absolute pressure of 14.696 pounds per square inch, with the air at the same temperature and pressure as the gas, after cooling the products of the combustion to the initial temperature of the gas and air, and after condensation of the water formed by combustion.
- 1.47 "Henry Hub Price" shall mean, with respect to any month, the final settlement price in dollars per MMBTU for the NYMEX Henry Hub Natural Gas futures contract for Gas to be delivered during such month, such final price to be based upon the last trading day for the contract for such month; provided, however that if the NYMEX Henry Hub Natural Gas futures contract settlement prices ceases to be published for any reason, the Parties shall select a comparable index to be used in its place that maintains the intent and economic effect of the original index.
- **1.48** "Incremental Costs" shall have the meaning set forth in Section 4.1(b).
- 1.49 "Input A" means *** per MMBTU unless otherwise modified by Section 4.4.
- "International LNG Terminal Standards" means, to the extent not inconsistent with the express requirements of this Agreement, the international standards and practices applicable to the design, equipment, operation or maintenance of LNG receiving and regasification terminals, established by the following (such standards to apply in the following order of priority): (i) a Governmental Authority having jurisdiction over FLNG; (ii) the Society of International Gas Tanker and Terminal Operators ("SIGTTO"); and (iii) any other internationally recognized non-governmental agency or organization with whose standards and practices it is customary for Reasonable and Prudent Operators of LNG receiving and regasification terminals to comply. In the event of a conflict between any of the priorities noted above, the priority with the lowest roman numeral noted above shall prevail.

- 1.51 "International LNG Vessel Standards" means, to the extent not inconsistent with the expressed requirements of this Agreement, the international standards and practices applicable to the ownership, design, equipment, operation or maintenance of LNG vessels established by the following (such standards to apply in the following order of priority): (i) a Governmental Authority; (ii) the International Maritime Organization; (iii) SIGTTO; and (iv) any other internationally recognized non-governmental agency or organization with whose standards and practices it is customary for Reasonable and Prudent Operators of LNG vessels to comply. In the event of a conflict between any of the priorities noted above, the priority with the lowest roman numeral noted above shall prevail.
- 1.52 "Liabilities" means all liabilities, costs, claims, disputes, demands, suits, legal or administrative proceedings, judgments, damages, losses and expenses (including reasonable attorneys' fees and other reasonable costs of litigation or defense), and any and all fines, penalties and assessments of, or responsibilities to, Governmental Authorities.
- 1.53 "Liquids" means liquid hydrocarbons capable of being extracted from LNG at the Freeport Facility, consisting predominately of ethane, propane, butane and longer-chain hydrocarbons.
- **1.54** "Loading Port" means the port at which a Cargo is loaded on board an LNG Vessel.
- 1.55 "LNG" means Gas in a liquid state.
- **1.56 "LNG Suppliers"** means, in relation to performance of the obligations of FLNG and Customer under this Agreement, those Persons who agree in writing pursuant to an agreement to supply or sell LNG to Customer or a Temporary Customer for delivery to the Freeport Facility.
- 1.57 "LNG Vessel" means an ocean-going vessel suitable for transporting LNG that Customer, a Temporary Customer, or an LNG Supplier uses for transportation of LNG to the Freeport Facility.
- **1.58 "Major Customer"** means Customer and any Other Customer whose terminal use agreement entitles such Other Customer to berth, unload and deliver at the Freeport Facility quantities of LNG greater than *** MMBTUs per Contract Year for a contractual term of at least *** (***) years.
- **1.59** "Major Customer Allocation Priority" shall have the meaning set forth in Section 17.7.
- **1.60** "***" shall have the meaning set forth in Section 3.3(b).
- **"Maximum LNG Reception Quantity"** means, except as otherwise provided in Section 3.1(c), 195,275,000 MMBTUs of LNG per Contract Year; provided, however, that for purposes of the first and last Contract Year, the Maximum LNG Reception Quantity shall be prorated based upon the ratio that the number of days during such Contract Year bears to three hundred sixty-five (365).

- 1.62 "***" shall have the meaning set forth in Section 3.3(c).
- **1.63** "MMBTU" means 1,000,000 BTUs.
- **"Monthly Component"** means twelve (12); provided, however, that for the first and last Contract Year, Monthly Component shall mean the number of months in the respective Contract Year with the first and last months in such Contract Year being prorated based upon the number of days in such month.
- **1.65** "Nomination Notice" shall have the meaning set forth in Section 5.2(c).
- **1.66** "Non-Major Customer" means any Other Customer other than a Major Customer.
- **1.67** "NOR Window" shall have the meaning set forth in Section 9.5(b)(ii).
- **1.68** "Notice of Readiness" or "NOR" shall have the meaning set forth in Section 9.5.
- 1.69 "Other Customers" means, from time to time, Persons (other than Customer and Temporary Customers) receiving LNG terminalling services from FLNG similar to the Services.
- 1.70 "Party" and "Parties" means FLNG and Customer, and their respective successors and assigns.
- **1.71 "Payment Period"** shall have the meaning set forth in Section 4.1.
- 1.72 "Peaking Gas" shall have the meaning set forth in Section 3.1(b)(iii).
- 1.73 "Person" means any individual, corporation, partnership, trust, unincorporated organization or other legal entity, including any Governmental Authority.
- 1.74 "Pilot" means any Person engaged by Transporter to come onboard an LNG Vessel to assist the Master in pilotage, mooring and unmooring of such LNG Vessel and to assist in coordinating the unloading of LNG at the Freeport Facility.
- 1.75 "Pipeline Standards" means, to the extent not inconsistent with the express requirements of this Agreement, the standards and practices applicable to the design, equipment, operation or maintenance of Gas pipelines, established by the following (such standards to apply in the following order of priority): (i) a Governmental Authority having jurisdiction over FLNG; and (ii) any nationally recognized non-governmental agency or organization with whose standards and practices it is customary for Reasonable and Prudent Operators of U.S. Gas pipelines to comply. In the event of a conflict between any of the priorities noted above, the priority with the lowest roman numeral noted above shall prevail.
- 1.76 "Port Charges" means all third-party charges of whatsoever nature (including rates, tolls, and dues of every description) in respect of an LNG Vessel entering or leaving the Freeport Facility, including charges imposed by fire boats, tugs and escort vessels, the

U.S. Coast Guard, a Pilot, and any other Person assisting an LNG Vessel to enter or leave the Freeport Facility. For purposes of this Agreement, the term "Port Charges" shall include, to the extent applicable to LNG Vessels, Port Use Fees (excluding rent) and Thru-put Fees (as such terms are defined under the Freeport Facility Lease as reflected on Exhibit A attached hereto) but shall exclude all other fees under the Freeport Facility Lease.

- **1.77** "psig" means pounds per square inch gauge.
- 1.78 "Reasonable and Prudent Operator" means a Person seeking in good faith to perform its contractual obligations, and in so doing, and in the general conduct of its undertaking, exercising that degree of skill, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled and experienced operator engaged in the same type of undertaking under the same or similar circumstances and conditions.
- 1.79 "Receipt Point" means the point at the Freeport Facility at which the flange coupling of the Freeport Facility's receiving line joins the flange coupling of the LNG unloading manifold on board an LNG Vessel.
- 1.80 "Regasified LNG" means Gas derived from the conversion of LNG (received by FLNG at the Receipt Point) from its liquid state to a gaseous state.
- **1.81** "Release Notice" shall have the meaning set forth in Section 6.2(b)(i)c.
- **1.82** "Reservation Fee" shall have the meaning set forth in Section 4.1(a).
- **1.83** "Retainage" shall have the meaning set forth in Section 4.2.
- **1.84 "Scheduled Unloading Window"** means, for any applicable Contract Year, an Unloading Window allocated either to Customer or any Other Customer pursuant to Article 5.
- 1.85 "Scheduling Period" means a Contract Year plus the three (3) month period immediately following such Contract Year.
- **1.86** "Scheduling Representative" means the individual appointed by Customer in accordance with Section 5.4.
- 1.87 "Services" shall have the meaning set forth in Section 2.1, as expanded from time to time in accordance with Section 2.2.
- **1.88** "Services Quantity" shall have the meaning set forth in Section 3.1(b).
- **1.89** "Services Unavailability" shall have the meaning set forth in Section 4.3(a).
- 1.90 "Sole Opinion" means an opinion, judgment or discretion of a Party that is not intended to be capable of being challenged in any legal or arbitral proceeding whatsoever.

- 1.91 "Standard Cubic Foot" means the quantity of Gas, free of water vapor, occupying a volume of one (1) Actual Cubic Foot at a temperature of 60.0 degrees Fahrenheit and at an absolute pressure of 14.696 pounds per square inch. For purposes of this Section 1.91, "Actual Cubic Foot" means a volume equal to the volume of a cube whose edge is one (1) foot.
- 1.92 "Storage" means the retention by FLNG of Customer's Inventory for a period of time in tanks or other facilities at the Freeport Facility (including the Freeport Facility Pipeline) or in a Gas Storage Facility.
- 1.93 "Storage Standards" means, to the extent not inconsistent with the express requirements of this Agreement, the standards and practices applicable to the design, equipment, operation or maintenance of Gas storage facilities, established by the following (such standards to apply in the following order of priority): (i) a Governmental Authority having jurisdiction over FLNG; and (ii) any nationally recognized non-governmental agency or organization with whose standards and practices it is customary for Reasonable and Prudent Operators of Gas storage facilities to comply. In the event of a conflict between any of the priorities noted above, the priority with the lowest roman numeral noted above shall prevail.
- 1.94 "Substantial Completion" means completion of the Freeport Facility so that it is ready to be used for its intended purpose, with (a) the contractor under the engineering, construction and procurement contract for the Freeport Facility having achieved all minimum acceptance requirements for "Substantial Completion" under such contract, (b) all Approvals necessary to commence operations of the Freeport Facility having been received by FLNG, and (c) the Freeport Facility being capable of performing the Services Quantity.
- 1.95 "Taxes" means all customs, taxes, royalties, excises, fees, duties, levies, sales and use taxes and value added taxes, charges and all other assessments, which may now or hereafter be enacted, levied or imposed, directly or indirectly, by a Governmental Authority, except Port Charges.
- 1.96 "Temporary Customer" shall have the meaning set forth in Section 6.2(a).
- 1.97 "Temporary Release" shall have the meaning set forth in Section 6.2(a).
- 1.98 "Temporary Release Inventory" means, at any given time, the quantity in MMBTUs that represents (a) the LNG received at the Receipt Point for Temporary Customer's account; and (b) Gas, whether or not such Gas is Regasified LNG or Gas derived or produced from sources other than Regasified LNG, held for Temporary Customer's account.
- **1.99** "**Term**" shall have the meaning set forth in Section 7.1.
- **1.100** "Transporter" means any Person who owns or operates an LNG Vessel.

- 1.101 "Unloading Window" means a forty-eight (48) hour window starting at 6:00 a.m., Central Time on a specified day and ending forty-eight (48) consecutive hours thereafter during which FLNG would make available berthing and LNG unloading services at the Freeport Facility to either Customer or one of the Other Customers.
- 1.102 "Variable Fee Component" shall have the meaning set forth in Section 4.1(a)(ii).

ARTICLE 2 SERVICES AND SCOPE

2.1 Services to be Provided by FLNG

During the Term, FLNG shall, acting as a Reasonable and Prudent Operator, make available the following services to Customer (such available services being herein referred to as the "Services") in the manner set forth in Article 3:

- (a) the berthing of LNG Vessels at the Freeport Facility;
- (b) the unloading and receipt of LNG from LNG Vessels at the Receipt Point;
- (c) the regasifying of LNG held in Storage;
- (d) Storage of Customer's Inventory;
- (e) the transportation and delivery of Gas in the Freeport Facility Pipeline to the Delivery Point (it being acknowledged that FLNG may, at its option, cause Gas to be redelivered to Customer from sources other than Regasified LNG); and
- (f) other activities directly related to performance by FLNG of the foregoing.

2.2 Additional Services

From time to time during the Term, the representatives of FLNG and Customer may supplement this Agreement in accordance with Section 27.1 to provide that FLNG will also make available services to Customer in addition to the Services set forth in Section 2.1.

2.3 Activities Outside Scope of this Agreement

For greater certainty, the Parties confirm that the following activities, *inter alia*, are not Services provided by FLNG to Customer and, therefore, such activities are outside of the scope of this Agreement:

- (a) harbor, mooring and escort services, including those relating to tugs, service boats, fire boats, and other escort vessels;
- (b) the construction, operation, ownership, maintenance, repair and removal of facilities downstream of the Delivery Point;

- (c) the transportation of Gas beyond the Delivery Point;
- (d) the marketing of Gas and all activities related thereto (except as expressly provided in Section 3.4); and
- (e) the removal, marketing and transportation of Liquids and all activities related thereto. For the avoidance of doubt, FLNG reserves the right to separate and/or extract Liquids from LNG upstream of the Delivery Point, provided that such separation does not result in Gas failing to meet the quality specifications at the Delivery Point required under Section 11.3.

ARTICLE 3 SALE AND PURCHASE OF SERVICES

3.1 Services Quantity

- (a) <u>Purchase and Sale of Services.</u> During each Contract Year, FLNG shall make available to Customer, and Customer shall purchase and pay for (in respect of the Payment Period) in an amount equal to the Fee, the Services Quantity.
- (b) <u>Services Quantity</u>. The quantity of Services FLNG shall make available to Customer (the "Services Quantity") during a Contract Year, and for which Customer shall purchase and pay for (in respect of the Payment Period) pursuant to Section 3.1(a), shall consist of the following:
 - (i) <u>Unloading of LNG</u>. FLNG shall make the Freeport Facility available during Unloading Windows to allow berthing, unloading and receipt of Customer's LNG in a quantity up to the Maximum LNG Reception Quantity;
 - (ii) <u>Storage of Customers' Inventory.</u> FLNG shall cause Customer's Inventory, net of Retainage, to be held temporarily in Storage until redelivered in accordance with Section 3.1(b)(iii) below; and
 - (iii) Redelivery of Gas at Delivery Point Subject to the provisions of this Agreement, including Sections 3.3(a), 3.3(b) and 3.3(c), FLNG shall, on a daily basis, make Gas from Customer's Inventory available to Customer at the Delivery Point at the rate nominated by Customer pursuant to Section 5.2(c), which nominated rate shall be no less than the *** and no more than the ***; provided, however, that FLNG shall, on a daily basis, make Gas in excess of the *** ("Peaking Gas") available to Customer at the Delivery Point in the quantities determined pursuant to Sections 5.2(a) and 5.2(b)(iv), but subject to the allocation under Section 3.3(d).
- (c) <u>Customer Option to Reduce Maximum LNG Reception Quantity.</u> No later than August 31, 2004, Customer shall, by providing written notice thereof (the "**Reduction Notice**") to FLNG, have the right to reduce the Maximum LNG Reception Quantity for each Contract Year to an amount no less than 97,637,501 MMBTUs.

- (d) <u>Expiration of Services Quantity</u>. Subject to Section 10.7, if Customer does not use any portion of the Services Quantity made available to Customer pursuant to the terms of this Agreement, including any portion of the Services Quantity not used in connection with a Temporary Release, Customer's right to such unused portion of the Services Quantity shall expire.
- (e) Excess Reception. Notwithstanding the provisions of Section 3.1(b)(i), in relation to any Contract Year, FLNG may, in its Sole Opinion, allow berthing, unloading and receipt of Customer's LNG in quantities in excess of the Maximum LNG Reception Quantity. Any such reception by FLNG of quantities in excess of the Maximum LNG Reception Quantity shall be subject to the Parties' prior agreement upon a temporary increase in the *** for a specified period of time in order to allow Customer to receive redelivery of such excess quantities within a reasonable time. The berthing, unloading and receipt of such quantities in excess of the Maximum LNG Reception Quantity shall not be construed to alter or release the obligations of the Parties under this Agreement, except that Customer shall pay the Excess Reception Fee for such excess quantities and except as otherwise provided in the preceding sentence.

3.2 Customer's Use of Services Quantity

Customer shall be entitled to use the Services Quantity in whole or in part by itself, or may contract with one or more third parties for a Temporary Release of such Services Quantity pursuant to the terms and conditions of Section 6.2.

3.3 Gas Redelivery

- (a) No Pre-Delivery Right. On any given day during a Contract Year, Customer shall not be entitled to receive quantities of Gas in excess of Customer's Inventory.
- (b) ***. For purposes of this Agreement, the term "***" means (i) the *** divided by (ii) ***.
- (c) ***. For purposes of this Agreement, the term "***" means, for any day in which Customer is nominating pursuant to Section 5.2(c) (a"Nomination Day"), the lesser of:
 - (i) (a) the sum of (x) ***, and (y) ***, <u>divided by</u> (b) the lesser of thirty (30) or the number of days from and including the Nomination Day until the last day of Customer's next Scheduled Unloading Window; or

- (ii) the "Alternate ***" as notified pursuant to Section 5.2(a), such amount being:
 - a. the sum of: (x) *** and, (y) ***; and (z) the aggregate of the following, calculated for each Cargo expected to arrive in such month at the Freeport Facility for Customer's account after the day in the relevant month that precedes the last day of such month by ***:

 (A) the quantity in MMBTUs of ***, divided by (B) *** and multiplied by (C) the number of ***, from and including the last day of the applicable Scheduled Unloading Window to and including the ***; divided by
 - b. ***.

provided, however, if the Expected Receipt Quantity for a particular Cargo scheduled to be delivered to the Freeport Facility in the following month exceeds *** MMBTUs, then FLNG shall have the right, in its notice under Section 5.2(a), to require *** to allow Other Customers to unload at the Freeport Facility their respective maximum LNG reception quantity on a *** over the Contract Year (such higher minimum rate being the "Large LNG Vessel Gas Redelivery Adjustment"), and further provided that if the application of the Large LNG Vessel Adjustment results in the *** exceeding the *** for a particular day, the *** shall automatically increase by the amount of such excess.

- (d) <u>Peaking Gas.</u> FLNG shall give due consideration to the notices provided to it by Customer and Other Customers electing to nominate quantities of Peaking Gas (the "Nominees") pursuant to Sections 5.2(c) and 5.2(d) and shall allocate such Peaking Gas among the Nominees as follows:
 - (i) First, to each Nominee in an amount not to exceed the lesser of (a) the quantity of Peaking Gas nominated by such Nominee, and (b) the product of (x) the quantity of ***, and (y) a fraction, the numerator of which is such Nominee's ***, and the denominator of which is the aggregate *** of all Nominees;
 - (ii) Second, if any excess quantity of Peaking Gas remains available, then to each Nominee with unfulfilled nominations for such Peaking Gas in an amount not to exceed the lesser of (a) the quantity of Peaking Gas nominated by such Nominee and not received under the prior allocation, and (b) the product of (x) the excess quantity of ***, and (y) a fraction, the numerator of which is the Nominee's ***, and the denominator of which is the aggregate *** of all Nominees with unfulfilled nominations for such ***; and
 - (iii) Third, if any excess quantity of Peaking Gas remains available, then by repeating the allocation in Section 3.3(d)(ii) until the entire quantity of Peaking Gas made available by FLNG has been allocated or all nominations for such Peaking Gas have been filled.

3.4 Failure to Take Delivery of Gas at Delivery Point

If on any day Customer fails materially to take redelivery of any Gas at the Delivery Point at the rate nominated by Customer pursuant to Article 5 and such failure is for reasons other than a Force Majeure or reasons attributable to FLNG, then FLNG may, at its option, take title to same free and clear of any Claims, and sell or otherwise dispose of such Customer's Inventory using good faith efforts to obtain the best available prices and to minimize costs. FLNG shall credit to Customer's account the net proceeds from the sale or other disposition of Customer's Inventory to which it takes title hereunder, minus actual and reasonable transportation costs, third-party charges, and an administrative fee of \$0.05 per MMBTU; provided, however, that if the amount of the credit exceeds the amount due to FLNG under the next monthly statement, then FLNG agrees to pay any such excess amount to Customer within five (5) Business Days after delivery of such monthly statement.

3.5 Freeport Services Manual

Acting as a Reasonable and Prudent Operator, FLNG shall develop and maintain a single services manual applicable to Customer and all Other Customers which contains detailed implementation procedures necessary for performance of this Agreement and agreements with Other Customers with regard to the matters set forth in Exhibit B attached hereto (but excluding the matters governed by the Freeport Facility Marine Operations Manual). In developing such a manual, FLNG shall provide Customer with a preliminary draft of the same (the "Preliminary Services Manual"). If Customer desires to consult with FLNG regarding the contents of the Preliminary Services Manual, Customer shall, no later than fifteen (15) days from delivery of said manual by FLNG, request to meet with FLNG by providing notice thereof to FLNG, and FLNG shall, no later than thirty (30) days after receipt of such notice, meet with Customer to discuss said manual. If (a) Customer does not submit the foregoing notice to FLNG on a timely basis or (b) Customer and FLNG meet pursuant to such a notice and are able during such meeting to agree upon revisions to the draft, then such draft, as so revised (and as amended from time to time), shall constitute the "Freeport Services Manual". If Customer and FLNG meet pursuant to the foregoing notice and are unable during such meeting to agree upon revisions to the Preliminary Services Manual, then FLNG shall determine, while using its reasonable efforts to accommodate Customer's views, the Freeport Services Manual. In the event FLNG intends to amend the Freeport Services Manual, then FLNG shall follow the procedure set forth above in relation to the Preliminary Services Manual. FLNG shall deliver to Customer and all Other Customers a copy of the Freeport Services Manual and any amendments thereto promptly after they have been finalized or amended, as the case may be. The Parties shall comply with the Freeport Services Manual in all material respects. FLNG will undertake to develop a Freeport Services Manual that is co

ARTICLE 4 COMPENSATION FOR SERVICES

4.1 Fee

Subject to the adjustments provided in Sections 4.3 and 4.7, Customer shall, as full compensation for the performance by FLNG of its obligations under this Agreement, bear the Retainage and pay to FLNG the sum of the following components (such sum collectively referred to as the "Fee") in respect of the period from the later of the Commercial Start Date or Substantial Completion until the end of the Term (the "Payment Period"):

- (a) <u>Reservation Fee</u>. A monthly reservation fee (the "Reservation Fee") consisting of the following:
 - (i) a fixed amount (the "Fixed Fee Component") payable in arrears and equal to the Maximum LNG Reception Quantity divided by the Monthly Component, multiplied by Input A; and
 - (ii) a variable amount (the "Variable Fee Component") payable in arrears and equal to *** multiplied by the Maximum LNG Reception Quantity divided by the Monthly Component; provided that:
 - a. if the Henry Hub Price is greater than *** but less than or equal to ***, the Variable Fee Component shall be the product of (x) *** plus *** of the incremental amount by which the Henry Hub Price is greater than ***; and (y) the Maximum LNG Reception Quantity divided by the Monthly Component; or
 - b. if the Henry Hub Price is greater than ***, the Variable Fee Component shall be *** multiplied by the Maximum LNG Reception Quantity divided by the Monthly Component;
- (b) <u>Incremental Costs</u>. The following incremental costs (the "Incremental Costs") payable in arrears:
 - the actual electric power cost associated with the Services provided hereunder to Customer, calculated as an amount equal to (x) all electric power costs incurred at the Freeport Facility and all Gas Storage Facilities, during the applicable period multiplied by (y) a fraction, the numerator of which is Customer's LNG received at the Freeport Facility during such period and the denominator of which is the quantity of LNG received at the Freeport Facility during such period for the account of Customer and all Other Customers (provided, however, that FLNG shall use reasonable efforts to (x) minimize all such electric power costs, and (y) obtain the best commercial rates for electric services available at the location of the Freeport Facility given the intended use of the Freeport Facility and its twenty-four (24) hour operation); provided, however, that the amount

- payable by Customer for electric power cost associated with Gas Storage Facilities shall not, in any Contract Year, exceed ***/MMBTU multiplied by the Maximum LNG Reception Quantity;
- (ii) the actual amount of all reasonable incremental direct costs, if any, incurred by FLNG for berthing an LNG Vessel after sunset at the Freeport Facility;
- (iii) the actual amount of all reasonable incremental direct costs, if any, incurred by FLNG when an LNG Vessel arrives more than three (3) hours after the ETA set forth in the Final Notice delivered pursuant to Section 9.4(c)(iv), provided that if such a delay is directly caused by a Governmental Authority or a Force Majeure or Adverse Weather Conditions, then fifty percent (50%) of such reasonable incremental direct costs; and
- (iv) excess berth fees, if any, under Section 9.9(b)(iii); and
- (c) Excess Reception Fee. An excess reception fee (the "Excess Reception Fee") if, in any Contract Year, FLNG receives quantities of LNG for Customer's account in excess of the Maximum LNG Reception Quantity, payable in arrears upon Customer exceeding the Maximum LNG Reception Quantity. In any Contract Year that Customer exceeds the Maximum LNG Received Quantity, then Customer shall have the option to carry forward into the next Contract Year one-half of the last Cargo without incurring the Excess Reception Fee for such carry-forward quantity. The Excess Reception Fee shall equal (i) the amount by which *** for such Contract Year (ii) multiplied by a fee equal to the sum of *** divided by *** for such Contract Year.

4.2 Retainage

For purposes of this Agreement, the term "Retainage" means the aggregate of *** of Customer's LNG, to be retained by FLNG as compensation for fuel for the Freeport Facility.

4.3 Services Unavailability

- (a) Meeting. If some or all of the Services are unavailable to Customer on any day (or portion of a day) during the Term (collectively referred to as Services Unavailability") as a result of (x) a failure of FLNG (including an unexcused failure of FLNG), (y) Force Majeure of FLNG, or (z) a curtailment or temporary discontinuation of Services pursuant to Section 18.2 but not Section 18.1, then the Parties agree ***.
- (b) Expert Determination. If, within *** following such meeting, the Parties are unable to agree upon a means by which to restore the Services lost during such Services Unavailability and/or to reduce the Fee during such Services Unavailability, the Parties shall immediately submit their Dispute to an expert

- pursuant to the provisions of Section 23.2 for determination by such expert of an amount by which the Fee shall be reduced as a result of the Services Unavailability, the intent being that Customer will not pay for Services to the extent not restored.
- (c) <u>Litigation</u>. Notwithstanding any provision herein to the contrary, including Sections 4.3(b) and 23.2, if a Party does not agree with the expert's final decision, then such Party may refer the Dispute to litigation pursuant to Article 23 *** of the date the expert's final decision is received by the Parties. If the expert's determination is not challenged in any such litigation within ***, then such determination shall become final and binding. In any such litigation the expert determination on the specific matter shall be entitled to a rebuttable presumption of correctness.
- (d) Other Remedies. For the avoidance of doubt, the provisions of this Section 4.3 are without prejudice to any other rights or remedies of Customer arising hereunder or by law or otherwise, including the right to recover damages or pursue other remedies that Customer may have as a result of such Services Unavailability.

4.4 Construction Cost Deviation Amount and Additional Capital Costs

- (a) Construction Cost Deviation Amount. The Fixed Fee Component is based upon estimated expenditures of *** for FLNG's construction of the Freeport Facility as such facility is approved by FERC in the order granting authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act. Input A shall be adjusted, upwards or downwards as the case may be, by ***/MMBTU for each *** (or fraction thereof) deviation in FLNG's actual cost of constructing the Freeport Facility (pursuant to the application filed by FLNG with FERC on March 28, 2003) as compared to such estimated expenditures of *** (such adjustment being the "Construction Cost Deviation Amount"); provided that:
 - (i) after issuance by FERC of the order granting authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act (pursuant to the application filed by FLNG with FERC on March 28, 2003), the Construction Cost Deviation Amount shall be based on the positive or negative difference between *** and the total lump sum turnkey price due to the EPC contractor from FLNG under the engineering, procurement and construction contract for the Freeport Facility as of the date of execution of such contract (including any amounts due as a result of any change orders thereunder to the extent necessary to comply with such FERC order pursuant to the application filed with FERC on March 28, 2003);
 - (ii) Input A shall not be adjusted in relation to any costs for any dredging of the main ship channel beyond the scope of that anticipated in the application filed by FLNG with FERC on March 28, 2003; and

- (iii) the Construction Cost Deviation Amount shall not increase the Input A to an amount above ***/MMBTU or decrease the Input A to an amount below ***/MMBTU
- (b) Additional Capital Costs. If (i) FLNG is required to incur any capital expenditure to comply with any law, rule or regulation imposed by FERC or any Governmental Authority subsequent to (x) approval by FERC of FLNG's application for authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act and (y) execution by FLNG of the engineering, procurement and construction contract for the Freeport Facility so authorized and (ii) such expenditure is beyond those originally foreseen as of the Effective Date ("Additional Capital Costs"), then (a) the Input A shall be increased by an amount equal to *** for each *** (or fraction thereof) of Additional Capital Costs and (b) FLNG shall require comparable incremental payments from Other Customers, in each case until such time as the Additional Capital Costs and finance costs associated therewith have been fully recouped by FLNG through such incremental payments by Customer and by comparable incremental payments from Other Customers. For the avoidance of any doubt, any item payable or excluded under Section 4.4(a) shall be outside the scope of this Section 4.4(b).
- (c) <u>Limitation</u>. Notwithstanding Sections 4.4(a) and 4.4(b) or any other provision of this Agreement, neither the application of the Construction Cost Deviation Amount nor the application of Additional Capital Costs shall increase the Reservation Fee to the extent that such increase would cause the quotient of
 - (i) the Reservation Fee (excluding any adjustment in Section 4.7); and
 - (ii) the Maximum LNG Reception Quantity <u>divided by</u> the Monthly Component, to exceed *** per MMBTU.

4.5 New Taxes

(a) Allocation. If, subsequent to the Effective Date, any Governmental Authority enacts any new Taxes on FLNG, the Services, or the Freeport Facility (*New Taxes"), then the Parties shall meet with a view to agreeing on amendments to this Agreement with respect to the equitable allocation of such New Taxes by and among the Parties and Other Customers. For purposes of this Agreement, New Taxes shall be considered to exclude Taxes on the capital, revenue, or income derived by FLNG, sales or use Taxes on construction of the Freeport Facility, an increase in existing Taxes, payments due under the Freeport Facility Lease, and any Taxes on real or personal property owned or leased by FLNG. If the Parties fail to reach agreement within ninety (90) days of the commencement of such negotiations, the Parties shall within thirty (30) days thereafter appoint an expert pursuant to Article 23 to allocate the effect of New Taxes between the Parties and

Other Customers. The expert shall determine a method by which the effects thereof may be equitably allocated among the Parties and the Other Customers. The expert shall be authorized to modify this Agreement in accordance with the expert's resolution of such allocation. The terms "equitable allocation" or "equitably allocated" under this Section 4.5 shall take into account: (a) the timing of when FLNG incurs New Taxes, (b) if there has been a reduction of Taxes and a concurrent enactment of New Taxes, which Party currently bears the risk of Taxes under this Agreement, (c) that Customer is paying, as part of the Fee, *** per MMBTU in relation to property taxes on the Freeport Facility, (d) the portions of LNG terminalling services contracted by Customer and by Other Customers with FLNG, and (e) the remaining duration of the Term and of the term of the terminal use agreement with each Other Customer. In guidance to the amendment of the Agreement, any New Taxes allocated to Customer hereunder shall be treated as part of the Fee for the purposes of Sections 4.3 and 4.7(b).

(b) Tax Proceedings. In the event FLNG or any of its Affiliates receives notice of any examination, claim, assessment, adjustment or other proceeding relating to the liability for New Taxes, FLNG shall notify Customer in writing within twenty (20) days of receiving notice thereof. The Parties shall cooperate with each other and with their respective Affiliates in the negotiations and settlement of any such examination, claim, assessment, adjustment or other proceeding.

4.6 Services Provided to Other Customers

- (a) <u>Identity of Other Customers</u>. FLNG shall from time to time inform Customer of the identity of all Other Customers who have signed with FLNG terminal use agreements having a term of at least *** (***) years.
- (b) <u>Terminal Use Agreements with Other Customers.</u> In its negotiation of terminal use agreements with Other Customers, FLNG shall use best efforts to include shipping, scheduling and operational provisions that are consistent in all material respects with the provisions in Article 5, Article 8, Article 9, Article 10, Article 11, and Article 17 herein.

4.7 Certain Adjustments to Reservation Fee

- (a) Reservation Fee Rebate. For purposes of Section 4.1, FLNG shall credit the Customer's statement for which payment is due on *** with an amount determined as follows (such amount the "Reservation Fee Rebate Amount"):
 - (i) ***
 - (ii) ***
 - (iii) ***
 - (iv) ***
 - (v) ***

- (b) Adjustment during Extension Terms. Notwithstanding any provision to the contrary in Section 4.1 and this Agreement, during any Extension Term, the Parties shall adjust the *** such that the Fee equals the then current market rates for similar LNG tolling services in the United States; provided, however, that the *** for the Extension Term, as adjusted, shall not exceed the *** applicable to the Initial Term and shall not be less than ***. If the Parties are unable to agree upon the then current market rates in the United States no later than *** prior to the expiration of the then current Term, the Parties agree to submit the Dispute to an expert pursuant to Section 23.1. If the Parties are able to agree to such adjustment of the ***, the Parties shall execute a supplement to this Agreement pursuant to Section 27.1. Notwithstanding any provision to the contrary in Section *** and this Agreement, during any Extension Term, the Parties shall adjust the *** set forth in Section *** such that the ***.
- (c) No Other Charges or Fee Increases. There shall be no charges due from Customer to FLNG nor any increases in the Fee unless specifically authorized by this Agreement.

ARTICLE 5 SCHEDULING

5.1 Customer LNG Receipt Schedule

- (a) FLNG Deliverables. Not later than one hundred twenty (120) days prior to the beginning of each Scheduling Period, FLNG shall provide to the Scheduling Representative a non-binding written assessment of the dates of any planned maintenance to or modifications of the Freeport Facility for such Scheduling Period and the expected impact of such activities on the availability of Services. FLNG shall limit the number of days of any planned maintenance to or modifications of the Freeport Facility and shall meet the requirements of Article 18.
- (b) Notice from Scheduling Representative. Not later than one hundred five (105) days prior to the beginning of each Scheduling Period, the Scheduling Representative shall notify FLNG of the following:
 - (i) a programming schedule for the unloading of (x) up to the Maximum LNG Reception Quantity over the course of the next Contract Year as well as (y) up to *** percent (***%) of the Maximum LNG Reception Quantity for the next succeeding Contract Year (such percentage to be adjusted to reflect any partial Contract Year) over the course of the last three months of the Scheduling Period, which schedule shall specify, for each Unloading Window, the proposed arrival date (the "Arrival Date") of the applicable LNG Vessel and which Arrival Date must (x) result in a delivery pattern whereby deliveries in any given month do not materially

- exceed ***, (y) result in deliveries in any given month ***, and (z) take into consideration the planned maintenance and modification dates furnished to Customer by FLNG as set forth in Section 5.1(a); and
- (ii) for each Arrival Date proposed pursuant to Section 5.1(b)(i), the name of the LNG Vessel expected to deliver LNG to the Freeport Facility (if the identity of the LNG Vessel is known to Customer at such time), the Expected Receipt Quantity, and the anticipated quality (expressed in terms of Gross Heating Value) of the LNG to be delivered at the Receipt Point during the Scheduling Period.
- (c) Notices from Other Customers. Customer acknowledges that Other Customers will submit similar notices to FLNG regarding the matters provided for in Section 5.1(b).
- (d) Preliminary Receipt Schedule. *** shall take into consideration the notices that it receives from the Scheduling Representative and the Other Customers and, not later than ninety (90) days prior to the beginning of each Scheduling Period, *** shall issue to *** via the *** (or via an alternative *** if the *** is unavailable) a preliminary receipt schedule for such Scheduling Period (the "*** Preliminary Receipt Schedule"). In issuing a *** Preliminary Receipt Schedule for a particular Scheduling Period, *** shall not alter, absent ***'s request, any Scheduled Unloading Window allocated to *** for the last *** months of the Customer LNG Receipt Schedule for the prior Scheduling Period. Customer may propose to FLNG to change any such Scheduled Unloading Window, and FLNG agrees to give due consideration to, and use reasonable efforts to accommodate, such change. The *** Preliminary Receipt Schedule shall set forth (i) *** and (ii) *** for purposes of changes pursuant to Section 5.1(i).
- (e) Other *** Preliminary Receipt Schedules and Mutual Cooperation. *** acknowledges that *** will issue to *** via the *** a preliminary receipt schedule similar to the *** Preliminary Receipt Schedule described in Section 5.1(d), but customized for each such *** ("*** Preliminary Receipt Schedules").

 *** also acknowledges that conflicts will occur in the preparation of the Preliminary Receipt Schedule and *** Preliminary Receipt Schedules because of the joint use of the Freeport Facility among Customer and Other Customers. Accordingly, the Parties agree to cooperate with each other to resolve any such conflict.
- (f) Consultation; Customer LNG Receipt Schedule. If the Scheduling Representative desires to consult with *** regarding the contents of the *** Preliminary Receipt Schedule, the Scheduling Representative shall, no later than fifteen (15) days from the issuance of the *** Preliminary Receipt Schedule, request to meet with *** by providing notice thereof (the "Consultation Notice") to ***, and *** shall, no later than fifteen (15) days after receipt of the Consultation Notice, meet with the Scheduling Representative to discuss the *** Preliminary Receipt Schedule. If (i) the Scheduling Representative does not submit a Consultation

Notice to *** on a timely basis or (ii) the Scheduling Representative and *** meet pursuant to a Consultation Notice and are able during such meeting to agree upon revisions to the *** Preliminary Receipt Schedule, then such *** Preliminary Receipt Schedule, as so revised (and as updated from time to time for such Scheduling Period by *** via the ***, such updates to be made in accordance with this Agreement and the Freeport Services Manual) shall constitute the "Customer LNG Receipt Schedule". If the Scheduling Representative and *** meet pursuant to a Consultation Notice and are unable during such meeting to agree upon revisions to the *** Preliminary Receipt Schedule, then *** shall determine, while using its reasonable efforts to accommodate Customer's views, the Customer LNG Receipt Schedule with the understanding that, for purposes of such determination, no Major Customer shall be given any preference in scheduling over any other Major Customer but Major Customers shall be given preferential consideration in scheduling over Non-Major Customers. FLNG shall issue via the *** (or via an alternative *** if the *** is unavailable) the Customer LNG Receipt Schedule no later than sixty (60) days prior to the first day of the Scheduling Period. The Customer LNG Receipt Schedule shall set forth (i) *** and (ii) ***.

- (g) Other Customer LNG Receipt Schedules. Customer acknowledges that FLNG shall issue to each Other Customer a final receipt schedule similar to the Customer LNG Receipt Schedule described in Section 5.1(f) but customized for each such Other Customer (such schedules referred to as "Other Customer LNG Receipt Schedules").
- (h) Adjustment to Scheduling Periods. Upon written request by the Customer, FLNG shall use reasonable efforts to modify the time periods expressly set forth in Sections 5.1(a), 5.1(b), 5.1(d), and 5.1(f) to allow Customer to interface these periods with corresponding time periods for scheduling agreed upon by Customer and its LNG Suppliers. For purposes of this Section 5.1(h), FLNG shall be deemed to have used reasonable efforts if FLNG rejects Customer's request because it determines, acting as a Reasonable and Prudent Operator, that any such modification would infringe on the rights of Other Customers.
- (i) <u>Customer Changes to Customer LNG Receipt Schedule</u>. The Parties agree as follows:
 - (i) Subject to the terms of this Section 5.1(i), at any time following the issuance of the Customer LNG Receipt Schedule, the Scheduling Representative may submit to FLNG a written request to change a Scheduled Unloading Window to an Unloading Window that is not presently allocated to Customer or Other Customers and which FLNG is making available to Customer and Other Customers under the Customer LNG Receipt Schedule (such request to change, a "Customer Open Window Request"). Customer understands that (x) Other Customers shall also have the right to submit to FLNG similar scheduling requests (each an "Other Customer Open Window Request"), (y) ***, and

- (z) *** as soon as possible but not later than 5:00 p.m., Central Time of the Business Day following the date of receipt by FLNG of the applicable Open Window Request. Upon accepting an Open Window Request, FLNG shall notify Customer and Other Customers thereof via the *** (or via an alternative *** if the *** is unavailable). Notwithstanding anything herein to the contrary, Customer shall use its reasonable efforts to keep to a minimum the number of Customer Open Window Requests it submits to FLNG.
- (ii) Subject to the terms of this Section 5.1(i), at any time following the issuance of the Customer LNG Receipt Schedule, the Scheduling Representative may submit to FLNG a written request to change a Scheduled Unloading Window to a forty-eight (48) hour period that is unavailable to Customer under the Customer LNG Receipt Schedule (such change, a "Change Request"). Customer acknowledges that any Change Request will (x) *** and (y) ***. Accordingly, FLNG shall *** any Change Request and shall notify Customer thereof via the *** (or via an alternative *** if the *** is unavailable) within three (3) Business Days of its receipt of a Change Request. Notwithstanding anything herein to the contrary, Customer shall use *** to keep to a minimum the number of Change Requests it submits to FLNG, and FLNG shall use its reasonable efforts to accommodate Customer's Change Requests.
- (j) Other Modifications Due to Services Unavailability. If, for any Scheduled Unloading Window, Customer is unable, due to a Services Unavailability, to berth and unload an LNG Vessel, each affected Scheduled Unloading Window allocated to Customer during such period shall be ***, to the extent affected. In the event a Scheduled Unloading Window is so cancelled, such cancellation shall be without prejudice to Customer's rights and remedies hereunder. Except as otherwise provided in Section 5.1(i), the Customer LNG Receipt Schedule shall be considered firm and shall not be subject to change by FLNG; provided, however, that FLNG ***, with preference to Major Customers, if such Services Unavailability caused the *** of one or more Scheduled Unloading Windows allocated to Customer and/or Other Customers, in order to maximize efficient usage of the Freeport Facility to assist Customer and Other Customers to unload quantities of LNG which would otherwise have been unloaded at the Freeport Facility during such cancelled Scheduled Unloading Windows. Moreover, in the event of a Services Unavailability, FLNG shall make reasonable efforts to *** of redelivery of Gas for Customer and Other Customers to maximize efficient usage of the Freeport Facility to assist Customer and Other Customers to *** which would otherwise have been received at the Delivery Point during such Services Unavailability.

5.2 Gas Delivery Procedure

(a) <u>Preliminary Nomination Schedule</u>. Not later than the fifteenth (15th) day of each month, commencing the month immediately prior to the Commercial Start Date.

FLNG shall provide to the Scheduling Representative a nomination schedule (the "Preliminary Nomination Schedule") that sets forth, for each day of the ensuing month, the following:

- (i) the ***; and
- (ii) all *** which, at the time of notification of the Preliminary Nomination Schedule, FLNG acting as a Reasonable and Prudent Operator *** from the Freeport Facility; provided, however, that FLNG shall not be obligated to offer or deliver *** to Customer or Other Customers from Gas Storage Facilities or to the extent that a Reasonable and Prudent Operator would not obligate itself to do so under similar circumstances and conditions

For the avoidance of doubt, FLNG shall be obligated to make available to Customer the quantities of *** notified in the Preliminary Nomination Schedule, as allocated pursuant to Section ***.

- (b) <u>Daily Records</u>. Commencing on the Commercial Start Date, FLNG shall, on each Business Day by the time specified in the Freeport Services Manual, post on the *** for access by Customer certain daily records (the "Daily Records"), including the following:
 - (i) Customer's Inventory held as of 11:59 p.m., Central Time on the day preceding the posting of the Daily Records;
 - (ii) the estimated quantity of Customer's Inventory expected to be held in Storage as of 11:59 p.m., Central Time on the Business Day on which the Daily Records are posted, taking into account expected receipts and deliveries during such Business Day;
 - (iii) if applicable, the estimated quantity of Customer's Inventory expected to be held in Storage as of 11:59 p.m., Central Time on each non-Business Day occurring between the Business Day on which the Daily Records are posted and the next Business Day, taking into account expected receipts and deliveries during such Business Days; and
 - either (x) the total quantity of ***, if any, which FLNG acting as a Reasonable and Prudent Operator has determined on such Business Day will be *** on (a) the Business Day or (b) the Business Day following the posting of the Daily Records or (y) if such posting falls on a Business Day that precedes any non-Business Day, the total quantity of ***, if any, which FLNG acting as a Reasonable and Prudent Operator has determined on such Business Day ***; provided, however, that FLNG shall not be obligated to offer or deliver *** to Customer or Other Customers from Gas Storage Facilities or to the extent that a Reasonable and Prudent Operator would not obligate itself to do so under similar circumstances and conditions, as long as the total quantity posted on the Daily Records is not less than those provided pursuant to Section 5.2(a)(ii).

- (c) Gas Nomination. Commencing on the Commercial Start Date, the Scheduling Representative shall, on each Business Day by the time specified in the Freeport Services Manual, nominate the quantities of Gas (including Peaking Gas, if available) that Customer desires to be delivered to it at the Delivery Point on the next Business Day and any intervening days that are not Business Days (such next day(s) hereinafter referred to as a "Delivery Date") by providing notice thereof (the "Nomination Notice") to FLNG. Subject to Section 3.3(a), the quantities nominated by the Scheduling Representative in a Nomination Notice shall in no event be less than the *** nor more than the ***, except in the case of nominations of *** where the Scheduling Representative must, in the Nomination Notice, request the *** as well as the quantities of *** requested by Customer for delivery. In the event FLNG does not receive a Nomination Notice on a timely basis, the Scheduling Representative shall be deemed to have nominated the *** unless Customer has nominated in writing for multiple days a higher amount not to exceed the ***.
- (d) Variations in Daily Redelivery Rates. FLNG shall use reasonable efforts to designate in the Freeport Services Manual certain blocks of time during a day on which Customer and Other Customers may elect to vary the rates by which FLNG is to redeliver Gas nominated for a given day without exceeding the total quantity of Gas nominated. If FLNG chooses to designate such blocks of time and Customer desires to vary its rates, Customer shall, in a Nomination Notice, set forth the specific quantities of Gas to be redelivered by FLNG during such blocks. The Freeport Services Manual shall include provisions which ensure that, if FLNG chooses to designate such blocks of time, Customer shall be allocated at least its pro-rata share of such variable rates in daily Gas redelivery.
- (e) Other Customer Nomination Notices. Customer acknowledges that Other Customers shall provide to FLNG notices similar to the Nomination Notice described in Section 5.2(c).
- (f) Allocation of ***; Peaking Notice. Customer understands that if quantities of *** are made available to Customer and Other Customers pursuant to Sections 5.2(a) and 5.2(b)(iv), each of Customer and the Other Customers shall have the option to nominate all or any portion of such quantities through the notices provided for in Sections 5.2(c) and 5.2(d). FLNG shall allocate such *** as provided in Section *** and, for any given day, shall notify Customer of the quantities of *** allocated to Customer within the time specified in the Freeport Services Manual.

5.3 Standard

FLNG shall act as a Reasonable and Prudent Operator in performing the scheduling activities required by this Article 5.

5.4 Scheduling Representative

By no later than one month prior to the Commercial Start Date, Customer shall appoint an individual to act as Scheduling Representative for the purposes of this Article 5; provided, however, that Customer shall have the right to change the identity of the Scheduling Representative at any time by notice to FLNG. Unless otherwise stated herein, Customer hereby authorizes the Scheduling Representative to do and perform any and all acts for and on behalf of Customer with regard to scheduling matters provided for in this Article 5.

5.5 Scheduling Coordination Among Customer and Other Customers

Customer shall have the right to request FLNG to arrange a joint meeting with Other Customers with respect to any matter in relation to the performance of this Article 5. FLNG shall use reasonable efforts to organize such a meeting, provided that FLNG may elect to include additional Other Customers if ***. If the Other Customers invited by FLNG agree to participate in such a joint meeting between Customer, Other Customers and FLNG, the joint meeting shall be held as soon as possible. Unless otherwise agreed, any such joint meeting shall be held in Houston, Texas or by telephone, as appropriate.

ARTICLE 6 RELEASE OF SERVICES

6.1 General

Customer may assign (a) all or a part of the Services Quantity as a Temporary Release in accordance with Section 6.2, or (b) all or a part of the Services Quantity in accordance with Article 19.

6.2 Temporary Release

- (a) General. Customer may from time to time assign part of the Services Quantity in writing to a third party (a"Temporary Customer") on a temporary basis for not more than the remainder of the then existing Scheduling Period (each such partial assignment referred to herein as a "Temporary Release"). Customer shall have the right to have occurring at any given time up to *** (***) Temporary Release for each *** MMBTUs of Maximum LNG Reception Quantity.
- (b) <u>Conditions</u>. A Temporary Release shall be subject to the following conditions:
 - (i) Notice and Consent. No Temporary Release shall be permitted, or shall become effective, unless and until:
 - The proposed Temporary Release is consistent with the terms and conditions of this Agreement;

- b. The Services Quantity that Customer seeks to assign by way of a proposed Temporary Release includes only Scheduled Unloading Windows under the Customer LNG Receipt Schedule;
- c. Customer has (x) delivered to FLNG a written notice in the form set forth in the Freeport Services Manual (a'Release Notice") disclosing in sufficient detail the terms and conditions of the proposed Temporary Release relevant to the Services Quantity for FLNG to be able to carry out its obligations under this Article 6, including the proposed effective date and expiration date of the Temporary Release but excluding any sensitive pricing information related thereto, and (y) furnished to FLNG all information reasonably requested by FLNG with respect to such Temporary Release to the extent necessary for FLNG to carry out its obligations under this Agreement; and
- d. Except as otherwise provided below, FLNG has consented to the Temporary Release by executing the Release Notice, such consent not to be withheld provided Customer complies in all respects with the provisions of Sections 6.2(b)(i)a through 6.2(b)(i)c.
- (ii) Temporary Releases for Subsequent Scheduling Periods Notwithstanding the requirement in Section 6.2(a) that a proposed Temporary Release relate to the then existing Scheduling Period, in order to assist Customer in its long-term business planning, Customer shall be entitled to submit to FLNG from time to time Temporary Releases for subsequent Scheduling Periods. Subject to the same consent rights (if any) set forth in Section 6.2(b)(i)d, FLNG shall execute a Release Notice for such Temporary Release for a subsequent Scheduling Period.
- (iii) Authorization. Following the execution by FLNG of the Release Notice, FLNG shall be authorized to perform the specified portion of the Services Quantity for Temporary Customer, subject to the provisions of this Agreement. In relation to a Temporary Release, Temporary Customer shall be fully authorized to act on behalf of Customer, and FLNG shall be entitled to rely on the nominations, notices and other submissions communicated to FLNG by the Temporary Customer in relation to the Temporary Release as if such nominations, notices and other submissions had been made by Customer itself. In the event of a conflict between the terms of the nominations, notices and other submissions issued by the Temporary Customer in relation to the Temporary Release and those issued by Customer, the terms issued by Customer shall control
- (iv) No Effect on ***. For the avoidance of doubt, a Temporary Release shall in no event increase the ***.

- (v) <u>Performance</u>. Commencing on the Release Date and continuing through the expiration or termination of the Temporary Release, (i) FLNG shall perform the Services assigned to Temporary Customer under the Temporary Release, and (ii) Customer shall cause Temporary Customer to perform, for the benefit of FLNG, all requirements and obligations of Customer under this Agreement in relation to the Temporary Release.
- (vi) Expiration of Temporary Release. Except as otherwise provided herein, FLNG shall render Services to Temporary Customer for the period set forth in the Temporary Release (such period herein referred to as the "Temporary Term"). Customer's Nomination Notices shall ensure that the Temporary Release Inventory is reduced to zero (0) MMBTUs as of the last day of the Temporary Term. In the event that the Temporary Release Inventory is not reduced to zero (0) MMBTU on last day of the Temporary Term, the Temporary Release Inventory remaining thereafter shall be deemed solely as Customer's Inventory.
- (vii) Remedial Actions. In the event Temporary Customer's actions are materially inconsistent with the requirements of this Agreement, FLNG shall provide written notice thereof to Customer. Customer shall (x) inform FLNG within five (5) days of its receipt of FLNG's notice of the remedial actions it intends to take to cause Temporary Customer's actions to be materially consistent with the requirements of this Agreement, and (y) cause Temporary Customer to be in compliance herewith within thirty (30) days after delivery of FLNG's notice. In the event Customer is unable to cause Temporary Customer to be in compliance herewith, FLNG may terminate the Temporary Release by written notice to Customer.
- (viii) Responsibility. Customer shall ensure that each Temporary Release is performed in a manner consistent with the Release Notice and the terms and provisions of this Agreement. Notwithstanding anything in this Article 6 to the contrary, FLNG shall invoice Customer in accordance with the provisions of Section 12.1 for the Fee attributable to a Temporary Release, and Customer shall pay, or cause to be paid, the Fee attributable to a Temporary Release. No Temporary Release or anything in this Section 6.2 shall relieve Customer or FLNG of any responsibility or liability under this Agreement. Customer shall remain liable to FLNG for all obligations of Customer and Temporary Customer in connection with any Temporary Release. FLNG shall remain liable to Customer for all obligations of FLNG in connection with any Temporary Release.
- (ix) No Third-Party Beneficiary. A Temporary Customer is not intended to be, and shall not be construed to be, a third-party beneficiary of this Agreement, nor shall a Temporary Release, or anything contained in this Agreement, create any contractual or quasi-contractual relationship or obligation between any Temporary Customer and FLNG.

ARTICLE 7 TERM

7.1 Term

- (a) <u>General</u>. Subject to the provisions of this Agreement, including Article 20, the term of this Agreement (the **Term**) shall consist of the Initial Term and, if applicable, any Extension Term.
- (b) <u>Initial Term</u>. The initial term of this Agreement (the "**Initial Term**") shall commence on the Commercial Start Date, and shall continue thereafter until the day immediately preceding the twentieth (20th) anniversary thereof.
- (c) <u>Extension Term.</u> Except as otherwise provided herein, at the expiration of the Initial Term, Customer shall have the right to elect to extend the term of this Agreement as follows (each an "Extension Term"):
 - (i) an extension term which shall commence on the day immediately following the expiration of the Initial Term and continue thereafter until February 28, 2033 (the day on which the initial term of the Freeport Facility Lease expires); and
 - (ii) in addition to the extension right under (i) above, up to two additional ten (10) year extension terms from February 28, 2033, the first of which shall commence on March 1, 2033 and continue until February 28, 2043, and the second of which shall commence on March 1, 2043 and continue until February 28, 2053.

If Customer desires to extend this Agreement by an Extension Term, Customer must (i) notify FLNG of its good faith desire to elect the applicable Extension Term at least one thousand eight hundred twenty-five (1,825) days prior to the expiration of the then current Term, and (ii) no later than one thousand four hundred sixty (1,460) days prior to the expiration of the then current Term send FLNG a binding confirmation ("Binding Confirmation") that the term is extended by an Extension Term. Upon Customer's delivery of a Binding Confirmation to FLNG, this Agreement will then be automatically extended for the applicable Extension Term.

7.2 Commencement of Deliveries

In accordance with the procedure set forth in this Section 7.2, FLNG shall notify Customer of the date on which Services for Customer will commence at the Freeport Facility (the final date so notified being the "Commercial Start Date"). The Commercial Start Date shall be a date within the period from April 1, 2007 to March 31, 2008 (such period being the "First Window Period"). The First Window Period shall be narrowed pursuant to the following provisions:

(a) No later than December 1, 2006, FLNG shall notify Customer of a two hundred seventy (270) day window ("Second Window Period") falling within the First Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Second Window Period shall be the latest possible two hundred seventieth (270th) day window period within the First Window Period;

- (b) No later than ninety (90) days in advance of the first day of the Second Window Period, FLNG shall notify Customer of a one hundred eighty (180) day window ("Third Window Period") falling within the Second Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Third Window Period shall be the latest possible one hundred eighty (180) day window period within the Second Window Period;
- (c) No later than sixty (60) days in advance of the first day of the Third Window Period, FLNG shall notify Customer of a ninety (90) day window (*Fourth Window Period*) falling within the Third Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Fourth Window Period shall be the latest possible ninety (90) day window period within the Third Window Period;
- (d) No later than thirty (30) days in advance of the first day of the Fourth Window Period, FLNG shall notify Customer of a sixty (60) day window ('Fifth Window Period') falling within the Fourth Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Fifth Window Period shall be the latest possible sixty (60) day period within the Fourth Window Period;
- (e) No later than fifteen (15) days in advance of the first day of the Fifth Window Period, FLNG shall notify Customer of a thirty (30) day window ('Sixth Window Period') falling within the Fifth Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Sixth Window Period shall be the latest possible thirty (30) day period within the Fifth Window Period;
- (f) No later than seven (7) days in advance of the first day of the Sixth Window Period, FLNG shall notify Customer of a fifteen (15) day window ('Final Window Period') falling within the Sixth Window Period for the Commercial Start Date; provided that if FLNG fails to give timely notice of same, the Final Window Period shall be the latest possible fifteen (15) day period within the Sixth Window Period; and
- (g) No later than three (3) days in advance of the first day of the Final Window Period, FLNG shall notify Customer of the Commercial Start Date falling within the Final Window Period; provided that if FLNG fails to give timely notice of same, the Commercial Start Date shall be the latest possible day in the Final Window Period.

The Commercial Start Date shall be the date so notified, regardless of whether any unloading of Customer's LNG at the Freeport Facility actually occurs on such date.

7.3 Delay Caused by Force Majeure

Should an event of Force Majeure occur that has the effect of delaying the Commercial Start Date, then the Commercial Start Date shall be postponed or delayed to fully address the effects of such event.

7.4 Construction Progress Reports

Beginning on April 1, 2004 and every quarter thereafter until the Commercial Start Date, FLNG shall furnish to Customer an interim progress report (collectively the "Progress Reports") specifying the progress since the last report and the expected progress towards completing the construction, testing and operational start-up of the Freeport Facility. Each Progress Report shall include the status and progress of all construction, an update of the construction schedule, and any other information which Customer has reasonably requested in writing in advance to enable Customer to evaluate the status and progress of construction, testing and operational start-up of the Freeport Facility.

ARTICLE 8 FREEPORT FACILITY

8.1 Freeport Facility

(a) Standard of Operation.

- (i) By the Commercial Start Date, FLNG shall cause the Freeport Facility to be engineered, constructed and commissioned in a good and workman-like manner, in conformance with this Agreement and in material compliance with all Approvals necessary to commence operations of the Freeport Facility.
- (ii) On and after the Commercial Start Date, FLNG shall at all times engineer, construct, modify, maintain and operate (or cause to be engineered, constructed, modified, maintained and operated):
 - a. the Freeport Facility in accordance with the following: (x) International LNG Terminal Standards, including those dealing with the environment, health and safety; and (y) to the extent not inconsistent with International LNG Terminal Standards, such good and prudent practices as are generally followed in the LNG industry by Reasonable and Prudent Operators of LNG receiving and regasification terminals;
 - b. the Freeport Facility Pipeline in accordance with (x) Pipeline Standards, including those dealing with the environment, health and safety; and (y) to the extent not inconsistent with Pipeline Standards, such good and prudent practices as are generally followed by Reasonable and Prudent Operators of U.S. Gas pipelines; and

- c. each Gas Storage Facility in accordance with (x) Storage Standards, including those dealing with the environment, health and safety; and (y) to the extent not inconsistent with Storage Standards, such good and prudent practices as are generally followed by Reasonable and Prudent Operators of Gas storage facilities.
- (b) Facilities to be Provided. The Freeport Facility shall at all times after the Commercial Start Date include at least the following:
 - (i) appropriate systems for communications with LNG Vessels;
 - (ii) berthing facilities at Quintana Island, Texas capable of receiving an LNG Vessel having a displacement of no more than 150,000 tonnes, an overall length of no more than 1,050 feet, a beam of no more than 165 feet, and a draft of no more than 42 feet, which LNG Vessels can safely reach, fully laden, and safely depart, and at which LNG Vessels can lie safely berthed and unload safely afloat;
 - (iii) lighting sufficient to permit unloading operations (other than berthing or departing berth) by day or by night, to the extent permitted by Governmental Authorities (it being acknowledged, however, that FLNG shall in no event be obligated to allow nighttime berthing operations at the Freeport Facility if FLNG determines, acting as a Reasonable and Prudent Operator, that such operations during nighttime hours could pose safety or operational risks to the Freeport Facility, an LNG Vessel, or a third party);
 - (iv) unloading facilities capable of receiving LNG at a rate of no less than 10,000 Cubic Meters per hour when the pressure at the Receipt Point is at least 5.6 bars (gauge), with three (3) unloading arms each having a reasonable operating envelope to allow for ship movement and manifold strainers of sixty (60) mesh;
 - (v) a vapor return line system of sufficient capacity to transfer to an LNG Vessel quantities of Gas necessary for the safe unloading of LNG at required rates, pressures and temperatures;
 - (vi) facilities allowing ingress and egress between the Freeport Facility and the LNG Vessel by (x) representatives of Governmental Authorities for purposes of unloading operations; and (y) representatives of Customer and/or an independent surveyor for purposes of conducting tests and measurements of LNG on board the LNG Vessel in accordance with Annex I;

- (vii) LNG storage facilities with a total gross capacity of at least three hundred twenty thousand (320,000) Cubic Meters of LNG;
- (viii) LNG regasification facilities with a total daily capacity of at least 1,605,000 MMBTUs; and
- (ix) the Freeport Facility Pipeline with a total daily capacity at the Delivery Point of at least 2,140,000 MMBTUs, with suitable interconnections with Downstream Pipelines capable of accepting such volumes.

If FLNG elects, in its Sole Opinion, to also provide Gas Storage Facilities, such facilities shall be made available in the quantities determined by FLNG from time to time.

(c) <u>Facilities Not Provided.</u> For the avoidance of doubt, services and facilities not provided at the Freeport Facility include the following: (i) facilities and loading lines for liquid or gaseous nitrogen to service an LNG Vessel; (ii) facilities for providing bunkers; and (iii) facilities for the handling and delivery to the LNG Vessel of ship's stores, provisions and spare parts.

8.2 Modifications to Freeport Facility

- (a) Freeport Facility. Customer acknowledges that it is familiar with the general specifications for the LNG berthing and unloading facilities of the Freeport Facility as of the date hereof. FLNG acknowledges that such specifications have taken into consideration the necessary ship shore compatibility in relation to typical LNG tankers existing as of the Effective Date. After the date hereof, Customer shall ensure, at no cost to FLNG except as set forth in Section 8.2(b), that each of the LNG Vessels is fully compatible with the Freeport Facility. Should an LNG Vessel fail materially either to be compatible with the Freeport Facility, or to be in compliance with the provisions of Article 9, Customer shall not employ such LNG Vessel until it has been modified to be so compatible or to so comply.
- (b) Modifications. FLNG shall have the right, but not the obligation, to from time to time modify the Freeport Facility (including via construction or acquisition of other facilities) in order to perform the Services or any other mode of LNG, Gas, or energy-related services, subject to (x) such modifications not rendering the Freeport Facility incompatible with an LNG Vessel, (y) such modifications not reducing the Services Quantity except as allowed in Section 18.1 and (z) such modifications not otherwise conflicting with each Party's rights and obligations under this Agreement, including the requirements of Sections 5.1(a), 8.1(b) and 18.1. Notwithstanding (x) but subject to (y) and (z) in the foregoing sentence, FLNG may make such modifications in a manner that would render it incompatible with an LNG Vessel provided that:
 - (i) such modification is made pursuant to a change in International LNG Terminal Standards; or

(ii) the LNG Vessel is capable of being modified to maintain compatibility with both the Freeport Facility and other terminals in its normal/intended trade and, in connection with a modification (other than pursuant to paragraph (i) above), FLNG reimburses Customer for the reasonable actual costs incurred by Customer in causing Transporter to modify the LNG Vessel to maintain compatibility with the Freeport Facility as so modified; provided, further, that Customer shall use its reasonable efforts to minimize costs to be borne by FLNG hereunder, shall notify FLNG reasonably in advance of the nature and expected cost of all such LNG Vessel modifications by Transporter, and shall certify to FLNG the actual amount and detail of all costs incurred for which such reimbursement from FLNG is requested.

8.3 Customer Inspection Rights

On and after the Commercial Start Date and upon obtaining FLNG's prior written consent, which consent shall not be unreasonably withheld or delayed, a reasonable number of Customer's designated representatives (including LNG Suppliers and Temporary Customers) may from time to time inspect the operation of the Freeport Facility so long as such inspection occurs from 8:00 a.m. to 5:00 p.m. on a Business Day. Any such inspection shall be at Customer's sole risk and expense. Customer (and its designees) shall carry out any such inspection without any interference with or hindrance to the safe and efficient operation of the Freeport Facility. Customer's right to inspect and examine the Freeport Facility shall be limited to verifying FLNG's compliance with FLNG's obligations under this Agreement and shall not entitle Customer to make direct requests to FLNG regarding any aspect of the Freeport Facility. No inspection (or lack thereof) of the Freeport Facility by Customer hereunder, or any requests or observations made to FLNG or its representatives by or on behalf of Customer in connection with any such inspection, shall (a) modify or amend FLNG's obligations, representations, warranties and covenants under this Agreement or under any agreement or instrument contemplated by this Agreement; or (b) constitute an acceptance or waiver by Customer of FLNG's obligations under this Agreement.

8.4 Responsible Practices

(a) Wastes. FLNG warrants that it shall perform all activities hereunder (including any construction, maintenance and operations activities), Services and activities related thereto safely and in material conformance with all applicable laws, regulations and ordinances dealing with hazardous wastes. FLNG shall be deemed the generator and owner of any "Wastes" (which term, as used in this Agreement, includes "hazardous substances" and/or "hazardous materials" which are "disposed or released" as provided in the Federal Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA"), 42 U.S.C.A. 9601 et seq., "waste" as provided in the Federal Resource Conservation and Recovery Act ("RCRA"), 42 U.S.C.A. 6901 et seq. and includes waste of any kind including routine process waste and by-products which are disposed) generated in connection with FLNG's performance under this Agreement. As

such, FLNG shall be solely and independently responsible for any and all Liabilities related to such Wastes and shall safely and in material compliance with all applicable laws, regulations and ordinances dispose of or arrange for the disposal of the same.

- (b) Safety, Security and Hygiene. FLNG shall become familiar with all materials, chemicals, LNG, Gas and Wastes, used, produced or generated, respectively, while performing under this Agreement. FLNG shall materially comply with the Federal OSHA "Hazard Communication Standard," codified as 29 C.F.R. 1910.1200, et seq., and any similar state "right-to-know" laws which are currently in force or may be enacted in the future. FLNG is solely responsible for informing its employees and contractors of the hazards associated with any materials, chemicals, LNG, Gas and Wastes handled pursuant to this Agreement and is also responsible for training its employees and contractors in the proper methods of handling the same.
- (c) Personnel. All FLNG personnel (other than those performing routine maintenance or cleaning functions) shall be fluent in written and oral English.
- (d) <u>Indemnification</u>. FLNG shall indemnify, defend and hold Customer (including Customer's Affiliates, Temporary Customers and their respective officers, directors, employees, servants, agents, successors, and assigns) harmless from and against any and all Liabilities arising out of any and all of the following:
 - (i) the failure of FLNG or those acting under or for FLNG to comply with all federal, state and local laws, regulations and ordinances (including those related to the environment, health and safety) in connection with FLNG's performance of its obligations under this Section 8.4;
 - (ii) any contamination of the environment or damage to natural resources, including damage to any environmental medium (air, water, groundwater or soil), arising out of the activities of FLNG at a facility owned or operated by FLNG or a facility/location chosen by FLNG for its disposal of Wastes or any other facility at which FLNG's Wastes may be released or threatened to be released, including any liability imposed by federal, state and local laws, regulations and ordinances, including CERCLA, RCRA, or comparable and applicable state and local legal requirements; or
 - (iii) any damage to any property or death or injury of persons (including FLNG's employees) arising out of FLNG's performance or nonperformance of its obligations under this Section 8.4 or out of FLNG's operation of the Freeport Facility.

The obligations in this Section 8.4(d) shall apply regardless of the type of assertion being made including any legal, equitable, or admiralty causes of action or rights (including negligence, strict liability in tort, other tort, express or implied

warranty, indemnity, contract, contribution or subrogation), whether the assertion is made by a Party or any third party, regardless of when the events underlying such assertion occurred and shall apply to all types of damages allowed under this Agreement.

ARTICLE 9 TRANSPORTATION AND UNLOADING

9.1 LNG Vessels

- (a) <u>Customer to Cause LNG Vessels to Comply.</u> As between Customer and FLNG, Customer shall be responsible for the transportation of LNG from the Loading Port to the Freeport Facility. In this regard, Customer shall cause each LNG Vessel to comply with the requirements of this Article 9 in all respects.
- (b) Approvals and Documentation. Each LNG Vessel shall comply with the regulations of, and obtain all Approvals required by, Governmental Authorities to enable such LNG Vessel to enter, leave and carry out all required operations at the Freeport Facility. Each LNG Vessel shall at all times have on board valid documentation satisfactory to FLNG evidencing all such Approvals. Each LNG Vessel shall comply fully with the International Safety Management Code for the Safe Operation of Ships and Pollution Prevention effective July 1, 1998, and at all times be in possession of a valid safety management certificate.
- (c) <u>Fireboats, Escort Vessels and Port Charges</u>. Customer shall arrange for, or cause the appropriate Person to arrange for, such number and types of fireboats and escort vessels as are required by Governmental Authorities to attend the LNG Vessel so as to permit safe and efficient movement of the LNG Vessel within the maritime safety areas located in the approaches to and from the Freeport Facility. Customer shall pay, or cause to be paid, all Port Charges directly to the appropriate Person; provided, however, that FLNG shall be solely responsible for payment of all charges under the Freeport Facility Lease other than the Port Use Fees (excluding rent) and Thru-Put Fees. Should FLNG amend the Freeport Facility Lease, such amendment shall not operate to increase the Port Charges.
- (d) <u>Requirements</u>. Each LNG Vessel must satisfy the following requirements:
 - Specifications. Except as otherwise mutually agreed in writing by the Parties, each LNG Vessel shall be compatible with the specifications of the Freeport Facility identified in Section 8.1(b). Notwithstanding the foregoing, in the event an LNG Vessel is compatible with the specifications set forth in Section 8.1(b) or otherwise acceptable to FLNG, but a Governmental Authority or Pilot prohibits or otherwise hinders the utilization of such LNG Vessel, Customer's obligations under this Agreement shall not be excused or suspended by reason of Customer's inability (pursuant to the foregoing) to use such a vessel as an LNG Vessel. FLNG will provide reasonable assistance to Customer in Customer's attempt to remove any such prohibition or hindrance.

- (ii) <u>LNG Vessel Capacity</u>. Except as otherwise agreed in writing by FLNG, each LNG Vessel shall have an LNG cargo containment capacity of no less than one hundred twenty thousand (120,000) Cubic Meters, determined at the time of loading of LNG.
- (iii) <u>Condition of the LNG Vessel</u>. Each LNG Vessel shall be in compliance with International LNG Vessel Standards. The location of the unloading manifold shall allow a safe margin for movement of the arms within the operating envelope.
- (iv) <u>Classification Society</u>. Each LNG Vessel shall at all times be maintained in class with any of the American Bureau of Shipping, Lloyds Register of Shipping or Det Norske Veritas or any other classification society that is mutually agreeable to the Parties.
- (v) <u>Construction</u>. Each LNG Vessel shall have been constructed to all applicable International LNG Vessel Standards (including the International Code For the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk).
- (vi) Operation and Maintenance. Each LNG Vessel shall comply with, and shall be fully equipped, supplied and maintained to comply with, all applicable International LNG Vessel Standards. Unless approved by FLNG in writing, which approval shall not be unreasonably withheld or delayed, an LNG Vessel shall be prohibited from engaging in any maintenance, repair or in water surveys while berthed at the Freeport Facility. Each LNG Vessel shall comply fully with the guidelines of any Governmental Authority of the United States, including the National Oceanographic and Atmospheric Administration (NOAA), in relation to actions to avoid strikes in U.S. waters with protected sea turtles and cetaceans (e.g., whales and other marine mammals) and with regard to the reporting of any strike by the LNG Vessel which causes injury to such protected species.
- (vii) <u>Crew.</u> The officers and crew of each LNG Vessel shall have the ability, experience, licenses and training commensurate with the performance of their duties in accordance with, internationally accepted standards as adopted on first-class LNG-carrying vessels and as required by Governmental Authorities and any labor organization having jurisdiction over the LNG Vessel or her crew. Without in any way limiting the foregoing:
 - a. all shipboard personnel shall hold valid certificates of competence in accordance with the requirements of the law of the flag state of the LNG Vessel and any applicable requirements of the laws of the United States of

- b. the Master, chief engineer, chief mate and cargo engineer (and such other officers of the LNG Vessel having responsibilities associated with the preparation of the LNG Vessel for unloading) shall be trained and certified to a standard customary for an operator of a first-class LNG vessel of the type and tonnage of the LNG Vessel and in compliance with the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978;
- c. the Master, chief engineer, all cargo engineers, and all deck officers shall be fluent in written and oral English and shall maintain all records and provide all reports with respect to the LNG Vessel in English, and there shall otherwise be on board sufficient personnel with a good working knowledge of the English language to enable cargo handling and unloading to be carried out efficiently and safely and to enable communications between the LNG Vessel and those unloading the LNG Vessel to be carried out quickly and efficiently; and
- d. none of the LNG Vessel's Master, officers or crew shall, while serving on the LNG Vessel, abuse the use of drugs or alcohol, and Transporter shall maintain a written policy to such effect, such policy to meet or exceed the standards of the Oil Companies International Marine Forum's Guidelines for the Control of Drugs and Alcohol Aboard Ship, 1995, as amended from time to time. If any Master, officer or crew member abuses the use of drugs or alcohol, such individual shall be dismissed from service on the LNG Vessel.
- (viii) Communications. Each LNG Vessel shall have communication equipment complying with applicable regulations of Governmental Authorities and permitting such LNG Vessel to be in constant communication with the Freeport Facility and with other vessels in the area (including fireboats, escort vessels and other vessels employed in port operations).
- (ix) Pumping Time. Provided that the Freeport Facility supplies a suitable vapor return line meeting the requirements of Section 8.1(b)(v), then:
 - a. an LNG Vessel with an LNG cargo containment capacity less than or equal to one hundred forty thousand (140,000) Cubic Meters shall be capable of unloading LNG in a maximum of fifteen (15) hours; and

b. an LNG Vessel with an LNG cargo containment capacity greater than one hundred forty thousand (140,000) Cubic Meters shall be capable of unloading LNG in the number of hours derived after applying the following formula:

15 + x = maximum LNG unloading time (in hours)

where:

y = the LNG cargo containment capacity of the LNG Vessel in excess of 140,000 Cubic Meters; and

x = y/10,000 Cubic Meters

Time for connecting, cooling, stripping and disconnecting, and cooling of liquid arms shall not be included in the computation of pumping time.

9.2 Freeport Facility Marine Operations Manual

Acting as a Reasonable and Prudent Operator, FLNG shall develop and maintain a single marine operations manual that governs activities at the Freeport Facility, applies to all LNG Vessels and vessels used by Other Customers, and is consistent with International LNG Vessel Standards (but excluding the matters governed by the Freeport Services Manual). In developing such a manual, FLNG shall provide Customer with a preliminary draft of the same (the "Preliminary Marine Operations Manual"). If Customer desires to consult with FLNG regarding the contents of the Preliminary Marine Operation Manual, Customer shall, no later than fifteen (15) days from delivery of said manual by FLNG, request to meet with FLNG by providing notice thereof to FLNG, and FLNG shall, no later than thirty (30) days after receipt of such notice, meet with Customer to discuss said manual. If (a) Customer does not submit the foregoing notice to FLNG on a timely basis or (b) Customer and FLNG meet pursuant to such a notice and are able during such meeting to agree upon revisions to the draft, then such draft, as so revised (and as amended from time to time) shall constitute the "Freeport Facility Marine Operations Manual". If Customer and FLNG meet pursuant to the foregoing notice and are unable during such meeting to agree upon revisions to the Preliminary Marine Operations Manual, then FLNG shall determine, while using its reasonable efforts to accommodate Customer's views, the Freeport Facility Marine Operations Manual. In the event FLNG intends to amend the Freeport Facility Marine Operations Manual, then FLNG shall deliver to Customer and all Other Customers a copy of the Freeport Facility Marine Operations Manual and any amendments thereto promptly after they have been finalized or amended, as the case may be. The Parties shall consistent with this Agreement; however, in the event of a conflict between the terms of this Agreement and the Freeport Facility Marine Operations Manual, the terms of this Agreement shall control.

9.3 LNG Vessel Inspections; Right to Reject LNG Vessel

- Inspections. During the Term, on prior reasonable notice to Customer, FLNG may, at its sole risk, send its representatives (including an independent internationally recognized maritime consultant) to inspect during normal working hours any LNG Vessel as FLNG may consider necessary to ascertain whether the LNG Vessel complies with the provisions of this Agreement. FLNG shall bear the costs and expenses in connection with any inspection conducted hereunder. Any such inspection may include, as far as is practicable having regard to the LNG Vessel's operational schedule, examination of the LNG Vessel's hull, cargo and ballast tanks, machinery, boilers, auxiliaries and equipment; examination of the LNG Vessel's deck and engine scrap/rough and fair copy/official log books; review of records of surveys by the LNG Vessel's classification society and relevant Governmental Authorities; and review of the LNG Vessel's operating procedures and performance of surveys, both in port and at sea. Any inspection carried out pursuant to this Section 9.3(a): (i) shall not interfere with, or hinder, any LNG Vessel's safe and efficient construction or operation; and (ii) shall not entitle FLNG or any of its representatives to make any request or recommendation directly to Transporter except through Customer. No inspection (or lack thereof) of an LNG Vessel hereunder shall (i) modify or amend Customer's obligations, representations, warranties and covenants under this Agreement or under any agreement or instrument contemplated by this Agreement; or (ii) constitute an acceptance or waiver by FLNG of Customer's obligations under this Agreement.
- (b) Right to Reject LNG Vessel. Without prejudice to any other rights and remedies arising hereunder or by law or otherwise, FLNG shall have the right to reject any LNG Vessel that Customer intends to use to deliver LNG to the Freeport Facility if such LNG Vessel does not comply materially with the provisions of this Agreement, provided that:
 - (i) neither the exercise nor the non-exercise of such right shall reduce the responsibility of Customer to FLNG in respect of such vessel and her operation, nor increase FLNG's responsibilities to Customer or third parties for the same; and
 - (ii) Customer's obligations under this Agreement shall not be excused or suspended by reason of Customer's inability (pursuant to the foregoing) to use a vessel as an LNG Vessel.

9.4 Advance Notices Regarding LNG Vessel and Cargoes

(a) Changes in Expected Receipt Quantity. If, subsequent to issuing the notice required under Section 5.1(b)(ii) herein, Customer has reason to foresee a change in the Expected Receipt Quantity for a particular Arrival Date, Customer shall promptly provide notice thereof to FLNG and include in such notice Customer's new estimate of the Expected Receipt Quantity. To the extent such new estimate increases the Expected Receipt Quantity contained in any prior notice for that

Scheduled Unloading Window, FLNG shall use reasonable endeavors to accept such larger quantity but shall at all times retain the right not to accept such increase if, in its Sole Opinion, such new estimate will result in excess inventory at the Freeport Facility.

- (b) <u>LNG Vessel Nomination</u>. As soon as possible but no later than five (5) days prior to the scheduled loading date for a Cargo, Customer shall notify FLNG of the information specified below:
 - (i) name of LNG Vessel and, in reasonable detail, the age, dimensions, specifications, operator, safety record, and condition of such LNG Vessel;
 - (ii) name of Loading Port;
 - (iii) expected departure date of LNG Vessel from Loading Port;
 - (iv) estimated arrival date at the Freeport Facility; and
 - (v) any changes in the Expected Receipt Quantity since Customer's prior notice.

In the event Customer has a reason to foresee a change in the information specified above, Customer shall promptly provide notice thereof to FLNG. Moreover, if the vessel that Customer proposes to use as an LNG Vessel has not, within the immediately preceding Contract Year, delivered LNG to the Freeport Facility, Customer shall endeavor to notify FLNG thereof as soon as possible but in no event later than twenty (20) days prior to the first day of the applicable Scheduled Unloading Window.

- (c) <u>LNG Vessel Movements.</u> With respect to each Cargo of LNG to be delivered hereunder, Customer shall give, or cause the Master of the LNG Vessel to give, to FLNG the following notices:
 - (i) A first notice ("First Notice"), which shall be sent upon the departure of the LNG Vessel from the Loading Port and which shall set forth the time and date that loading was completed, the volume (expressed in Cubic Meters) of LNG loaded on board the LNG Vessel, the estimated time of arrival of the LNG Vessel at the Arrival Location ("ETA"), and any operational deficiencies in the LNG Vessel that may affect its performance at the Freeport Facility or berth;
 - (ii) A second notice ("Second Notice"), which shall be sent ninety-six (96) hours prior to the ETA set forth in the First Notice, stating the LNG Vessel's then ETA. If, thereafter, such ETA changes by more than six (6) hours, Customer shall give promptly, or cause the Master of the LNG Vessel to give promptly, to FLNG notice of the corrected ETA;

- (iii) A third notice ("Third Notice"), which shall be sent twenty-four (24) hours prior to the ETA set forth in the Second Notice (as corrected), confirming or amending such ETA. If, thereafter, such ETA changes by more than three (3) hours, Customer shall give promptly, or cause the Master of the LNG Vessel to give promptly, to FLNG notice of the corrected ETA;
- (iv) A fourth notice ("Final Notice"), which shall be sent twelve (12) hours prior to the ETA set forth in the Third Notice (as corrected), confirming or amending such ETA. If, thereafter, such ETA changes by more than one (1) hour, Customer shall give promptly, or cause the Master of the LNG Vessel to give promptly, to FLNG notice of the corrected ETA; and
- (v) An NOR, which shall be given at the time prescribed in Section 9.5 below.
- (d) <u>Characteristics of Cargoes.</u> With the First Notice, Customer shall notify FLNG, or cause FLNG to be notified, for FLNG's information only, of the following characteristics of the LNG comprising its Cargo as determined at the time of loading:
 - (i) Gross Heating Value per unit;
 - (ii) molecular percentage of hydrocarbon components and nitrogen;
 - (iii) average temperature; and
 - (iv) density at loading.
- (e) Right to Reject Certain Quantities Without prejudice to any other rights and remedies arising hereunder or by law or otherwise, FLNG shall for any reason (including limitations in LNG Storage) have the right to reject unloading of that quantity of LNG on board an LNG Vessel that exceeds by more than five percent (5%) the Expected Receipt Quantity for such Cargo as specified in, whichever applicable, (i) the notice delivered pursuant to Section 5.1(b)(ii) and utilized by FLNG for the purposes of determining the Customer LNG Receipt Schedule or (ii) any subsequent notice delivered pursuant to Section 9.4(a) and accepted by FLNG.

9.5 Notice of Readiness

(a) <u>Issuance</u>. Subject to any applicable restrictions, including any nighttime transit restrictions imposed by Governmental Authorities or Pilots or any other reasonable timing restrictions imposed by FLNG, the Master of an LNG Vessel or its agent shall give to FLNG its NOR to unload (berth or no berth) upon arrival of such LNG Vessel at the specific location off the Freeport Facility designated for such purposes in the Freeport Facility Marine Operations Manual (such location referred to as the "Arrival Location").

- (b) <u>Effectiveness</u>. An NOR given under Section 9.5(a) shall become effective as follows:
 - (i) For an LNG Vessel arriving at the Arrival Location at any time before 6:00 a.m., Central Time on the first day of the Scheduled Unloading Window allocated to such LNG Vessel, an NOR shall be deemed effective at the earlier of (x) 6:00 a.m., Central Time on the first day of such Scheduled Unloading Window; or (y) the time unloading commences;
 - (ii) For an LNG Vessel arriving at the Arrival Location at any time between the period of 6:00 a.m., Central Time on the first day of the Scheduled Unloading Window allocated to such LNG Vessel and two (2) hours before sunset, Central Time on the second day of such Scheduled Unloading Window (such period referred to as the "NOR Window"), an NOR shall become effective at the time of its issuance; or
 - (iii) For an LNG Vessel arriving at the Arrival Location at any time after the expiration of the NOR Window, an NOR shall become effective upon FLNG's notice to the LNG Vessel that it is ready to receive the LNG Vessel at berth.

9.6 Berthing Assignment

- (a) General Rule. FLNG shall determine the berthing sequence of all LNG Vessels and other vessels at the Freeport Facility in order to ensure compliance with the Customer LNG Receipt Schedule and the Other Customer LNG Receipt Schedules. If an LNG Vessel arrives not ready to unload for any reason, FLNG may refuse to allow it to berth.
- (b) Timely Arrival. FLNG shall berth an LNG Vessel arriving before or during its NOR Window at the first opportunity that FLNG determines such LNG Vessel will not interfere with unloading by any other scheduled vessel. Subject to Section 9.6(c), FLNG shall not be obligated to berth an LNG Vessel arriving later than two (2) hours prior to sunset at the Freeport Facility on the second day of the Scheduled Unloading Window allocated to such LNG Vessel (hereinafter referred to as the "Berthing Deadline"). For the avoidance of doubt, if FLNG does not berth such LNG Vessel by the Berthing Deadline, Customer's sole recourse and remedy for FLNG's breach thereof is demurrage pursuant to Section 9.7(c) and excess boil-off pursuant to Section 9.7(d). If berthing Services are unavailable to Customer and all Other Customers, then such unavailability shall be a Services Unavailability under Section 4.3 as opposed to a failure to meet the Berthing Deadline under this Section 9.6(b).
- (c) <u>Late Arrival</u>. FLNG shall berth an LNG Vessel arriving after its NOR Window at the first opportunity that FLNG reasonably determines such LNG Vessel will not interfere with unloading by any scheduled vessel.

9.7 Unloading Time

- (a) Allotted Unloading Time. The allotted unloading time for each LNG Vessel ("Allotted Unloading Time") shall be thirty-six (36) hours, subject to extensions for:
 - (i) reasons attributable to Customer, a Pilot, a Governmental Authority, the LNG Vessel or its Master, crew, owner or operator;
 - (ii) Adverse Weather Conditions;
 - (iii) Force Majeure;
 - (iv) unscheduled curtailment or temporary discontinuation of operations at the Freeport Facility in accordance with Section 18.2;
 - (v) occupancy of the berth by an LNG vessel that arrived at berth at the Freeport Facility no later than two (2) hours prior to sunset on the second day of the Scheduled Unloading Window allocated to such vessel, which shall result in an extension of no more than nine (9) hours;
 - (vi) failure to send the Final Notice; and
 - (vii) nighttime transit restrictions.
- (b) Actual Unloading Time. The actual unloading time for each LNG Vessel ("Actual Unloading Time") shall commence when the NOR is effective under Section 9.5(a) and shall end when the unloading and return lines of the LNG Vessel are disconnected from the Freeport Facility's unloading and return lines.
- (c) <u>Demurrage at the Freeport Facility.</u>
 - (i) In the event Actual Unloading Time exceeds Allotted Unloading Time (including any extension in accordance with Section 9.7(a)) ("Demurrage Event"), FLNG shall pay to Customer as liquidated damages demurrage in United States dollars (which shall be prorated for a portion of a day) determined in accordance with the rate set out in the following table:

LNG Vessel Cargo Capacity	Demurrage Rate in \$/day
Less than 120,000 Cubic Meters	Rate to be established by agreement of the Parties
120,000 Cubic Meters or greater up to, but not including, 160,000 Cubic Meters	***
160,000 Cubic Meters or greater up to, but not including, 200,000 Cubic Meters	***
200,000 Cubic Meters or greater	***

- (ii) If a Demurrage Event occurs, Customer shall invoice FLNG for such demurrage pursuant to Section 12.2.
- (d) Excess Boil-Off. If an LNG Vessel is delayed in commencement of unloading due to an event occurring at the Freeport Facility and for a reason that would not result in an extension of Allotted Unloading Time under Section 9.7(a), and if, as a result thereof, the commencement of unloading is delayed beyond *** after the Notice of Readiness is effective, then, for each full hour by which commencement of unloading is delayed beyond such *** period, FLNG shall pay Customer as liquidated damages an amount, on account of excess boil-off, equal to the Henry Hub Price multiplied by the quantity in MMBTUs equal to *** of the Cargo per day. Customer shall invoice FLNG for such excess boil-off pursuant to Section 12.2.

9.8 Unloading at the Freeport Facility

- (a) <u>Efficiency</u>. FLNG shall cooperate with Transporters (or their agents) and with the Master of each LNG Vessel to facilitate the continuous and efficient delivery of LNG hereunder.
- (b) <u>Vapor Return Line</u>. During unloading of each Cargo of LNG, FLNG shall return to the LNG Vessel Gas in such quantities as are necessary for the safe unloading of the LNG at such rates, pressures and temperatures as may be required by the design of the LNG Vessel, and such returned Gas shall not be deemed to be volume unloaded for Customer's account.

9.9 LNG Vessel Not Ready for Unloading; Excess Berth Time

(a) <u>Vessel Not Ready for Unloading.</u> If any LNG Vessel, previously believed to be ready for unloading, is determined to be not ready after being berthed, FLNG may direct the LNG Vessel's Master to vacate the berth and proceed to anchorage, whether or not other LNG vessels are awaiting the berth, unless it appears reasonably certain to FLNG that such LNG Vessel can be made ready without

disrupting the overall unloading schedule of the Freeport Facility or operations of the Freeport Facility. When an unready LNG Vessel at anchorage becomes ready for unloading, its Master shall notify FLNG. Upon the reberthing of any LNG Vessel vacated pursuant to this Section 9.9(a), Customer shall be responsible for any actual costs incurred by FLNG acting as a Reasonable and Prudent Operator as a result of such LNG Vessel not being ready for unloading, with FLNG using reasonable efforts to minimize such costs.

(b) <u>Berth Limitations</u>.

- (i) An LNG Vessel shall complete unloading and vacate the berth as soon as possible but not later than the following allowed berth time:
 - twenty-four (24) hours, in the case of an LNG Vessel with an LNG cargo containment capacity less than or equal to one hundred forty thousand (140,000) Cubic Meters; or
 - b. in accordance with the following formula, in the case of an LNG Vessel with an LNG cargo containment capacity greater than one hundred forty thousand (140,000) Cubic Meters:

+ x = allowed berth time (in hours)

where:

y = the LNG cargo containment capacity of the LNG Vessel in excess of 140,000 Cubic Meters; and

x = y/10,000 Cubic Meters.

Notwithstanding the foregoing, the aforementioned time restrictions shall be extended for: (a) reasons attributable to FLNG; (b) reasons attributable to a Pilot or to a Governmental Authority; (c) Adverse Weather Conditions; (d) Force Majeure; and (e) nighttime transit restrictions.

- (ii) If an LNG Vessel fails to depart at the end of its allowed berth time, FLNG may direct the LNG Vessel to vacate the berth and proceed to sea at utmost dispatch.
- (iii) If an LNG Vessel fails to vacate the berth after expiration of its allowed berth time after receipt of FLNG's notice to do so under this Section 9.9, Customer shall reimburse FLNG for any and all reasonable and actual damages its incurs as a result thereof, including amounts FLNG becomes contractually obligated to pay as demurrage to any of the Other Customers or to pay any Other Customer for excess boil-off.

(iv) Subject to this Section 9.9, in the event an LNG Vessel fails to vacate the berth and Customer is not taking actions to cause it to vacate the berth, FLNG may effect such removal at the expense of the Customer.

ARTICLE 10 RECEIPT OF LNG

10.1 Title, Custody and Risk of Loss

- (a) <u>Title to Customer's Inventory.</u> Subject to Section 3.4, title with respect to Customer's Inventory shall remain with Customer even during periods when it is in the possession and control of FLNG (including while held at a Gas Storage Facility and as provided in Section 10.5). For the avoidance of doubt, title and risk of loss with respect to Retainage shall pass to FLNG at the Receipt Point.
- (b) <u>Possession, Risk of Loss and Control</u>. Possession, risk of loss and control of Customer's LNG shall pass from Customer to FLNG upon delivery of same at the Receipt Point. Possession, risk of loss and control of Customer's Inventory shall pass from FLNG to Customer upon delivery of same at the Delivery Point.

10.2 No Encumbrance

- (a) Customer's Covenants. Customer warrants to FLNG that (i) Customer has title to all of Customer's Inventory, other than to the Temporary Release Inventory; and (i) the relevant Temporary Customer has title to all of the Temporary Release Inventory. Customer covenants that Customer's Inventory shall remain free of all encumbrances and Liabilities therefor, and that no circumstances will exist which could give rise to any Liabilities or encumbrances relating thereto (collectively, "Claims") other than those that may be caused by acts or omissions of FLNG or Other Customers. Customer agrees to fully defend, indemnify and hold FLNG and its Affiliates harmless against all Claims regarding Customer's Inventory, including Claims brought by Other Customers, other than any Claims caused by acts or omissions of FLNG or Other Customers. For purposes of Section 10.2, the term "encumbrance" shall include any mortgage, pledge, lien, charge, adverse claim, proprietary right, assignment by way of security, security interest, title retention, preferential right or trust arrangement or any other security agreement or arrangement having the effect of security.
- (b) FLNG's Covenants. FLNG covenants that it has the right to deliver to Customer at the Delivery Point all Gas held for Customer's account free from all Claims relating thereto. FLNG covenants that Customer's Inventory, while in FLNG's possession or control, shall remain free of all Claims, other than those that may be caused by Customer's acts or omissions. FLNG also covenants that the net proceeds resulting from any sale of Customer's Inventory under Section 3.4 shall remain free of all Claims, other than those that may be caused by Customer's acts or omissions. FLNG agrees to fully defend, indemnify and hold Customer and its Affiliates harmless from and against all Claims regarding Customer's Inventory or such net proceeds, other than Claims caused by the acts or omissions of Customer.

10.3 Receipt of LNG

The receipt of LNG from an LNG Vessel at the Receipt Point shall be carried out by use of pumps and other equipment on the LNG Vessel under such reasonable and customary conditions specified in the Freeport Facility Marine Operations Manual; provided that such reasonable and customary conditions for the receipt of LNG specified in the Freeport Facility Marine Operations Manual shall not (a) exceed the specifications of the LNG Vessel, provided that such specifications meet the requirements of this Agreement; or (b) violate the requirements of a Governmental Authority having jurisdiction over the LNG Vessel.

10.4 Quality and Measurement of Customer's LNG

Customer's LNG shall be measured and tested in accordance with Annex I. Customer shall ensure that all LNG delivered at the Receipt Point for Customer's account shall conform to the following specifications:

(a) Gross Heating Value

- (i) LNG when delivered by Customer to FLNG shall have, in a gaseous state, a Gross Heating Value of not less than 950 BTU per Standard Cubic Foot and not more than *** BTU per Standard Cubic Foot.
- (ii) If the Gross Heating Value of LNG to be delivered hereunder is higher than the limits set forth in Section 10.4(a)(i) by reason of boil-off occurring during a delay (other than a delay which extends Allotted Unloading Time under Section 9.7(a)) caused by FLNG in unloading an LNG Vessel of more than thirty (30) hours after NOR becomes effective, such LNG shall be deemed to have met the quality specifications of this Agreement regarding Gross Heating Value.

(b) <u>Components</u>.

- (i) At the Receipt Point, the LNG delivered by Customer to FLNG shall, when in a gaseous state, contain not less than eighty six (86) molecular percentage (86 MOL%) of methane (C₁) and, for the components and substances listed below, such LNG shall not contain more than the following:
 - a. Nitrogen (N₂), 0.5 MOL%;
 - b. Ethane (C₂), 11 MOL%;
 - c. Propane (C₃), 3.5 MOL%;

- d. Butanes (C₄) and heavier, 2 MOL%;
- e. Pentanes (C₅) and heavier, .09 MOL%;
- f. Hydrogen sulfide (H,S), .25 grains per 100 Standard Cubic Feet (.25 grains/100 SCF); and
- g. Total sulfur content, 5 grains per 100 Standard Cubic Feet (5 grains/100 SCF).
- (ii) The LNG when delivered by Customer to FLNG shall contain no water, mercury, active bacteria or bacterial agents (including sulfate reducing bacteria or acid producing bacteria) and other contaminants or extraneous material.

10.5 Off-Specification LNG

- (a) Refusal of Off-Spec LNG. Without prejudice to any other rights and remedies of FLNG hereunder, subject to Section 10.5(b), FLNG may refuse to take delivery of all or part of any LNG not conforming to the quality specifications set forth in Section 10.4 ("Off-Spec LNG"). However, if FLNG refuses to take delivery of LNG and such LNG is subsequently determined not to have been Off-Spec LNG, then FLNG shall pay all Liabilities incurred by Customer as a result of FLNG's refusal, with Customer using reasonable efforts to minimize such Liabilities.
- (b) Notice. Customer shall provide notice to FLNG as soon as reasonably practicable of any existing or anticipated failure of the LNG available for delivery to FLNG hereunder to conform to the quality specifications set forth in Section 10.4, giving details of the nature and expected magnitude of the variance, the cause of the non-compliance and the probable duration thereof, including the Cargoes and Scheduled Unloading Windows to be affected thereby. If so notified, FLNG shall as soon as possible inform Customer whether it intends to reject any of such Off-Spec LNG. If FLNG is notified by Customer prior to the commencement of unloading of a Cargo at the Freeport Facility that the LNG is Off-Spec LNG and the quantity is delivered to the Freeport Facility, FLNG shall use reasonable endeavors to take delivery of any Cargoes which it would otherwise be entitled to reject; provided, however that FLNG shall be entitled to delay unloading of Off-Spec LNG for the period of time reasonably required for FLNG to determine whether it can take delivery of such Off-Spec LNG pursuant to this Section 10.5(b). Subject to FLNG first using its reasonable endeavors to take delivery of any Cargoes containing Off-Spec LNG, FLNG shall:
 - (i) notify Customer that FLNG will take delivery of some or all of the affected Cargoes, without prejudice to FLNG's rights and remedies with respect to such Off-Spec LNG other than FLNG's right to reject said Cargo; or
 - (ii) reject all or any of the affected Cargoes.

Unless FLNG was notified by Customer of Off-Spec LNG, FLNG shall provide notice to Customer as soon as reasonably practicable of any Off-Spec LNG, giving details of the nature and expected magnitude of the variance and the affected Cargo.

10.6 Customer's Responsibility and Reimbursement

- (a) No Continuing Waiver. Acceptance of Off-Spec LNG shall not prevent FLNG from refusing future deliveries of Off-Spec LNG. No waiver by FLNG of any default by Customer of any of the specifications set forth in this Article 10 shall ever operate as a continuing waiver of such specification or as a waiver of any subsequent default, whether of a like or different character.
- (b) <u>Delivery of a Cargo of Off-Spec LNG</u>. If FLNG accepts delivery of a Cargo of Off-Spec LNG which it would otherwise be entitled to reject, Customer shall:
 - (i) bear the financial responsibility for all reasonable and actual incremental costs (other than capital costs) and Liabilities incurred by FLNG or any of FLNG's Affiliates, in each case acting as a Reasonable and Prudent Operator, in connection with receiving and treating Off-Spec LNG by such means as are appropriate, including mixing such Off-Spec LNG with lower calorific value Gas or injecting nitrogen, with FLNG using reasonable efforts to minimize such costs; and
 - (ii) indemnify and hold harmless FLNG, its Affiliates and their respective directors, officers and employees from any and all Liabilities, including any of same attributable to claims of any Person (including Temporary Customers) and any Other Customers, which arise out of, are incident to or result from the acceptance, handling, disposal or use of Off-Spec LNG.
- (c) Extended Delivery of Off-Spec LNG If (i) Customer notifies FLNG pursuant to Section 10.5(b) of an anticipated delivery of two (2) or more Cargoes of Off-Spec LNG and (ii) the Parties agree for FLNG to incur incremental capital costs in order to accept delivery of such Cargoes, then Customer shall, in addition to its payment and indemnification obligations under Section 10.6(b), bear the financial responsibility for and directly fund, at FLNG's election, all such incremental capital costs.

10.7 Subsequent Deliveries

Unless otherwise requested by Customer, any quantities of LNG which were, under the Customer LNG Receipt Schedule, scheduled to be unloaded during the Contract Year but were actually unloaded at the Freeport Facility within the first *** (***) days in the following Contract Year shall be, for the purposes of the Maximum LNG Reception Quantity, deemed to have been received by FLNG in the Contract Year in which such quantities were originally scheduled to be unloaded.

ARTICLE 11 REDELIVERY OF GAS

11.1 General

- (a) <u>Delivery Point.</u> Subject to Section 3.3(a), the volume of Gas nominated by Customer for any day pursuant to Section 5.2 shall be delivered at the Delivery Point.
- (b) <u>Commingled Stream.</u> Customer acknowledges and agrees that Customer's Inventory shall be delivered by FLNG in a commingled stream, including that combined with LNG received by FLNG from Temporary Customers and any Other Customers. Customer furthers acknowledges and agrees that Customer shall have no right to receive Gas of the same quality as Customer's LNG, provided that the specifications of the commingled Gas stream at the Delivery Point satisfy the requirements set forth in Section 11.3.
- (c) <u>Odorization</u>. FLNG will deliver Customer's Inventory at the Delivery Point in its natural state without the addition of any odorizing agent, and FLNG shall not be obligated to add odorizing agents to any Gas unless required to do so by a Governmental Authority. FLNG does not assume any responsibility for Liabilities by reason of the fact that it has not odorized Customer's Inventory prior to its delivery to Customer.

11.2 Customer's Responsibility

(a) Downstream Arrangements. Customer shall arrange for the purchase and transportation of Gas by Downstream Pipelines in order to meet its obligations to take redelivery of Gas in accordance with the provisions of Section 3.4 at the rates nominated pursuant to Article 5. Customer shall be solely responsible for making all necessary arrangements with third parties at or downstream of the Delivery Point to enable FLNG to deliver Gas to Downstream Pipelines on a timely basis pursuant to the terms and conditions of this Agreement. Customer shall also be solely responsible for ensuring that all such arrangements are consistent with the terms and conditions of this Agreement and shall require all relevant third parties to confirm to FLNG all of Customer's nominations and scheduling of deliveries of Gas, such confirmation to be by telephone, electronic transmission, or other means acceptable to FLNG. Such third-party arrangements shall be timely communicated to, and coordinated with, FLNG, and FLNG shall have no liability whatsoever for any failure of any such third party to provide downstream arrangements. The manner in which Customer's Inventory is transported from or purchased at the Delivery Point shall be subject to the rules, guidelines, and policies of the Downstream Pipeline transporting or purchasing any such Gas (as may be changed from time to time by the Downstream Pipeline). Customer and FLNG recognize that the receipt and delivery on the Downstream Pipeline's facilities of Gas shall be subject to the operational procedures of such Downstream Pipeline.

- (b) Imbalance Charges. In the event a Downstream Pipeline imposes scheduling fees, imbalance charges, cash out costs or similar costs, fees or damages for imbalances associated with Customer's Gas ("Imbalance Charges"), Customer shall be obligated to use its reasonable efforts to avoid imposition of such Imbalance Charges. Customer shall indemnify and hold harmless FLNG, its Affiliates and their respective directors, officers and employees from all Liabilities arising out of, incident to or resulting from any Imbalance Charge directly resulting from Customer's acts or omissions. FLNG shall indemnify and hold harmless Customer, its Affiliates and their respective directors, officers and employees from all Liabilities arising out of, incident to or resulting from any Imbalance Charge directly resulting from FLNG's acts or omissions. FLNG will use reasonable efforts to minimize any Imbalance Charges and will notify Customer of any Imbalance Charges included in or imposed by any operational balancing agreement.
- (c) <u>Limitation</u>. Customer shall ensure that its Gas transportation and sales arrangements are in compliance with all applicable laws and regulations. In this regard, Customer agrees that it shall transport, or cause to be transported, Customer's Inventory only into intra-state Gas pipelines or storage facilities unless otherwise approved by Governmental Authorities.

11.3 Specifications and Measurement of Gas at the Delivery Point

Gas delivered to Customer at the Delivery Point shall be measured and tested in accordance with Annex II. FLNG shall ensure that all Gas delivered at the Delivery Point for Customer's account shall conform to the following specifications:

- (a) Gross Heating Value. Gas when delivered by FLNG to Customer shall have a Gross Heating Value of not less than 950 BTU per Standard Cubic Foot and not more than 1150 BTU per Standard Cubic Foot.
- (b) <u>Components</u>.
 - (i) At the Delivery Point, Gas delivered by FLNG to Customer shall contain not less than eighty six (86) molecular percentage (86 MOL%) of methane (C₁) and, for the components and substances listed below, such Gas shall not contain more than the following:
 - a. Nitrogen (N₂) three (3) MOL%;
 - b. Pentanes (C₅) and heavier, 0.1 MOL%;
 - c. Hydrogen sulfide (H₂S), .25 grains per 100 Standard Cubic Feet (.25 grains/100 SCF);
 - d. Total sulfur content, 5 grains per 100 Standard Cubic Feet (5 grains/100 SCF);

- e. Oxygen (O_2) , ten (10) parts per million;
- f. Carbon dioxide (CO₂), two (2) MOL%; and
- g. Water (H₂O), seven (7) pounds per one million (1,000,000) cubic feet.
- (ii) Gas when delivered by FLNG to Customer shall contain no mercury, active bacteria or bacterial agents (including sulfate reducing bacteria or acid producing bacteria) and other contaminants or extraneous material.
- (c) <u>Gas Delivery Pressure</u>. Customer's Inventory shall be delivered at the Delivery Point at an adequate pipeline pressure. In this regard, FLNG shall provide Gas at an adequate pressure (not to exceed 1250 psig) into the Freeport Facility Pipeline in order to deliver Gas at the Delivery Point.

11.4 Nonconforming Gas

- (a) Right to Reject. Without prejudice to any other rights and remedies of Customer hereunder, Customer shall have the right to reject Gas that does not conform to the specifications set forth in Section 11.3 ("Nonconforming Gas"); provided that Customer shall first use reasonable endeavors to take delivery of any Nonconforming Gas which it would otherwise be entitled to reject.
- (b) FLNG Indemnity. If Customer accepts delivery of Nonconforming Gas which it would otherwise be entitled to reject, FLNG shall indemnify and hold harmless Customer, its Affiliates and their respective directors, officers and employees from any and all Liabilities, including any of same attributable to claims of any Person (including Other Customers, a Downstream Pipeline, a Downstream Purchaser and Temporary Customers), which arise out of, are incident to or result from the acceptance, handling, disposal or use of Nonconforming Gas. If Customer accepts delivery of Nonconforming Gas which it would otherwise be entitled to reject, FLNG shall bear the financial responsibility for all reasonable and actual incremental costs (other than capital costs) and Liabilities incurred by Customer or any of Customer's Affiliates, in each case acting as a Reasonable and Prudent Operator, in connection with accepting delivery of Nonconforming Gas.

ARTICLE 12 PAYMENT

12.1 Monthly Statements

Between the first (1st) day of each month and the tenth (10th) day of each month, commencing with the month after the beginning of the Payment Period, FLNG shall deliver to Customer a statement setting forth the following:

(a) the Fixed Fee Component of the Reservation Fee for the prior month;

- (b) the Variable Fee Component of the Reservation Fee for the prior month;
- (c) the Incremental Costs, if any, for the prior month;
- (d) the Excess Reception Fee, if any, for the prior month;
- (e) any adjustment pursuant to Section 4.3; and
- (f) any adjustment pursuant to Section 4.7(a), in October of each year.

All statements delivered by FLNG to Customer shall as much as practicable account separately for the Fee related to each Temporary Release from all other amounts owed by Customer. FLNG shall reflect the Retainage for the prior month on any such statement.

12.2 Other Statements

If any other amount is due from one Party to the other hereunder and if provision for the invoicing of that amount due is not made elsewhere in this Article 12, then the Party to whom such amount is due shall furnish a statement therefor to the other Party, along with pertinent information showing the basis for the calculation thereof. Upon request, the Party who issued a statement under this Article 12 shall provide reasonable supporting documentation to substantiate any amount claimed to be due.

12.3 Adjustment

If, within two (2) years of the issuance of a statement, either Party acquires information indicating the necessity of an adjustment to such statement rendered hereunder, then the Party acquiring the information shall promptly serve on the other Party a written notice setting forth that information. Unless otherwise provided herein, after obtaining that information, the Party that prepared the prior statement which by reason of that information must be adjusted, shall promptly prepare and serve on the other Party an adjusted statement, showing the necessary payment, the calculation of the payment amount, and the Party from whom the payment is owing.

12.4 Payment Due Dates

- (a) <u>Due Date for Monthly Statement</u>. Each monthly statement submitted pursuant to Section 12.1 shall become due and payable on the later of (i) ten (10) days after delivery by FLNG of such monthly statement or (ii) the twenty fifth (25th) day of the month in which such monthly statement was received; provided that if such day is not a Business Day, it shall become due and payable on the next Business Day.
- (b) <u>Due Date for Other Statements</u>. Each statement submitted pursuant to Section 12.2 shall become due and payable on the thirtieth (3th) day after the date on which it is received; provided that if such payment due date is not a Business Day, the due date for such payment shall be extended to the next Business Day. For purposes of this Section 12.4(b), a facsimile copy of an invoice shall be deemed received by a Party on the next Business Day following the day on which it was sent.

(c) Interest. If the full amount of any statement is not paid when due, the unpaid amount thereof shall bear interest at the Base Rate, compounded annually, from and including the day following the due date up to and including the date when payment is made.

12.5 Payment

Each Party shall pay, or cause to be paid, in United States dollars in immediately available funds, all amounts that become due and payable by such Party pursuant to any statement issued hereunder, to a bank account or accounts designated by and in accordance with instructions issued by the other Party on the statement. Each payment of any amount owing under Section 4.1(a) and each payment of undisputed amounts (the disputed portion of which is addressed under Section 12.7) owing under any other provisions hereunder shall be in the full amount due without reduction or offset for any reason (except as expressly allowed under this Agreement), including Taxes, exchange charges, or bank transfer charges. Notwithstanding the preceding sentence, the paying Party shall not be responsible for a designated bank's disbursement of amounts remitted to such bank, and a deposit in immediately available funds of the full amount of each statement with such bank shall constitute full discharge and satisfaction of the statement.

12.6 Nonpayment

The term "Cumulative Delinquency Amount" shall mean, with respect to a Party, the cumulative amount owed that is not disputed under Section 12.7, expressed in United States dollars, that is owed by that Party to the other Party under this Agreement and is past due. Without prejudice to a Party's right of offset, if a Party's failure to pay when due an amount owing hereunder causes its Cumulative Delinquency Amount to exceed ***, then the Party to which such amount is owed shall have the right, upon giving thirty (30) days written notice (such notice hereinafter referred to as the "Delinquency Notice") to the owing Party, to suspend performance of its obligations under this Agreement until such amount, with interest in accordance with Section 12.4(c), has been paid in full; provided, however, that (a) no such suspension of a Party's obligations under this Section 12.6 shall excuse the owing Party from the performance of its obligations hereunder, and (b) in the event that FLNG suspends performance under this Section 12.6, Customer shall continue to be liable for the Fee pursuant to Article 4. If any such Cumulative Delinquency Amount has not been paid within sixty (60) days after the issuance of the Delinquency Notice, then the Party to whom such amount is owed shall have the right, upon not less than thirty (30) days notice to the other Party, to terminate this Agreement without the necessity of any further action, unless within that thirty (30) day period, the Party to which such amount is owed receives payments from or on behalf of the owing Party equal to the Cumulative Delinquency Amount. Any such termination shall be without prejudice to any other rights and remedies of the terminating Party arising hereunder or by law or otherwise, including the right of such Party to receive payment in respect of all obligations and claims that arose or accrued prior to such termination or by reason of such default by the owing Party.

12.7 Disputed Statements

In the event of disagreement concerning any statement, Customer or FLNG (as the case may be) shall make provisional payment of the total amount owing under Section 4.1(a) and the undisputed amounts under the remaining provisions hereof and shall immediately notify the other Party of the reasons for such disagreement. Statements may be contested by Customer or FLNG (as the case may be) only if, within a period of two (2) years after a Party's receipt thereof, Customer or FLNG (as the case may be) serves on the other Party notice questioning their correctness. If no such notice is served, statements shall be deemed correct and accepted by both Parties. Promptly after resolution of any Dispute as to a statement, the amount of any overpayment or underpayment (plus interest as provided in Section 12.4(c)) shall be paid by FLNG or Customer to the other, as the case may be.

12.8 Final Settlement

Within sixty (60) days after expiration of the Term, FLNG and Customer shall determine the amount of any final reconciliation payment. After the amount of the final settlement has been determined, FLNG shall send a statement to Customer, or Customer shall send a statement to FLNG, as the case may be, in United States dollars for amounts due under this Section 12.8, and FLNG or Customer, as the case may be, shall pay such final statement no later than twenty (20) days after the date of receipt thereof.

ARTICLE 13 SECURED FINANCING

13.1 UCC-1

FLNG shall execute and file (and re-file upon expiration), and grants to Customer the right to execute and file (and re-file upon expiration) on FLNG's behalf, in the proper office of the proper jurisdiction a UCC-1 for the purpose of giving notice to the creditors of FLNG that (a) Customer's Inventory is owned by Customer, (b) for purposes of the Uniform Commercial Code, the nature of the relationship between Customer and FLNG with regard to such LNG and Gas is that of bailer and bailee (and if the form of the UCC1 then in effect contained a "bailment" box to be checked, the bailment relationship would be indicated), and (c) neither FLNG nor its creditors shall have any ownership or other right arising from such bailment. Nothing contained in a UCC-1 shall act as an amendment to the terms of this Agreement.

13.2 Secured Financing

(a) Acknowledgement. In the event FLNG obtains any secured financing (except those rising out of the operation of law for the procurement or installation of fixtures and equipment), FLNG shall obtain from the lender providing such financing an acknowledgement that (i) Customer's Inventory is owned by

Customer, and (ii) neither FLNG nor such lender shall have any ownership or other right arising from this Agreement except as expressly provided in Sections 3.4 and 10.2(b).

(b) Non-Disturbance. In the event FLNG obtains any secured financing (except those rising out of the operation of law for the procurement or installation of fixtures and equipment), FLNG shall obtain from the lender providing such financing a non-disturbance agreement protecting the rights of Customer under this Agreement in the event of a foreclosure or other realization by such lender on its collateral. Customer acknowledges and agrees that in order to obtain such non-disturbance agreement, Customer may be required to execute and deliver to such lender an agreement to attorn to such lender on the terms of this Agreement in the event of a foreclosure or other realization by such lender on its collateral, and the obligation of FLNG to obtain a non-disturbance agreement shall be conditioned on Customer providing such agreement to attorn. For purposes of this Section 13.2(b), "attorn" shall mean that Customer would recognize the lender as the holder of FLNG's rights and obligations under this Agreement.

ARTICLE 14 DUTIES, TAXES AND OTHER GOVERNMENTAL CHARGES

As between FLNG and Customer, Customer shall be responsible for and pay, or cause to be paid, all Taxes that may be imposed or levied on Customer's Inventory and the LNG Vessels. Customer shall not be responsible to reimburse FLNG for any Taxes or New Taxes other than those that may be allocated to Customer under Section 4.5.

ARTICLE 15 INSURANCE

15.1 FLNG's Insurance

FLNG shall be responsible for obtaining and maintaining (a) insurance for the Freeport Facility to the extent required by applicable law; and (b) additional insurance, as is reasonably necessary and available on reasonable commercial terms, against such other risks and at such levels as a Reasonable and Prudent Operator of a shared use LNG receiving and regasification terminal would obtain. FLNG shall obtain such insurance from a reputable insurer (or insurers) reasonably believed to have adequate financial reserves. Any insurance policy required pursuant to this Section 15.1 shall contain a standard waiver of subrogation endorsement. Such insurance (other than business interruption insurance) shall be primary to any insurance held by Customer and shall name Customer, including its Affiliates and their directors, officers, agents and employees, as an additional insured. FLNG shall be deemed to have obtained insurance at a level that a Reasonable and Prudent Operator would have obtained in the event FLNG obtains the following minimum insurance coverages:

(a) Commercial General Liability Insurance and Wharf Owners Liability Insurance in an amount of ***, such insurance to be either obtained independently, or as part of an umbrella policy;

- (b) Statutory Worker's Compensation and Employers Liability Insurance in an amount of ***;
- (c) All-Risk Property Insurance equal to replacement value or maximum foreseeable loss if less than replacement value (except, however, for flood and earthquake, the minimum coverage shall be ***);
- (d) Business Interruption Insurance equal to ***, based on a forty-five (45) day waiting period; and
- (e) Pollution Insurance in an amount equal to ***.

FLNG shall furnish to Customer detailed written information concerning all insurance policies required under Section 15.1 prior to the Commercial Start Date and thereafter at least once each Contract Year. The receipt of such information shall not impose any obligation on Customer.

15.2 Customer's Insurance

- (a) <u>LNG Vessel Insurance</u>. Customer shall ensure that insurances are procured and maintained for each LNG Vessel in accordance with the following provisions. In all cases, such insurance shall establish insurance coverages consistent with insurances to the standards which a shipowner operating reputable LNG vessels, as a Reasonable and Prudent Operator, should observe in insuring LNG vessels of similar type, size, age and trade as such LNG Vessel. In this regard:
 - (i) Hull and Machinery Insurance shall be placed and maintained with reputable marine underwriters; and
 - (ii) Protection & Indemnity Insurance ("P&I Insurance") shall be placed and maintained as an unlimited entry, if such entry is available, with and subject to and on the basis of the rules of any of the reputable P&I insurance associations experienced in providing P&I Insurance for LNG vessels ("Approved Provider"). The terms of the P&I Insurance shall be those of the standard rules of the Approved Provider, provided that special provisions resulting from Transporter's acceptance, if it does so, of the Port Liability Agreement pursuant to Section 15.3 shall be incorporated into the terms of Transporter's P&I Insurance.
- (b) Evidence of Insurance. Prior to the commencement of deliveries to the Freeport Facility and thereafter at least once each Contract Year, Customer shall furnish the following evidence of insurance to FLNG in relation to each LNG Vessel: cover notes, certificates of entry, the latest rules of the particular Approved Provider, and detailed written information concerning all required insurance policies.

15.3 Port Liability Agreement

Notwithstanding any other provision of this Agreement and any rights that a Transporter may have under applicable law, each of FLNG and Customer agree to the Port Liability Agreement set forth on Exhibit C in relation to Liabilities for incidents caused by an LNG Vessel occurring at the Freeport Facility. Customer shall cause Transporter to execute the Port Liability Agreement in the form set forth on Exhibit C prior to Transporter's LNG Vessel's arrival at the Freeport Facility. ***

ARTICLE 16 LIABILITIES

16.1 Limitation of Liability

- (a) <u>Limitation of FLNG</u>. In no case shall the Liabilities of FLNG to Customer arising out of, relating to, or connected with an Event under this Agreement exceed ***; provided, however, that the foregoing limitation shall not apply to (i) FLNG's obligations under Section 8.4 (excluding Section 8.4(d)(iii)); (ii) FLNG's obligations under Section 10.2(b) and Article 13; (iii) FLNG's obligation to credit the net proceeds under Section 3.4; or (iv) Liabilities caused by the Gross Negligence/Willful Misconduct of FLNG.
- (b) <u>Limitation of Customer</u>. In no case shall the Liabilities of Customer to FLNG arising out of, relating to, or connected with an Event under this Agreement exceed ***; provided, however, that the foregoing limitation shall not apply to (i) the Fee; (ii) the obligation of Customer to indemnify and hold FLNG harmless under Section 15.3; or (iii) Liabilities caused by the Gross Negligence/Willful Misconduct of Customer.
- (c) <u>Definitions</u>. For purposes of this Section 16.1, an "Event" means any occurrence or series of occurrences having the same origin, and "Gross Negligence/Willful Misconduct" means any act or failure to act (whether sole, joint or concurrent) by FLNG which was intended to cause, or which was in reckless disregard of or wanton indifference to, harmful consequences FLNG knew, or should have known, such act or failure would have on the safety or property of another Person.

16.2 Limitation on Damages

No Party shall be liable to the other Party for or in respect of:

- (a) loss of profits or business interruption; or
- (b) any exemplary or punitive damages,

suffered or incurred by the other Party or any Person resulting from breach of or failure to perform this Agreement or the breach of any representation or warranty hereunder, whether express or implied, and whether such damages are claimed under breach of warranty, breach of contract, tort, or other theory or cause of action at law or in equity, except to the extent such damages have been awarded to a third party and are subject to allocation between or among the parties to the Dispute.

ARTICLE 17 FORCE MAJEURE

17.1 Events of Force Majeure

Neither Party shall be liable to the other for any delay or failure in performance hereunder if and to the extent such delay or failure is a result of Force Majeure. Subject to the provision of this Article 17, the term "Force Majeure" shall mean any act, event, or circumstance that is not reasonably within the control of and that prevents or delays a performance by a Party. Nothing in this Article 17 shall be construed to require a Party to observe a higher standard of conduct than that required of a Reasonable and Prudent Operator as a condition to claiming the existence of Force Majeure.

17.2 Limitation on Scope of Force Majeure Protection

17.3 Notice

A Force Majeure event shall take effect at the moment such an event or circumstance occurs. Upon the occurrence of a Force Majeure that prevents, interferes with or delays the performance by FLNG or Customer, in whole or in part, of any of its obligations hereunder, the Party affected shall give notice thereof to the other Party describing such event and stating the obligations the performance of which are affected (either in the original or in supplemental notices) and stating, as applicable:

- (a) the estimated period during which performance may be prevented, interfered with or delayed, including, to the extent known or ascertainable, the estimated extent of such reduction in performance;
- (b) the particulars of the program to be implemented to resume normal performance hereunder;
- (c) the anticipated portion of the Services Quantity for a Contract Year that will not be made available or received, as the case may be, by reason of Force Majeure; and
- (d) where Section 17.7 applies, the quantity of Services that FLNG reasonably expects to allocate to Customer.

Such notices shall thereafter be updated at least monthly during the period of such claimed Force Majeure specifying the actions being taken to remedy the circumstances causing such Force Majeure.

17.4 Measures

In order to resume normal performance of this Agreement within the shortest time practicable, the Party affected by the Force Majeure shall take all measures to this end which are reasonable under the circumstances, taking into account the consequences resulting from such event of Force Majeure. Prior to resumption of normal performance, the Parties shall continue to perform their respective obligations under this Agreement to the extent not excused as a result of such event of Force Majeure.

17.5 No Extension of Term

The Term shall not be extended as a result of or by the duration of an event of Force Majeure.

17.6 Settlement of Industrial Disturbances

Settlement of strikes, lockouts, or other industrial disturbances shall be entirely within the discretion of the Party experiencing such situations, and nothing herein shall require such Party to settle industrial disputes by yielding to demands made on it when it considers such action inadvisable.

17.7 Allocation of Services

If, as a result of an event of Force Majeure, FLNG is unable to meet its contractual obligations to Customer and any Other Customers under LNG terminal use agreements, FLNG shall allocate the available capability of the Freeport Facility to perform activities similar to the Services in the following order of priority (such allocation herein referred to as the "Major Customer Allocation Priority"):

- (a) first among Major Customers only, based on the ratio that the Maximum LNG Reception Quantity bears to the Aggregate Contracted Capacity for the remainder of such Contract Year (but including Major Customers only); and
- (b) then the remaining capability, if any, among Non-Major Customers based on the same ratio (but including Non-Major Customers only).

During the period of Force Majeure, FLNG shall not (i) provide LNG terminalling services to any Other Customer who is not an Other Customer at the time the Force Majeure commenced; or (ii) otherwise take any action which would reduce the Customer's allocation under Section 17.7.

ARTICLE 18 CURTAILMENT OF SERVICES OR TEMPORARY DISCONTINUATION OF SERVICES

18.1 Scheduled Curtailment or Temporary Discontinuation of Services

If FLNG has notified Customer in connection with the preparation of the Customer LNG Receipt Schedule in Section 5.1 of maintenance to or modification of the Freeport Facility, FLNG shall have the right to curtail or temporarily discontinue the Gas redelivery portion of the Services during the allotted time period specified in such schedule, but only to the extent due to such maintenance or modification. During the period of such curtailment or temporary discontinuation of Services, FLNG shall, from time to time, use reasonable endeavors to update Customer on the expected progress towards completing the maintenance or modification, whichever is applicable. For purposes of this Section 18.1, a curtailment of or temporary discontinuation of Services shall last no more than three (3) consecutive days. Notwithstanding the foregoing, FLNG agrees that, for purposes of this Section 18.1, neither a curtailment nor a temporary discontinuation of Services shall reduce FLNG's obligations to allow berthing, unloading and receipt of Customer's LNG in a quantity up to the Maximum LNG Reception Quantity.

18.2 Unscheduled Curtailment or Temporary Discontinuation of Services

In addition to the rights set forth in Section 18.1, FLNG shall have the right to curtail or temporarily discontinue the Services, in whole or in part, at any time in order to (a) repair the Freeport Facility or (b) protect persons and property, including the Freeport Facility, from harm or damage due to operational or safety conditions. FLNG shall use reasonable endeavors to provide Customer a notice of curtailment or temporary discontinuation as is reasonable under the circumstances, and such notice may be issued for a specific period of time or until further notice is given. FLNG shall use reasonable endeavors to minimize any curtailment or discontinuation of Services to Customer. If, as a result of any unscheduled curtailment or temporary discontinuation of Services pursuant to this Section 18.2, FLNG is unable to meet its contractual obligations to Customer and any Other Customers under LNG terminal use agreements, FLNG shall allocate the available capability of the Freeport Facility to perform activities similar to the Services in accordance with the Major Customer Allocation Priority. If a curtailment or temporary discontinuation of Services occurs under this Section 18.2, FLNG may direct Customer to adjust receipts of LNG and deliveries of Customer's Inventory, as the case may be, with preference given to Major Customers. Notwithstanding the foregoing, FLNG shall have no responsibility to inform Transporters, LNG Vessels, Downstream Pipelines, LNG Suppliers, or any other Persons involved in the transaction as to such curtailment or temporary discontinuation of Services.

ARTICLE 19 ASSIGNMENT

19.1 Restrictions on Assignment

- (a) Consent of Other Party Required. Except as otherwise provided in this Article 19, neither this Agreement nor any rights or obligations hereunder may be assigned by any Party without the prior written consent of the other Party, which consent shall not be unreasonably withheld or delayed. FLNG shall not withhold its consent if (i) the credit rating at the time of the assignment of the proposed assignee is equal to or better than the lower of the credit rating of Customer and a *** rating as per Moody's (or equivalent rating of another reputable credit rating agency, including S&P, or, in the event S&P or Moody's cease providing such ratings, by any equivalent statistical rating organization recognized in the United States), and (ii) the requirements of Section 19.1(b) are met. For greater certainty, a Temporary Release under Section 6.2 shall not be regarded as an assignment under this Article 19.
- (b) Obligation of Assignee. If consent is granted pursuant to Section 19.1(a), the assignee to such assignment must, as a condition to such assignment, deliver to the non-assigning Party its written undertaking to be bound by and perform all obligations of the assignor under this Agreement from and after the effective date of the assignment.

19.2 Permitted Assignments

- (a) <u>Financing</u>.
 - (i) Notwithstanding the provisions of Section 19.1, FLNG shall be entitled to assign, mortgage, or pledge all or any of its rights, interests, and benefits hereunder to a financial institution that provide at least *** million of financing, to secure payment of any indebtedness or working capital incurred or to be incurred, in connection with the construction and term financing of the Freeport Facility. Customer shall provide to the financial institution to whom such indebtedness or working capital is owed a consent to assignment in form and substance customary for similar financing transactions and agreed by such financial institution and Customer.
 - (ii) Notwithstanding the provisions of Section 19.1, Customer shall be entitled to assign, mortgage, or pledge all or any of its rights, interests, and benefits hereunder to a financial institution that provide at least *** million of financing, to secure payment of any indebtedness incurred or to be incurred. FLNG shall provide to the financial institution to whom such indebtedness is owed a consent to assignment in form and substance customary for similar financing transactions and agreed by such financial institution and FLNG.

- (b) Partial Assignments for Periods of *** Years or More Notwithstanding the provisions of Section 19.1 and subject to the provisions of this Section 19.2(b), Customer shall under this Section 19.2(b) be entitled to assign all or a part of its right to use the Services Quantity for a period that is no less than *** (***) years (a "Partial Assignment") if such Partial Assignment (a) is for a quantity of no less than *** MMBTUs per Contract Year and (b) is to:
 - (i) an Affiliate upon notice to, but without requiring the consent of, FLNG; or
 - (ii) any other Person upon the written consent of FLNG, which consent shall not be unreasonably withheld or delayed; provided that, FLNG shall not withhold its consent if (i) the credit rating at the time of the assignment of the proposed assignee is equal to or better than the lower of the credit rating of Customer and a *** rating as per Moody's (or equivalent rating of another reputable credit rating agency, including S&P, or, in the event S&P or Moody's cease providing such ratings, by any equivalent statistical rating organization recognized in the United States), and (ii) the requirements of this Section 19.2(b) are met.

Any Partial Assignment is subject to (a) the assignee executing a terminal use agreement with FLNG (a"Parallel TUA"); and (b) Customer agreeing to appropriate modifications to the Gas redelivery provisions of this Agreement (to ensure that FLNG is at all times capable of performing this Agreement and the Parallel TUA) and to the quantity provisions to reflect such assignment. The terms of the Parallel TUA shall be substantially the same as the terms of this Agreement except to the extent necessary to reflect the differences necessary to implement the Partial Assignment. For the avoidance of doubt, Customer shall have no obligation or liability under the Parallel TUA and the assignor shall be relieved of all rights and obligations hereunder to the extent of the assignment from and after the effective date of the assignment.

19.3 Assignment as Novation

An assignment under this Section 19.1 shall not serve as a novation of this Agreement unless and until, but shall serve as a novation if, the assignee delivers to the non-assigning Party its written undertaking to be bound by and perform all obligations of the assignor under this Agreement from and after the effective date of the novation, as if it were the assignor. In the event of a novation, the assignee shall be deemed to be a Party to this Agreement for all purposes with respect to rights and obligations pertaining to operations hereunder from and after the effective date of the assignment, and the assignor shall be relieved of all rights and obligations hereunder from and after the effective date of the assignment under Sections 19.2(a) and (b) shall not constitute a novation.

ARTICLE 20 CONDITIONS PRECEDENT

20.1 Conditions Precedent

The Parties recognize and agree that only the provisions of Article 1, Article 29, Article 23, Article 24, Article 25, Article 26, Article 27 and this Article 20 shall become effective and binding as of the Effective Date, and that all other provisions contained herein shall not become effective and binding unless and until both of the following conditions ("Conditions Precedent") have been satisfied or waived by FLNG:

- (a) Approvals. By no later than December 31, 2004, FLNG shall have obtained all Approvals that are necessary for FLNG to construct and operate the Freeport Facility; and
- (b) <u>Financing of Freeport Facility.</u> By no later than June 30, 2005, FLNG has provided to Customer evidence that it has successfully arranged and closed on financing of the Freeport Facility.

20.2 Satisfaction of Conditions Precedent

- (a) General. FLNG shall endeavor to procure the satisfaction of the Conditions Precedent by the applicable date provided in Section 20.1 and shall keep Customer reasonably informed as to the progress being made towards satisfaction of each of the Conditions Precedent.
- (b) <u>Notification</u>. FLNG shall notify Customer of such satisfaction promptly upon satisfaction of any Condition Precedent. Moreover, FLNG shall notify Customer promptly upon FLNG's determination that a Condition Precedent cannot be met.
- (c) <u>Waiver</u>. Satisfaction of any Condition Precedent may be waived only by FLNG.
- (d) <u>Termination</u>. If the Condition Precedent in question has not been satisfied or waived by the applicable deadline in Section 20.1, then either Party may thereafter terminate this Agreement with immediate effect by giving notice of such termination to the other Party. Upon termination of this Agreement under this Section 20.2(d), the Parties shall be discharged from any further obligations or Liabilities under this Agreement without prejudice to any rights, obligations or Liabilities which may have accrued up to such date of termination.

ARTICLE 21 TERMINATION

21.1 Early Termination Events

If during the period of construction of the Freeport Facility, Customer reasonably determines that Substantial Completion will not occur by ***, then Customer may terminate this Agreement (such termination referred to as an "Early Termination Event") pursuant to the other provisions of this Article 21.

21.2 Exercise of Right of Termination

- (a) Notice. Upon the occurrence of an Early Termination Event, Customer shall give notice thereof to FLNG.
- (b) <u>Cure.</u> At any time after the expiration of a period of thirty (30) days after Customer gives notice of an Early Termination Event pursuant to Section 21.2(a), Customer may terminate this Agreement with immediate effect by giving notice of such termination to FLNG; provided, however, that Customer may not terminate this Agreement if the circumstances constituting the Early Termination Event have been fully remedied or have ceased to apply.

21.3 Other Termination Provisions

This Agreement is also subject to the termination provisions provided in Section 12.6 and Section 20.2(d).

21.4 Consequences of Termination

Termination of this Agreement under this Article 21 or any other provision of this Agreement shall be without prejudice to any other rights and remedies of either Party arising hereunder or by law or otherwise which arose or accrued prior to or as a result of such termination or by reason of default of either Party.

ARTICLE 22 APPLICABLE LAW

The substantive laws of the State of New York, United States of America, exclusive of any conflicts of laws principles that could require the application of any other law, shall govern this Agreement for all purposes, including the resolution of all Disputes between the Parties.

ARTICLE 23 DISPUTE RESOLUTION

23.1 Dispute Resolution

(a) <u>Notification</u>. A Party who desires to submit a Dispute for resolution shall commence the resolution process by providing the other Party written notice of the Dispute ("Notice of Dispute"). The Notice of Dispute shall contain a brief statement of the nature of the Dispute and the relief requested. The submission of a Notice of Dispute shall toll any applicable statutes of limitation related to the Dispute, pending the conclusion or abandonment of resolution proceedings under this Section 23.1.

- Negotiations. The Parties shall seek to resolve any Dispute by negotiation between Senior Executives. A"Senior Executive" means any individual who has authority to negotiate the settlement of the Dispute for a Party. Within thirty (30) days after the date of the receipt by the applicable Party of the Notice of Dispute (which notice shall request negotiations among Senior Executives), the Senior Executives representing the Parties to the Dispute shall meet at a mutually acceptable time and place to exchange relevant information in an attempt to resolve the Dispute. If a Senior Executive intends to be accompanied at the meeting by an attorney, the other Party's Senior Executive shall be given written notice of such intention at least three (3) days in advance and may also be accompanied at the meeting by an attorney. Notwithstanding the above, any Party may initiate litigation proceedings pursuant to Section 23.1(c) concerning such Dispute within sixty (60) Days after the date of receipt of the Notice of Dispute.
- (c) <u>Litigation</u>. The Parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in any Federal court located in the State of Texas, this being in addition to any other remedy to which they are entitled at law or in equity. In addition, each of the Parties hereto irrevocably and unconditionally (a) consents to submit itself to the personal jurisdiction of any Federal court located in the State of Texas in the event any Dispute arises out of this Agreement, (b) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court, (c) waives (and agrees not to plead or claim) any objection to the laying of venue of any action, suit or proceeding arising out of this Agreement, (d) agrees that it will not bring any action relating to this Agreement, in any court other than any Federal court sitting in the State of Texas, and (e) waives any right to trial by jury with respect to any action related to or arising out of this Agreement.

23.2 Expert Determination

For any decision referred to an expert by the Parties under this Agreement (including under Sections 4.3, 4.5, 4.7, Annex I or Annex II), the Parties hereby agree that such decision shall be conducted expeditiously by an expert selected unanimously by the Parties to the Dispute. The expert shall not be deemed to be acting in an arbitral capacity. The Party desiring an expert determination shall give the other Party to the Dispute notice of the request for such determination. If the parties to the Dispute are unable to agree upon an expert within ten (10) days after receipt of the notice of request for an expert determination, then, upon the request of any of the parties to the Dispute, the International Centre for Expertise of the International Chamber of Commerce shall appoint such expert and shall administer such expert determination through the ICC's Rules for Expertise. The expert shall be and remain at all times wholly impartial, and, once appointed, the expert shall have no expart communications with any of the Parties to the Dispute concerning the expert determination or the underlying Dispute. Both

Parties agree to cooperate fully in the expeditious conduct of such expert determination and to provide the expert with access to all facilities, books, records, documents, information and personnel necessary to make a fully informed decision in an expeditious manner. Before issuing a final decision, the expert shall issue a draft report and allow the parties to the Dispute to comment on it. The expert shall endeavor to resolve the Dispute within thirty (30) days (but no later than sixty (60) days) after his appointment, taking into account the circumstances requiring an expeditious resolution of the matter in dispute. Except as otherwise provided in Section 4.3(c), the expert's decision shall be final and binding on the Parties.

ARTICLE 24 CONFIDENTIALITY

24.1 Confidentiality Obligation

Except as otherwise expressly provided in Section 4.6(a), a Party shall not communicate this Agreement to third parties without the written consent of the other Party. Furthermore, information or documents that come into the possession of a Party (the "Recipient") by means of, or on behalf of, the other Party (the "Discloser") in connection with this Agreement may not be used nor communicated to Persons (other than the Parties) without the written consent of the Discloser. Notwithstanding the prior two sentences, either Party shall have the right to disclose this Agreement and such information or documents without obtaining the other Party's prior consent in any of the situations described below:

- (a) accountants, other professional consultants or underwriters, provided such disclosure is solely to assist the purpose for which the aforesaid were so engaged and further provided that such Persons agree to hold such information or documents under terms of confidentiality equivalent to this Section 24.1, and for the benefit of the Parties;
- (b) bona fide prospective purchasers of all or a part of a Party's or its Affiliate's business, bona fide prospective assignees of all or part of a Party's interest in this Agreement, or providers of finance to either Party in relation to this Agreement or the Freeport Facility, provided that such Persons agree to hold such information or documents under terms of confidentiality equivalent to this Section 24.1, and for the benefit of the Parties;
- (c) to legal counsel, provided such disclosure is solely to assist the purpose for which the aforesaid were so engaged;
- (d) if required by any court of law or any law, rule, or regulation, or if requested by a Governmental Authority having or asserting jurisdiction over a Party and having or asserting authority to require such disclosure in accordance with that authority (including in connection with the resolution of a Dispute under Section 23.1), or pursuant to the rules of any recognized stock exchange or agency established in connection therewith;

- (e) to the operator of a Gas Storage Facility, to Temporary Customers, to prospective Temporary Customers, to Transporters, to LNG Suppliers and to any of the purchasers under the Customer's Gas sales contracts from Customer's Inventory, in each case only to the extent required for the administration of such contracts, and provided that such Persons agree to hold such information or documents under terms of confidentiality equivalent to this Section 24.1, and for the benefit of the Parties;
- (f) to its Affiliates, its shareholders, partners, or its shareholders' and partners' Affiliates, provided that such recipient entity has a bona fide business need for such information and agrees to hold such information or documents under terms of confidentiality equivalent to this Section 24.1;
- (g) to any Government Authorities to the extent such disclosure assists FLNG and Customer in obtaining Approvals;
- (h) to an Expert in connection with the resolution of a Dispute pursuant to Section 23.2;
- (i) to the extent any such information or document has entered the public domain other than through the fault or negligence of the Party making the disclosure;
- (j) to Other Customers by FLNG only in order to allow FLNG to perform its obligations under Section 4.6(b) herein;
- (k) Recipient already possessed the information or document prior to receipt from Discloser;
- (I) Recipient acquired the information or document from a third party without an obligation of confidence; and
- (m) the information or document was independently developed by or for Recipient.

Notwithstanding the foregoing, each Party acknowledges and agrees that certain providers of finance to FLNG as well as FLNG's shareholders and partners may disclose this Agreement and information or documents disclosed pursuant to this Section 24.1 if required by any court of law or any law, rule, or regulation, or if requested by a Governmental Authority having or asserting jurisdiction over such Persons and having or asserting authority to require such disclosure in accordance with that authority, or pursuant to the rules of any recognized stock exchange or agency established in connection therewith. The obligations of Article 24 shall terminate two (2) years after the termination of this Agreement.

24.2 Public Announcements

(a) <u>General.</u> Neither Party may issue or make any public announcement, press release or statement regarding this Agreement unless, prior to the release of the public announcement, press release or statement, such Party furnishes the other

Party with a copy of such announcement, press release or statement, and obtains the approval of the other Party; provided that, notwithstanding any failure to obtain such approval, no Party shall be prohibited from issuing or making any such public announcement, press release or statement if it is necessary to do so in order to comply with the applicable laws, rules or regulations of any Governmental Authority, legal proceedings or stock exchange having jurisdiction over such Party.

(b) Promotional Materials. Notwithstanding any provision in this Section 24.2(a) to the contrary, either Party may use the following in external announcements and publications: (i) information concerning the signing of this Agreement; (ii) the general nature of the Services; and (iii) the general nature of Customer's involvement in the Freeport Facility project; provided, however, that the Party making such external announcement or publication shall not, in doing so, use the trademark, service mark and tradename of the other Party without such other Party's prior written consent.

ARTICLE 25 REPRESENTATIONS AND WARRANTIES

25.1 Representations and Warranties of Customer

As of the date hereof and until the expiration of this Agreement, Customer represents, undertakes and warrants that:

- (a) Customer is and shall remain duly organized and in good standing under the laws of the State of Delaware, U.S.A., duly qualified to do business in those jurisdictions where the nature of its activities or property requires such qualification and to perform its obligations under this Agreement;
- (b) Customer has taken all necessary action to authorize the execution, delivery and performance of its obligations hereunder;
- (c) Customer has not retained, employed or used any broker or intermediary in connection with the negotiation of this Agreement and has no obligation to any third party by way of commissions, finder's fees or similar fees with respect to the execution of this Agreement; and
- (d) neither the execution, delivery nor performance of this Agreement, nor the consummation of any action contemplated herein, conflicts or will conflict with, results or will result in a breach of, or constitutes or will constitute a default under, any provision of Customer's constitutive instruments or any law, judgment, order, decree, rule or regulation of any court, administrative agency or other instrumentality of any Governmental Authority or of any other agreement or instrument to which Customer is a party.

25.2 Representations and Warranties of FLNG

As of the date hereof and until the expiration of this Agreement, FLNG represents, undertakes and warrants that:

- (a) FLNG is and shall remain duly organized and in good standing under the laws of the State of Delaware, U.S.A., duly qualified to do business in those jurisdictions where the nature of its activities or property requires such qualification and to perform its obligations under this Agreement;
- (b) FLNG has taken all necessary action to authorize the execution, delivery and performance of its obligations hereunder;
- (c) FLNG has not retained, employed or used any broker or intermediary in connection with the negotiation of this Agreement and has no obligation to any third party by way of commissions, finder's fees or similar fees with respect to the execution of this Agreement; and
- (d) neither the execution, delivery nor performance of this Agreement, nor the consummation of any action contemplated herein, conflicts or will conflict with, results or will result in a breach of, or constitutes or will constitute a default under, any provision of FLNG's constitutive instruments or any law, judgment, order, decree, rule or regulation of any court, administrative agency or other instrumentality of any Governmental Authority or of any other agreement or instrument to which FLNG is a party.

ARTICLE 26 NOTICES

Except as otherwise specifically provided, all notices authorized or required between the Parties by any of the provisions of this Agreement shall be in writing (in English) and delivered in person or by courier service or by any electronic means of transmitting written communications which provides written confirmation of complete transmission, and addressed to such Party. Oral communication does not constitute notice for purposes of this Agreement, and e-mail addresses and telephone numbers for the Parties are listed below as a matter of convenience only. The foregoing notwithstanding, notices given from LNG Vessels at sea may be given by radio, and notices required under Article 5 may be given by e-mail or via ***. A notice given under any provision of this Agreement shall be deemed delivered only when received by the Party to whom such notice is directed, and the time for such Party to deliver any notice in response to such originating notice shall run from the date the originating notice is received. "Received" for purposes of this Article 26 shall mean actual delivery of the notice to the address of the Party specified hereunder or, in the event notice was given by radio from an LNG Vessel at sea, actual receipt of the communication by radio, or to be thereafter notified in accordance with this Article 26. Each Party shall have the right to change its address at any time and/or designate that copies of all such notices be directed to another Person at another address, by giving written notice thereof to the other Party.

FREEPORT LNG DEVELOPMENT L.P.

Attention: President

1200 Smith Street, Suite 600

Houston, Texas 77002-4310

Fax: (713) 980-2903

Email: president@freeportlng.com

Telephone: (713) 980-2888

THE DOW CHEMICAL COMPANY

Attention: Senior Commercial Manager, LNG

400 W. Sam Houston Pkwy. S.

Houston, Texas 77042-1299

ARTICLE 27 MISCELLANEOUS

27.1 Amendments

This Agreement may not be amended, modified, varied or supplemented except by an instrument in writing signed by FLNG and Customer.

27.2 Approvals

After satisfaction of the Conditions Precedent, each Party shall use reasonable endeavors to maintain in force all of its respective Approvals necessary for its performance under this Agreement. Customer and FLNG shall cooperate with each other wherever necessary for this purpose.

27.3 Successors and Assigns

This Agreement shall inure to the benefit of and be binding upon the respective successors and permitted assigns of the Parties.

27.4 Waiver

No failure to exercise or delay in exercising any right or remedy arising from this Agreement shall operate or be construed as a waiver of such right or remedy. Performance of any condition or obligation to be performed hereunder shall not be deemed to have been waived or postponed except by an instrument in writing signed by the Party who is claimed to have granted such waiver or postponement. No waiver by either Party shall operate or be construed as a waiver in respect of any failure or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver.

27.5 No Third Party Beneficiaries

The interpretation of this Agreement shall exclude any rights under legislative provisions conferring rights under a contract to Persons not a party to that contract. Nothing in this Agreement shall otherwise be construed to create any duty to, or standard of care with reference to, or any liability to, any Person other than a Party.

27.6 Supplemental Terms

The Parties agree that the supplemental terms set forth in Exhibit D are hereby incorporated into and made a part hereof.

27.7 Rules of Construction

(a) <u>Drafting.</u> Each provision of this Agreement shall be construed as though all Parties participated equally in the drafting of the same. Consequently, the Parties acknowledge and agree that any rule of construction that a document is to be construed against the drafting party shall not be applicable to this Agreement.

(b) <u>Priority</u>.

- (i) In the event of a conflict between the terms of this Agreement excluding Annexes I and 11 and Exhibits A, B and C (after such exclusion, the "Base Agreement") and the terms of Annexes I and II or Exhibits A, B and C, all terms of the Base Agreement shall take precedence over Annexes I and II and Exhibits A, B and C.
- (ii) In the event of a conflict between the terms of this Agreement including Annexes I and II and Exhibits A, B, and C and the terms of Exhibit D, the terms of Exhibit D shall govern.
- (iii) In the event that any conflict arises between this Agreement and the Freeport Facility Marine Operations Manual, this Agreement shall prevail. In the event that any conflict arises between this Agreement and the Freeport Services Manual, this Agreement shall prevail.

27.8 Survival of Rights

Any termination or expiration of this Agreement shall be without prejudice to any rights, remedies, obligations and Liabilities which may have accrued to a Party pursuant to this Agreement or otherwise under applicable law. All rights or remedies which may have accrued to the benefit of either Party (and any of this Agreement's provisions necessary for the exercise of such accrued rights or remedies) prior to the termination or expiration of this Agreement shall survive such termination or expiration. Furthermore, the provisions of Section 8.4, Article 12, Article 14, Article 21, Article 22, Article 23, Article 24, Article 26, and Article 27 shall survive the termination or expiration of this Agreement.

27.9 Rights and Remedies

Except where this Agreement expressly provides to the contrary, the rights and remedies contained in this Agreement are cumulative and not exclusive of any rights and remedies provided by law.

27.10 Interpretation

- (a) <u>Headings</u>. The topical headings used in this Agreement are for convenience only and shall not be construed as having any substantive significance or as indicating that all of the provisions of this Agreement relating to any topic are to be found in any particular Article.
- (b) <u>Singular and Plural</u>. Reference to the singular includes a reference to the plural and vice versa.
- (c) <u>Gender</u>. Reference to any gender includes a reference to all other genders.
- (d) Article. Unless otherwise provided, reference to any Article, Section, Schedule, Annex or Exhibit means an Article, Section, Schedule, Annex or Exhibit of this Agreement. Unless otherwise provided, reference to any Clause means a Clause to Exhibit D of this Agreement.
- (e) <u>Include</u>. The words **"include"** and **"including"** shall mean include or including without limiting the generality of the description preceding such term and are used in an illustrative sense and not a limiting sense.
- (f) <u>Time Periods.</u> References to "day," "month," "quarter" and "year" shall, unless otherwise stated or defined, mean a day, month, quarter and year of the Gregorian calendar, respectively. For the avoidance of doubt, a "day" shall commence at 24:00 midnight.
- (g) <u>Statutory References</u>. Unless the context otherwise requires, any reference to a statutory provision is a reference to such provision as amended or reenacted or as modified by other statutory provisions from time to time and includes subsequent legislation and regulations made under the relevant statute.
- (h) <u>Currency</u>. References to all dollars in any form shall be a reference to the lawful currency from time to time of the United States of America.

27.11 Disclaimer of Agency

The rights, duties, obligations and Liabilities of the Parties under this Agreement shall be individual, not joint or collective. It is not the intention of the Parties to create, nor shall this Agreement be deemed or construed to create, a partnership, joint venture or other association or a trust. This Agreement shall not be deemed or construed to authorize any Party to act as an agent, servant or employee for the other Party for any purpose whatsoever except as explicitly set forth in this Agreement. In their relations with each other under this Agreement, the Parties shall not be considered fiduciaries.

27.12 Severance of Invalid Provisions

If and for so long as any provision of this Agreement shall be deemed to be judged invalid for any reason whatsoever, such invalidity shall not affect the validity or

operation of any other provision of this Agreement except only so far as shall be necessary to give effect to the construction of such invalidity, and any such invalid provision shall be deemed severed from this Agreement without affecting the validity of the balance of this Agreement.

27.13 Compliance with Laws

In performance of their respective obligations under this Agreement, each Party agrees to comply with all applicable laws, statutes, rules, regulations, judgments, decrees, injunctions, writs and orders, and all interpretations thereof, of all Governmental Authorities having jurisdiction over such Party.

27.14 Conflicts of Interest

- (a) Restriction on FLNG as Other Customer. FLNG shall avoid any conflict between its own interests and the interests of Customer in relation to obtaining LNG terminalling services from the Freeport Facility. In this regard, FLNG shall not become one of the Other Customers during the Term hereof unless Customer has first consented in writing (such consent not to be unreasonably withheld or delayed) to such expanded business role by FLNG.
- (b) Affiliated Customer Limitations. If it is expected that any of FLNG's joint venture partners or affiliated entities or any partner, shareholder, member, or other direct or indirect equity owner of FLNG or an Affiliate thereof (each an "Affiliated Customer") will become one of the Other Customers during the Term hereof, FLNG shall provide Customer sixty (60) days prior notice thereof. Thereafter, if Customer requests such a meeting, the Parties shall meet to discuss the Affiliated Customer's arrangements with FLNG regarding LNG terminalling services. In the event an Affiliated Customer becomes an Other Customer, the following additional restrictions shall apply:
 - (i) No Affiliated Customer shall be a Major Customer;
 - (ii) No Affiliated Customer shall be given any preference in scheduling over Customer; and
 - (iii) In relation to any rights hereunder which are based on FLNG's Sole Opinion (including FLNG's right under Section 3.1 to allow berthing, unloading and receipt of Customer's LNG in quantities in excess of the Maximum LNG Reception Quantity), FLNG shall not, in exercising such rights, give any preference to an Affiliated Customer which would adversely affect Customer's rights hereunder.
- (c) <u>Conflicts Generally.</u> Except as provided above, the Parties and their Affiliates are free to engage or invest (directly or indirectly) in an unlimited number of activities or businesses, any one or more of which may be related to or in competition with the business activities contemplated under this Agreement, without having or incurring any obligation to offer any interest in such business activities to the other Party.

27.15 Expenses

Each Party shall be responsible for and bear all of its own costs and expenses incurred in connection with the preparation and negotiation of this Agreement.

27.16 Scope

This Agreement constitutes the entire agreement between the Parties relating to the subject matter hereof and supersedes and replaces any provisions on the same subject contained in any other agreement between the Parties, whether written or oral, prior to the date of the original execution hereof.

27.17 Counterpart Execution

This Agreement may be executed in any number of counterparts and each such counterpart shall be deemed an original Agreement for all purposes; provided that no Party shall be bound to this Agreement unless and until both Parties have executed a counterpart.

IN WITNESS WHEREOF, each of the Parties has caused this Agreement to be duly executed and signed by its duly authorized officer as of the Effective Date.

FREEPORT LNG DEVELOPMENT, L.P.

Senior Commercial Manager

By:	Free	Freeport LNG-GP, Inc., its General Partner	
	By:		
		Michael S. Smith	
		Chief Executive Officer	
THE DOW CHEMICAL COMPANY			
	By:		
		B. J. Sumrall, Jr.	

ANNEX I

MEASUREMENTS AND TESTS FOR LNG AT RECEIPT POINT

The Parties shall agree, no later than ***, to the provisions of Annex I; in the event the Parties fail to so agree by December 31, 2004, the matter shall be referred to an expert under Section 23.2.

ANNEX II

MEASUREMENTS AND TESTS FOR GAS AT DELIVERY POINT

The Parties shall agree, no later than ***, to the provisions of Annex II; in the event the Parties fail to so agree by December 31, 2004, the matter shall be referred to an expert under Section 23.2.

EXHIBIT A

TERMS OF FREEPORT FACILITY LEASE

3.2 Adjustment of Base Rent and Other Fees

(a) Consumer Price Index. As used herein, the term "Consumer Price Index" means the Consumer Price Index for all urban consumers published by the Bureau of Labor Statistics, United States Department of Labor (1993-1995 equals 100) for the entire United States. If the publication of the Consumer Price Index is discontinued prior to the expiration or termination of this Lease, or if the present method of ascertaining the index figure is materially or substantially changed, Landlord and Tenant will negotiate in good faith to arrive at a substitute standard which should produce substantially the same results as the Consumer Price Index was designed to give on the date of the execution of this Lease. If necessary, the parties will submit the issue on non-binding mediation prior to and as a condition of instituting litigation in accordance with the following provisions. If, following such non-binding mediation, the parties have not agreed to a substitute standard, the substitute standard shall be determined, upon the written application of either party hereto, by a judge of a district court of Brazoria County, Texas. If said judge will not make such a determination, a substitute standard will be determined by arbitration under the Texas General Arbitration Act (or any similar Texas act then in existence). Each party hereto shall appoint an arbitrator and the two so selected shall select a third, or if they are unable to agree, the third arbitrator shall be appointed by the court under the Texas General Arbitration Act. In the event a new base for the Consumer Price Index is established, and thereafter, for the purposes of making computation as herein provided, the controlling indices stipulated in this Lease shall be adjusted to conform to the new base so that the calculations herein provided shall have the same results as if made on the Consumer Price Index "1993-1995 equals 100". At any time when a substitute standard has not been either agreed upon or determined in accordance with the provisions hereof, Tenant sha

(c) <u>Adjustment of Thru-put Fees</u>. The amount of the Thru-put Fees shall be adjusted as of the first day of the fifth (5th) Lease Year following the GM Start Date and as of the first day of each fifth (5th) Lease Year thereafter to an amount equal to the lesser of (i) the applicable Tariff Rate for Thru-put Fees, or (ii) an amount determined by increasing the rate of \$0.30 per short ton (2000 pounds) by a factor equal to the percentage increase in the Consumer Price Index as determined by the latest published figures next preceding the date on which such adjustment is to be made compared to the Consumer Price Index last reported prior to March 1, 2005.

(f) No Reductions. Notwithstanding any reductions that may occur in the Consumer Price Index at any time, no adjustment of the . . . Thru-put Fees . . . will be made if, following such adjustment, the amount of the applicable fee(s) would be lower than the amount paid or payable by Tenant in the immediately preceding Lease Year.

3.3. Dockage, Wharfage and Thru-put Fees

Tenant understands and agrees that Tenant (or carriers shipping products to or receiving products from Tenant, as the case may be, into and from Port Freeport) will pay to Landlord the rents, dockage and mooring fees, wharfage fees, thru-put charges and other charges (collectively, the "Port Use Fees") as published in Port Freeport, Port Tariff No. 5, or subsequent revisions thereof (the "Tariff Rate"), for services provided by Landlord to Tenant or such other parties, in connection with such party's actual use of the Dock Facilities and other Port Facilities owned by Landlord. Notwithstanding anything to the contrary set forth herein, thru-put fees for LNG (the "Thru-put Fees") during the Initial Term will not exceed the lesser of (a) the applicable Tariff Rate, or (b) the sum of \$0.30 per short ton (2000 pounds), as adjusted from time to time as provided in Section 3.2 hereof. . . .

* *

Capitalized terms used in this Exhibit A are defined in the Freeport Facility Lease as follows:

- "Dock" means a slip and a dock for the berthing of ships (LNG tankers) within the Quintana Terminal area
- "Dock Facilities" means the Dock and the Dock Machinery.
- "Dock Machinery" means such equipment and machinery on the Dock as are reasonably required by Tenant for the loading and unloading of LNG from vessels.
- "Escalation Date" means the earlier to occur of (a) one hundred eighty (180) days following the date of FERC Approval, (b) the date on which construction of the LNG Facility commences or (c) March 1, 2005.
- "GM Start Date" means the date that is the earlier to occur of (a) forty-two (42) months following the Escalation Date, regardless of whether the Dock Facilities have been completed by that time, or (b) the date of Substantial Completion of the LNG Facility.
- "Initial Term" means an initial term of thirty (30) years beginning on March 1, 2003, subject to the provisions of the Lease.
- "Landlord" means Brazos River Harbor Navigation District of Brazoria County, Texas.
- "Lease" means the Ground Lease and Development Agreement entered into as of December 12, 2002 by and between Landlord and Tenant.
- "Lease Year" means a period of twelve (12) consecutive full calendar months.
- "Port Facilities" means property on or about the channels, waterways, docks, slips and other facilities and improvements owned, operated or controlled by Landlord.

"Substantial Completion" means the date on which a certificate of substantial completion or mechanical completion or mechanical completion is issued by the engineer in charge of supervising construction (or, if none, the general contractor in charge of construction).

"Tenant" means Freeport LNG Development, L.P., a Delaware limited partnership, acting by and through its sole general partner, Freeport LNG GP, Inc., a Delaware corporation.

EXHIBIT B

FREEPORT SERVICES MANUAL

The Freeport Services Manual referred to in Section 3.5 shall be limited to the following matters and other matters of a similar nature:

- 1. Details associated with the implementation of Section 5.1 among FLNG, Customer and Other Customers
- 2. Details associated with the Gas delivery procedures in Section 5.2 among FLNG, Customer and Other Customers
- 3. Details associated with the content and format of the ***
- 4. Form of the Release Notice referred to in Section 6.2(b)(i)c
- 5. Details associated with the invoicing process under Article 12, including:
 - a. Format of invoices (electronic and original)
 - b. Numbering systems/codes for all invoice-related documents

EXHIBIT C

PORT LIABILITY AGREEMENT - PORT OF FREEPORT, TEXAS

THIS PORT LIABILITY AGREEMENT (this "Agreement") is effective as of _______, 20___, and is made by and between FREEPORT LNG DEVELOPMENT, L.P., a Delaware limited partnership, represented herein by Freeport LNG - GP, Inc., its General Partner ("FLNG"), and [INSERT NAME(S) OF VESSEL OWNER(S), a [TYPE OF ENTITY AND JURISDICTION OF ORGANIZATION] ([collectively] "Vessel Owner").

RECITALS

WHEREAS, Vessel Owner, using the LNG vessel set forth below under its name and signature ("Vessel"), proposes to deliver certain quantities of liquefied natural gas to FLNG at its marine terminal and receiving, storage and regasification facilities located in the Freeport, Texas area (as more fully defined below, the "Marine Terminal"); and

WHEREAS, Vessel Owner and FLNG (collectively, the "Parties" and individually a "Party") have agreed to allocate the risk of and responsibility for loss and damage resulting from an Incident (as defined below) at the Marine Terminal in the following manner;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. The following terms shall have the following meanings when used herein:
 - "Affiliate" means, with respect to any Person, any other Person which, directly or indirectly, controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities or otherwise.
 - "Incident" means any occurrence or series of occurrences having the same origin in which the Vessel causes (i) damage to the Marine Terminal and/or (ii) an obstruction or danger affecting or interfering with the normal operation of the Marine Terminal.
 - "Terminal Interests" means (i) FLNG, (ii) all Affiliates of FLNG, (iii) all Persons (other than the Vessel Interests and Persons providing fire boats, tugs and escort vessels to Vessel at the Port) employed or providing services at the Marine Terminal in connection with the unloading, storage, or regasification of LNG at the Marine Terminal, and (iv) the employees and agents of all Persons referred to in this paragraph.
 - "Marine Terminal" means FLNG's marine terminal and LNG receiving, storage and regasification facilities located at the Port, including all berths, buoys, gear, craft, equipment, plant, facilities and property of any kind (whether afloat or ashore) located thereat or adjacent thereto and in the ownership, possession or control of the Terminal Interests.

- "Person" means any individual, firm, corporation, trust, partnership, association, joint venture (incorporated or unincorporated), or other business entity.
- "Port" means the port at or near Freeport, Texas, including its anchorage, turning basin and approaches into the Marine Terminal associated therewith.
- "Vessel Interests" means (i) Vessel Owner, (ii) all Affiliates of Vessel Owner, (iii) all Persons (other than the Terminal Interests) participating, employed, or providing services in connection with the ownership or operation (including all operations related to navigation and berthing/unberthing) of the Vessel, and (iv) the employees and agents of all Persons referred to in this paragraph.
- 2. In all circumstances, the Master of the Vessel shall remain solely responsible on behalf of the Vessel Interests for the proper navigation and safety of the Vessel and her cargo.
- 3. Any liability for damage to the Port or the Marine Terminal arising from an Incident shall, as between the Vessel Interests and the Terminal Interests, be borne (i) by the Vessel Interests alone, if the Vessel Interests are wholly or partially at fault and the Terminal Interests are not at fault, (ii) by the Terminal Interests alone, if the Terminal Interests are wholly or partially at fault and the Vessel Interests are not at fault, (iii) by the Vessel Interests and the Terminal Interests, in proportion to the degree of their respective fault, if both are at fault and the degree of such fault can be established, or (iv) by the Vessel Interests and the Terminal Interests equally if neither of them appears to be at fault or it is not possible to establish the degree of their respective fault. In this regard, any acts or omissions of Persons providing fire boats, tugs and escort vessels to Vessel at the Port shall be deemed to be the responsibility of the Vessel Interests.
- 4. (i) Subject to Section 4(ii) below, the total aggregate liability of the Vessel Interests to the Terminal Interests, however arising, in respect of any one Incident, shall not exceed ***. Payment of an aggregate sum of *** to any one or more of the Terminal Interests in respect of any one Incident shall be a complete defense to any claim, suit or demand relating to such Incident made by the Terminal Interests against the Vessel Interests. The liability of the Vessel Interests hereunder shall be joint and several.
 - Vessel Interests shall provide to the Terminal Interests at all times sufficient written evidence that the Vessel's Protection and Indemnity Association has agreed to (a) cover the Vessel Interests as a member of the Association against the liabilities and responsibilities provided for in this Agreement in accordance with its Rules; (b) give the Terminal Interests prior notice of cancellation of the Vessel's entry in such Protection and Indemnity Association; and (c) waive in favor of the Terminal Interests all rights of subrogation of claims by the Protection and Indemnity Association against the Terminal Interests to the extent such claims have been waived in this Agreement by the Vessel Interests.

- 5. As to matters subject to this Agreement and regardless of fault or negligence on the part of any Party, with respect to an Incident:
 - (i) except to the extent expressly preserved in this Agreement, Terminal Interests hereby expressly, voluntarily and intentionally waive any right or claims they might otherwise have against the Vessel Interests under applicable laws or under any port liability agreement or similar port conditions of use previously signed by the Master for the Port; and
 - (ii) except to the extent expressly preserved in this Agreement, Vessel Interests hereby expressly, voluntarily and intentionally ***.

The foregoing waivers shall apply to all Persons claiming through the Terminal Interests or through the Vessel Interests.

6. The substantive law of New York, without regard to any conflicts of law principles that could require the application of any other law, shall govern the interpretation of this Agreement and any dispute, controversy, or claim arising out of, relating to, or in any way connected with this Agreement, including, without limitation, the existence, validity, performance, or breach hereof.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized representatives effective as of the date first set forth above.

FREEPORT LNG DEVELOPMENT, L.P.		[INSERT SIGNATURES OF EA	[INSERT SIGNATURES OF EACH OF VESSEL INTERESTS]		
By:	Freeport LNG - GP, as General Partner				
	Ву:	Ву:			
	Title:	Title:			
		As owner of the			
		Registration No.	Name of Vessel		
		State of Registry			
		·	·		

EXHIBIT D

SUPPLEMENTAL TERMS

In addition to the terms and conditions contained in this Agreement including Annexes I and II and Exhibits A, B, and C (hereinafter the "Terminal Use Agreement"), the Parties further agree to the following:

Financing.

- (a) The Parties acknowledge and agree that they executed a certain Loan Agreement on June 19, 2003 with Freeport LNG Investments, LLP (f/k/a Freeport LNG Investments, LLC) ("Loan Agreement"). The Parties agree that the Loan Agreement shall continue in full force and effect according to its own terms.
- (b) If FLNG desires to obtain equity financing for the Freeport Facility, FLNG shall:
 - (i) discuss the various options with Customer and the merits of Customer becoming an equity owner in the Freeport Facility; and
 - (ii) allow Customer to compete to be one of the participants providing such equity financing.

Moreover, if the Condition Precedent specified in Section 20.1(b) of the Agreement has not been satisfied or waived thereunder, then prior to terminating the Terminal Use Agreement pursuant to Section 20.2, FLNG shall for a period not exceeding thirty (30) days negotiate in good faith with Customer for Customer to potentially provide or arrange the necessary equity or debt financing to allow such condition precedent to be satisfied.

(c) FLNG will discuss with Customer the various financing options and give Customer the opportunity to provide reasonable assistance, if requested by FLNG.

2. Potential Extraction of Liquids.

- (a) The Parties acknowledge that the facilities required under any Liquids extraction project (as discussed below) have not been included in FLNG's application for authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act available at 15 U.S.C. §717f(c) and are not included in capital expenditures addressed in Section 4.4 of the Terminal Use Agreement.
- (b) FLNG will use reasonable efforts to negotiate terminal use agreements with all Other Customers that will provide FLNG the right to extract the Liquids contained in the LNG delivered to the Freeport Facility. In the event such a project is constructed, the Parties agree that FLNG shall provide to Customer the rights to Customer's proportionate share of Liquids contained in equivalent volumes of Gas delivered to the Freeport Facility under the Terminal Use Agreement. The Parties recognize that Customer's LNG will be commingled with that of Other Customers and as a result the share of Liquids available to Customer will reflect the commingled content.

- (c) Because the Freeport Facility will serve only as a tolling facility, FLNG does not intend, under its terminal use agreements, to take title to any LNG, Gas or Liquids. Therefore, it is contemplated that Customer, if Customer so desires, will need to enter into purchase or exchange agreements with Other Customers for the rights to Liquids from the LNG such Other Customers deliver to the Freeport Facility.
- (d) FLNG shall in good faith discuss with Customer the development of and financing for a Liquids extraction project as part of the Freeport Facility and the integration of Customer's existing underground Stratton Ridge, Texas storage facilities for these products. If necessary, the Parties shall amend the Terminal Use Agreement to reflect any agreement regarding the foregoing project. Such agreement regarding a Liquids extraction project would contain a mutually agreeable development and operation fee to FLNG. Customer would have the right to ownership in such extraction project of an amount at least proportional to Customer's share of the Liquids extracted from LNG at the Freeport Facility.

3. Adjustments Related to Compensation Under Other Terminal Use Agreements

- (a) ***
 - (i) the Fee (excluding the Excess Reception Fee),
 - (ii) the U.S. dollar value of Retainage,
 - (iii) the Construction Cost Deviation Amount under Section 4.4(a), if any,
 - (iv) Customer's share of any Additional Capital Costs under Section 4.4(b), if any,
 - (v) New Taxes under Section 4.5, if any, and
 - (vi) Reservation Fee Rebate Amount under Section 4.7, if any,

such sum adjusted to a dollar per MMBTU equivalent assuming that Customer delivers the Maximum LNG Reception Quantity, is referred to as "Customer's TUA Compensation".

- (b) Subject to Clause 3(d) below, FLNG agrees that, with respect to Customer's TUA Compensation only, FLNG shall accord to Customer compensation terms that are no less favorable than ***.
- (c) ***
- (d) ***

4. Right to Delivery of LNG.

In the event Customer requests FLNG to deliver Customer's LNG in liquid form rather than in gaseous form and to deliver such LNG at a point other than the Delivery Point, FLNG shall provide reasonable cooperation to Customer and shall discuss in good faith with Customer the following:

- (a) the construction, operation and maintenance, at Customer's sole cost and expense, of pipelines and other facilities which would be utilized to deliver LNG from an agreed LNG reception point at the Freeport Facility to Customer's downstream facilities at Freeport, Texas;
- (b) Customer's obtaining all easements required for such facilities, provided that FLNG shall use reasonable efforts to assist Customer to obtain at no cost any necessary easements leased to FLNG under the Freeport Facility Lease; and
- (c) Customer's obtaining all Approvals for such facilities at its sole cost, risk and expense.

In relation to constructing, operating and maintaining any such facilities, Customer shall not interfere with operation of the Freeport Facility. In relation to all such LNG delivered to Customer in liquid form, Customer shall not be obligated to bear Retainage. The Parties acknowledge that the required pipelines and other facilities which would be utilized to deliver LNG from an agreed LNG reception point at the Freeport Facility to Customer's downstream facilities at Freeport, Texas have not been included in FLNG's application for authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act. In the event the Parties agree to deliver Customer's LNG in liquid form rather than in gaseous form in accordance with the above, the Parties shall execute as soon as reasonably possible an appropriate amendment to the Terminal Use Agreement confirming such arrangements.

5. Consultation Regarding Freeport Facility Project Implementation

Except as otherwise provided in the Terminal Use Agreement, FLNG is responsible for all aspects and costs of the Freeport Facility; however, FLNG shall seek input from Customer with regard to the following matters relating to the Freeport Facility:

- (a) FLNG's existing application for authorization for the Freeport Facility under Section 3(a) of the Natural Gas Act;
- (b) execution of the engineering, procurement and construction contract;
- (c) construction management;
- (d) start up and LNG terminal operations; and

- (e) expansion of the Freeport Facility in the following ways:
 - (i) adding a third storage tank;
 - (ii) adding a second berthing facility;
 - (iii) adding facilities necessary to allow daily deliveries of gas at the Delivery Point in excess of two trillion BTUs per day; and
 - (iv) increasing the vaporization capacity of the Freeport Facility.

Notwithstanding anything to the contrary in this Clause 5, FLNG shall not be required to obtain Customer's consent prior to carrying out any of the above-mentioned activities.

6. Exclusivity.

Customer and FLNG each agrees that, through and until ***, it will not, without the prior written consent of the other Party (in its Sole Opinion), (a) develop, build, own or operate an LNG receiving facility within a *** radius of the LNG berthing facilities at the Freeport Facility, or (b) assist another Person in the construction, development or operation of any LNG receiving facility within a *** radius of such site.

7. <u>LNG Supplies</u>

FLNG shall provide Customer with reasonable assistance, to the extent requested by Customer, of FLNG's commercial team in locating and contracting for LNG supplies.

8. Additional Negotiations.

The Parties shall negotiate in good faith with a view to agreeing by no later than *** an amendment to Article 21 allowing Customer to terminate the Terminal Use Agreement, subject to Customer's payment to FLNG of an agreed amount upon termination. FLNG's position in this regard is that such amount shall take into consideration the Reservation Fee payable for the remainder of the Term, ***.

9. <u>Effectiveness</u>

The Parties recognize and agree that only the provisions of Clauses 1, 3, 5, 6, 7 and 8 and this Clause 9 shall become effective and binding as of the Effective Date, and that the provisions of Clauses 2 and 4 shall not become effective and binding unless and until both of the Conditions Precedent set forth in Article 20 have been satisfied or waived.

Personal and Confidential

THE SHARES OF COMMON STOCK, \$0.003 PER SHARE ("SHARES"), OF CHENIERE ENERGY, INC. ("CHENIERE") CONSTITUTE SECURITIES THAT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE APPLICABLE SECURITIES LAWS OF ANY STATE ("STATE LAWS"). THE SHARES MAY NOT, AT ANY TIME, BE OFFERED FOR SALE, SOLD OR OTHERWISE TRANSFERRED WITHOUT REGISTRATION UNDER THE ACT AND STATE LAWS, OR DELIVERY TO CHENIERE OF AN OPINION OF LEGAL COUNSEL SATISFACTORY TO CHENIERE THAT SUCH REGISTRATION IS NOT REQUIRED. RESTRICTIONS ON TRANSFER WILL BE IMPRINTED ON THE DOCUMENTS EVIDENCING THE SHARES TO THE FOREGOING EFFECTS.

THE PURCHASE OF SHARES INVOLVES A HIGH DEGREE OF RISK AND SHOULD BE CONSIDERED ONLY BY PERSONS WHO CAN BEAR THE RISK OF LOSING THEIR ENTIRE INVESTMENT.

SUBSCRIPTION AGREEMENT

Cheniere Energy, Inc. c/o Don A. Turkleson 333 Clay Street, Suite 3400 Houston, Texas 77002

Ladies and Gentlemen:

The undersigned (the "<u>Subscriber</u>") hereby subscribes for the number of shares of common stock, \$0.003 per share (the <u>'Shares</u>"), of Cheniere Energy, Inc. (<u>'Cheniere</u>") specified in <u>Section 13</u>. The Subscriber understands that the Shares have not been registered under the Act or any State Law and is being made only to "accredited investors" (as defined in Rule 501 of Regulation D under the Act).

- 1. <u>Subscription</u>. Subject to the terms and conditions hereof, the Subscriber hereby irrevocably subscribes for the number of Shares and makes the purchase payment set forth in <u>Sections 13</u> and <u>14</u> (the "<u>Subscription</u>"). The total amount due for such purchase shall be payable in full in readily available funds, as described ir <u>Section 3</u>. The Subscriber acknowledges that the Shares will be subject to restrictions on transfer pursuant to applicable law and the terms set forth in this Subscription Agreement.
- 2. Acceptance of Subscription and Issuance of Shares. It is understood and agreed that Cheniere shall have the sole right, at its complete discretion, to accept or reject this Subscription, in whole or in part, for any reason and that the same shall be deemed to be accepted by Cheniere only when it is signed by a duly authorized officer of Cheniere. It is understood and agreed that, upon execution of this Subscription Agreement, Cheniere has, in reliance upon the representations and warranties of the Subscriber and against payment for the

Shares, accepted this Subscription. Notwithstanding anything in this Subscription Agreement to the contrary, there shall be no obligation to issue any Shares if such issuance would constitute a violation of the Act or State Laws.

- 3. <u>Payment for Shares</u>. Payment for the Shares shall be received by Cheniere from the Subscriber on or before execution of this Subscription Agreement by the Subscriber, in cash, cashier's check or by wire transfer.
- 4. Representations, Warranties and Covenants of the Subscriber. The Subscriber hereby represents, warrants and covenants to Cheniere and each officer, director and agent of Cheniere that:

(a) General

- (i) If the Subscriber is a natural person, he or she has the legal capacity and all requisite authority to enter into, execute and deliver this Subscription Agreement, to purchase the Shares and to perform all of the obligations required to be performed by the Subscriber hereunder. If the Subscriber is a corporation, partnership, trust or other entity, it is authorized to purchase the Shares and otherwise to comply with its obligations under this Subscription Agreement. The person signing this Subscription Agreement on behalf of such entity is duly authorized by such entity to do so. This Subscription Agreement is the valid and binding agreement of the Subscriber and enforceable against the Subscriber in accordance with its terms.
- (ii) The principal residence of the Subscriber is at the address listed on page 6 of this Subscription Agreement, or if the Subscriber is a corporation, partnership, trust or other entity, such Subscriber is organized and qualified under the law of the state listed in its applicable organizational document provided herewith.
- (iii) The Subscriber is subscribing to invest in Cheniere solely for his, her or its own account, and is not acquiring the Shares as an agent or otherwise for any other person.

(b) Information Concerning the Offering:

- (i) The Subscriber has had access to all of the reports filed by Cheniere with the Securities and Exchange Commission (<u>\$EC</u>") since December 31, 2002 (including, without limitation, Cheniere's annual report on Form 10-K for the year ended December 31, 2002, all quarterly reports on Form 10-Q filed after December 31, 2002, reports on Form 8-K filed on March 7, 2003, June 11, 2003, August 4, 2003, December 22, 2003 and January 14, 2004, and the registration statement on Form S-3 filed on December 22, 2003) (collectively, the "<u>\$EC Documents</u>"). In connection with the Subscription, the Subscriber has had access to, and has relied solely upon, the SEC Documents
- (ii) In formulating a decision to invest in the Shares, the Subscriber has been furnished by Cheniere with all of the information regarding the Company which the

Subscriber has requested or desired to know, has been afforded the opportunity to ask questions of and receive answers from duly authorized officers or other representatives of the Company concerning the Shares and has received any additional information which the Subscriber has requested. The Subscriber has not relied or acted on the basis of any information purported to be given on behalf of Cheniere, except as set forth in the SEC Documents (it being understood that no person has been authorized by Cheniere to furnish any information except as set forth in the SEC Documents).

- (iii) The Subscriber understands that the purchase of the Shares involves various risks, including those outlined in the SEC Documents and in this Subscription Agreement. Investment in the Shares being offered should be regarded as speculative and involving a high degree of risk. The Subscriber is fully aware of the nature of his, her or its investment in the Shares and the lack of liquidity of his, her or its investment in the Shares.
- (iv) The Subscriber understands that no federal or state agency has passed upon the Shares or made any finding or determination concerning the fairness or advisability of this Subscription Agreement.
- (v) The undersigned understands that estimates and projections like those contained in the SEC Documents, by their nature, involve significant elements of subjective judgment and analysis that may or may not be correct; that there can be no assurance that such projections or goals will be attained; and that the projections and estimates contained in the SEC Documents should not be relied upon as a promise or representation of the future performance of Cheniere.

(c) Status of Subscriber; Additional Information:

- (i) The Subscriber has such knowledge, skill and experience in business, financial and investment matters so that he, she or it is capable of evaluating the merits and risks of an investment in the Shares. To the extent necessary, the undersigned has retained, at his, her or its own expense, and relied upon, appropriate professional advice regarding the investment, tax and legal merits and consequences of this Subscription Agreement and owning the Shares.
- (ii) The Subscriber represents and warrants that he, she or it has read the definition of "accredited investor" as defined in Rule 501 of Regulation D under the Act, and acknowledges by his, her or its signature that he, she or it is an "accredited investor," fully capable of subscribing for the Shares by means of meeting the requirements for an accredited investor.
- (iii) The Subscriber agrees to furnish any additional information requested to assure compliance with the Act and State Laws in connection with the purchase and sale of the Shares.

(d) Restrictions on Transfer or Sale of Shares:

- (i) The Subscriber is acquiring the Shares solely for his, her or its own beneficial account, for investment purposes, and not with a view to, or for resale in connection with, any distribution of the Shares. The Subscriber understands that the Shares have not been registered under the Act or any State Law by reason of specific exemptions under the provisions thereof, which depend in part upon the investment intent of the Subscriber and of the other representations made by the Subscriber in this Subscription Agreement. The Subscriber understands that Cheniere is relying upon the representations and agreements contained in this Subscription Agreement (and any supplemental information) for the purpose of determining whether this transaction meets the requirements for such exemptions.
- (ii) The Subscriber shall not sell, assign, pledge, give, transfer or otherwise dispose of any Shares or any interest therein, or make any offer or attempt to do any of the foregoing, except pursuant to a registration of the Shares under the Act and State Laws or in a transaction that is exempt from the registration provisions of the Act and any applicable State Laws. The Company agrees that, within 60 days after the date hereof, it shall file with the SEC on Form S-3 (or any successor form) a shelf registration statement (the "Shelf Registration") pursuant to Rule 415 of the Act covering the offer and resale by the Subscriber of all of the Shares and shall use its reasonable best efforts to cause the Shelf Registration to be declared effective by the SEC as promptly thereafter as possible; *provided, however*, that if the Company is engaged in, or has definitive plans to engage in, within 90 days after the date hereof, any activity or negotiations that, in the good faith determination of the Board of Directors of the Company, would be adversely affected by disclosure that would be required in connection with the registration to the material detriment of the Company, then the Company may delay such registration for a period of 80 days from the date of termination or disclosure of such activity or negotiations. The Company agrees to use its commercially reasonable best efforts to cause the Shelf Registration to remain effective for a period of two years from its initial effectiveness.
- (iii) The Subscriber has not offered or sold any portion of his, her or its Shares and has no present intention of dividing his, her or its Shares with others or of reselling or otherwise disposing of any portion of the Subscriber's Shares either currently or after the passage of a fixed or determinable period of time or upon the occurrence or nonoccurrence of any predetermined event or circumstance.
- 5. <u>Condition to Obligations</u>. The Subscription made hereby may be accepted or rejected by Cheniere at any time after the execution hereof by the Subscriber. No Subscriber shall have the right to demand a return of his, her or its Subscription under any circumstances.
- 6. Waiver, Amendment, Binding Effect. Neither this Subscription Agreement nor any provisions hereof shall be modified, changed, discharged or terminated except by an instrument in writing, signed by the party against whom any waiver, change, discharge or termination is sought. The provisions of this Subscription Agreement shall be binding upon and accrue to the benefit of the parties hereto and their respective heirs, legal representatives, successors and assigns.

- 7. Assignability. Neither this Subscription Agreement nor any right, remedy, obligation or liability arising hereunder or by reason hereof shall be assignable by Cheniere or the Subscriber without the prior written consent of the other.
- 8. <u>Applicable Law.</u> This Subscription Agreement shall be governed by and construed in accordance with the laws of Texas, without regard to the conflict of laws provisions thereof.
- 9. Counterparts. This Subscription Agreement may be executed in any number of counterparts and by facsimile, each of which when so executed and delivered shall be deemed to be an original and all of which together shall be deemed to be one and the same agreement.
- 10. Notices. All notices and other communications provided for herein shall be in writing and shall be deemed to have been duly given if delivered personally or sent by registered or certified mail, return receipt requested, postage prepaid:
 - (a) If to Cheniere, to it at the following address:

Cheniere Energy, Inc. c/o Don A. Turkleson 333 Clay Street, Suite 3400 Houston, Texas 77002

- (b) If to the Subscriber, to him, her or it at the address set forth on the signature page hereto; or at such other address as either party shall have specified by notice in writing to the other.
- 11. <u>Survival</u>. All representations, warranties and covenants contained in this Subscription Agreement shall survive (i) the acceptance of the Subscription by Cheniere and (ii) the death or disability of the Subscriber.
- 12. <u>Notification of Changes</u>. The Subscriber hereby covenants and agrees to notify Cheniere upon the occurrence of any event prior to the closing of the purchase of the Shares pursuant to this Subscription Agreement which would cause any representation, warranty or covenant of the Subscriber contained in this Subscription Agreement to be false or incorrect.

13. <u>Number of Shares Purchased</u> . The undersigned hereby subscribes to purchase the following number of shares of Securities:	
Shares at \$ per Share for an aggregate purchase price of \$	
14. Purchase Payment. The purchase price is being paid herewith by delivery of either cash, cashier's check or wire transfer payable to "Cheniere En	nergy, Inc." in the
amount of \$(\$per share). All payments made as provided in this Section 14 shall be deposited as soon as practicable.	

This Subscription Agreement is executed effective as of	f the day of January, 2004, at	(city),	(state).	
BY: (CHECK ONE)				
INDIVIDUAL CORPORATION (Please include a copy an PARTNERSHIP (Please include a copy of t	the Partnership Agreement authorizing s	signature.)	•	
TRUST (Please include name of trust, name	Please print the EXACT nather purchaser desires the records of the C	ame (registration) to appear in	Agreement of other autho	ization of signature.)
	Address of Purc	chaser		
	Social Security or Taxpay Number of Purc			

	CCUTION: se execute this Subscription Agreement by completing the appropriate section below.
1.	If the subscriber is an INDIVIDUAL, complete the following:
Sign	ature of Purchaser
Nam	e (Please type or print)
2.	If the subscriber is a CORPORATION , complete the following:
	The undersigned hereby represents, warrants and covenants that the undersigned has been duly authorized by all requisite action on the part of the corporation listed below (the "Corporation") to acquire the Shares, that the Corporation has all requisite authority to acquire such Shares, and that the Corporation was not formed for the purposes of acquiring such Shares.
	The officer signing below represents and warrants that each of the above representations or agreements or understandings set forth herein has been made by the Corporation and that he or she has authority under the Articles of Incorporation, bylaws and resolutions of the Board of Directors of such Corporation to execute and deliver this Subscription Agreement on behalf of the Corporation. Such officer has enclosed a true copy of the Articles of Incorporation, the bylaws and, as necessary, the resolutions of the Board of Directors authorizing a purchase of the investment herein, in each case as amended to date.
Nam	e of Corporation (Please type or print)
By:	
Nam	e:
Title	:

	The undersigned represents and warrants that each of the above representations or agreements or understandings set forth herein has been made by the Partnership and that he or she is authorized by such Partnership to execute and deliver this Subscription Agreement. The undersigned has enclosed a true copy of the Partnership Agreement of said Partnership, as amended to date, together with a current and complete list of all partners thereof.
Nam	e of Partnership (Please type or print)
By:	
Nam	e:
Title	:
4.	If the subscriber is a TRUST , complete the following:
	The undersigned hereby represents, warrants and covenants that he or she is duly authorized by the terms of the trust agreement described below (the <u>Trust Instrument</u> ") to acquire the Shares and that the undersigned, as trustee, has all requisite authority to acquire such Shares for the Trust.
	The undersigned, as trustee, executing this Subscription Agreement on behalf of the Trust, represents and warrants that each of the above representations or agreements or understandings set forth herein has been made by the Trust and that he or she is authorized by such Trust to execute and deliver this Subscription Agreement. Such trustee has enclosed a true copy of the Trust Instrument, as amended to date.
Nam	e of Trust (Please type or print)
By:	
Nam	e:

The undersigned hereby represents, warrants and covenants that the undersigned is a general partner of the Partnership named below (the 'Partnership''), the undersigned has been duly authorized by the Partnership to acquire the Shares, the Partnership has all requisite authority to acquire such Shares, and that the Partnership was not formed

If the subscriber is a **PARTNERSHIP**, complete the following:

for the purposes of acquiring such Shares.

3.

Title:

Accepted by Cheniere Energy, Inc. this day of January, 2004.					
		CHENIERE ENERGY, INC.			
		By:			
		Name:			
		Title:			
	9				

CERTIFICATION BY CHIEF EXECUTIVE OFFICER REQUIRED BY RULE 13A-14(a) AND 15d-14(a) UNDER THE EXCHANGE ACT

I, Charif Souki, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Cheniere Energy, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report, based on such evaluation; and
 - c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weakness in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 13, 2004 /s/ Charif Souki

Charif Souki Chief Executive Officer

CERTIFICATION BY CHIEF FINANCIAL OFFICER PURSUANT TO SECURITIES EXCHANGE ACT RULE 13A-14

I, Don A. Turkleson, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Cheniere Energy, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report, based on such evaluation as of the Evaluation Date; and
 - c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weakness in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 13, 2004
/s/ Don A. Turkleson

Don A. Turkleson

Chief Financial Officer

CERTIFICATION BY CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Cheniere Energy, Inc. (the "Company") on Form 10-Q for the period ending March 31, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Charif Souki, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1305, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, to my knowledge, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Charif Souki

Charif Souki Chief Executive Officer

May 13, 2004

CERTIFICATION BY CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Cheniere Energy, Inc. (the "Company") on Form 10-Q for the period ending March 31, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Don A. Turkleson, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1305, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, to my knowledge, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Don A. Turkleson

Don A. Turkleson Chief Financial Officer

May 13, 2004